

Solution-generating transformations in duality-invariant theories and the fluid/gravity correspondence

Fitzhardinge-Berkeley, Joel Alan

The copyright of this thesis rests with the author and no quotation from it or information derived from it may be published without the prior written consent of the author

For additional information about this publication click this link. http://qmro.qmul.ac.uk/xmlui/handle/123456789/9532

Information about this research object was correct at the time of download; we occasionally make corrections to records, please therefore check the published record when citing. For more information contact scholarlycommunications@qmul.ac.uk

Solution-generating transformations in duality-invariant theories and the fluid/gravity correspondence

Joel Alan Fitzhardinge-Berkeley

Submitted in partial fulfilment of the requirements of the Degree of Doctor of Philosophy

> Centre for Research in String Theory, School of Physics and Astronomy, Queen Mary University of London, Mile End Road, London, E1 4NS, UK

Declaration

I, Joel Alan Fitzhardinge-Berkeley, confirm that the research included within this thesis is my own work or that where it has been carried out in collaboration with, or supported by others, that this is duly acknowledged below and my contribution indicated. Previously published material is also acknowledged below.

I attest that I have exercised reasonable care to ensure that the work is original, and does not to the best of my knowledge break any UK law, infringe any third party's copyright or other Intellectual Property Right, or contain any confidential material.

I accept that the College has the right to use plagiarism detection software to check the electronic version of the thesis.

I confirm that this thesis has not been previously submitted for the award of a degree by this or any other university.

The copyright of this thesis rests with the author and no quotation from it or information derived from it may be published without the prior written consent of the author.

Signature:

Date:

Details of collaboration and publications:

The material presented further describes results of the publications [1] (collaboration with David Berman), and [2] (collaboration with David Berman and Felix Rudolph).

Acknowledgements

This PhD thesis has been financially supported by an STFC grant.

My many thanks are to my supervisor Professor David Berman, who has been a knowledgeable and inspiring collaborator, and an understanding and patient guide. I would also like to thank Felix Rudolph for a fruitful, informative and enjoyable collaboration, and Edward Hughes for discussions on differential forms.

I have been incredibly fortunate to have had my family's presence, support, and love, in its countless forms throughout this process. I thank from my heart my dear friend and teacher Maitreyabandhu, who has been there with attentiveness, sense, and the warmest compassion. I also thank Dorcas, Jess, Swati, Zack, Amalavajra, Manjusiha, and the QMUL counselling service, who found me when I lost my way.

There are countless others who have made this work possible whom I have not managed to mentioned here. I am grateful for their contributions nonetheless.

Abstract

We explore dualities and solution-generating transformations in various contexts. Our focus is on the T-duality invariant form of supergravity known as double field theory, the SL(5)-invariant M-theory extended geometry, and metrics dual under the fluid/gravity correspondence to an incompressible Navier-Stokes fluid.

In double field theory (DFT), a wave solution is shown to embed both the F1 string and the pp-wave. For the former, the Goldstone mode dynamics reproduce the duality symmetric string introduced by Tseytlin.

We consider solution-generating techniques in DFT in the presence of an isometry, firstly via Buscher-like transformations in the DFT string σ -model, and secondly via the DFT equations of motion.

In the SL(5)-invariant geometry, we provide a chain rule derivation of the covariant equations of motion, and present a wave solution embedding the M2 brane.

Lastly, solution-generating transformations for metrics with an isometry are considered in the context of the fluid/gravity correspondence. Our focus is on the vacuum Rindler metric dual to a codimension one Navier-Stokes fluid. In particular, when there is a radially directed Killing vector, the dual fluid is found to exhibit an energy scaling invariance valid to all orders in the hydrodynamic expansion.

Contents

1	Introduction		8	
	1.1	Duality-invariant theories	9	
		1.1.1 Double field theory \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots	10	
		1.1.2 U-duality invariant M-theory $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	11	
	1.2	The solution-generating transformation	12	
	1.3	The fluid/gravity correspondence	13	
	1.4	Structure of the thesis 1		
2	Dοι	ible field theory	15	
	2.1	NS-NS supergravity	15	
	2.2	T-duality	16	
	2.3	The doubled space \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots	19	
		2.3.1 Action and equations of motion	22	
	2.4	A double field theory wave solution recovers the string \ldots \ldots \ldots	25	
		2.4.1 Goldstone modes of the wave solution $\ldots \ldots \ldots \ldots$	27	
		2.4.2 Comparison with the σ -model evaluated in the string or wave		
		background	31	
	2.5	Buscher-type transformation in the double field theory $\sigma\text{-model}$	32	
3	U-d	uality invariant M-theory	36	
	3.1	The $SL(5)$ -invariant theory $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	38	
		3.1.1 Action and equations of motion	40	
		3.1.2 An $SL(5)$ wave solution recovers the M2-brane $\ldots \ldots$	43	
4	Solı	ution-generating transformations in NS-NS supergravity and		
	dou	ble field theory	46	
	4.1	Reduction of Einstein gravity with respect to an isometry 47		
	4.2	Solution-generating symmetries from an effective action		

		4.2.1	An example: $(3+1)$ -d vacuum Einstein gravity with an isometry	52							
	4.3	Applic	ation to NS-NS supergravity	54							
		4.3.1	Static Einstein-dilaton theory	56							
		4.3.2	5-d static magnetic NS-NS supergravity	57							
	4.4	Extens	sion to double field theory	58							
5	Solution-generating transformations in the fluid/gravity corre-										
	\mathbf{spo}	ndence		62							
	5.1	Hydro	dynamics	62							
		5.1.1	The fluid equations	63							
		5.1.2	Non-relativistic Navier-Stokes fluids	65							
		5.1.3	Open problems in the Navier-Stokes equations	67							
	5.2	The flu	uid/gravity correspondence	68							
		5.2.1	The Navier-Stokes fluid on a Rindler boundary	69							
	5.3	Dualit	y in the context of holography	74							
	5.4	The generalised Ehlers transformation in the fluid/gravity correspon-									
		dence		75							
		5.4.1	Introducing the transformed fluid metric ansatz	75							
		5.4.2	Deriving the transformation on the fluid	76							
		5.4.3	Energy scaling invariance from an isometry into the bulk $\ .$.	78							
		5.4.4	Fixed viscosity \mathbb{Z}_2 transformations $\ldots \ldots \ldots \ldots \ldots \ldots$	79							
		5.4.5	Are the metric transformations vacuum to vacuum? \ldots .	81							
		5.4.6	Extension to magnetohydrodynamics	83							
6	Con	Conclusion and outlook 8									
	6.1	Double	e field theory and U-duality-invariant M-theory	87							
	6.2	2 Solution-generating techniques in supergravity, double field theory									
		and be	eyond	88							
	6.3	Solutio	on-generating transformations in the fluid/gravity correspondence	90							
\mathbf{A}	Con	Conventions and useful formulae									
	A.1	Differe	ential forms	93							

List of Tables

3.1	The symmetry groups G and their maximally compact subgroups H	
	of 11-dimensional supergravity compactified on T^n	37
3.2	The tangent space and coordinate representation D of E_d duality-	
	invariant extended geometries	37

List of Figures

5.1	The Rindler spacetime dual to a thermal state in global equilibrium.	
	Long-wavelength perturbations of the spacetime are dual at the hy-	
	persurface Σ_c to a codimension one fluid	70

Chapter 1

Introduction

It is perhaps because all phenomena play out in one continuous, inseparable experience, that we look to find a theory which describes all physical phenomena on a unified level, that we search for a unified "theory of everything".

Two great edifices of modern physics, the standard model of particle physics and Einstein's general theory of relativity, both describe in fantastic detail and accuracy their respective phenomena. The standard model is a quantum field theory (QFT) which describes the states and interactions of half-integer spin fermions and integer spin bosons, which are quite naturally unified in the supersymmetric standard model. Supersymmetry additionally removes shortcomings in the standard model such as the hierarchy problem, and provides a dark matter candidate. Meanwhile, Einstein's theory of gravitation describes the interplay between space-time and energy/matter. These two theories, supersymmetric QFT and general relativity can be unified in what is known as supergravity. However, supergravity is non-renormalisable, and so can only provide a low energy effective unified theory.

One candidate for a unified theory valid at all energy scales is superstring theory, a supersymmetric quantum theory whose fundamental objects are one-dimensional rather than the point-like particles of standard QFT. There exist five different forms of string theory: Heterotic SO(32), Heterotic $E_8 \times E_8$, and Types I, IIA and IIB. In their low energy limits, each reproduces a supergravity theory composed of a fermionic sector, and a bosonic sector of two parts: the RR and NS-NS sectors composed respectively of tensor products of fermions and bosons. The NS-NS sector is common to the low energy limit of all string theories.

For superstring theories to preserve Lorentz invariance, they must have a critical

spacetime dimension of ten. Clearly these ten dimensions needs to be reconciled with our apparent four-dimensional world. Analogous to rolling a piece of paper into a thin, infinitely long tube, whose circumference is now much smaller than the wavelength of any common measuring device, and is thus largely invisible, dimensions in string theory are wrapped at distance scales too small for low energy objects to probe. This process is called compactification, and using it one can reduce a spacetime theory from ten to four macroscopic dimensions.

In this process, certain connections are revealed between the different string theories, which rely on the string's extended nature to form closed cycles round compact dimensions. One finds that a Type IIA string compactified on an S^1 of radius Ris equivalent to a Type IIB string compactified on an S^1 with radius α'/R , where $1/2\pi\alpha'$ is the string tension. The same holds for the two heterotic string theories. This equivalence is dubbed T-duality, and is not alone in the string dualities. There also exists a duality between theories at strong and weak coupling, called S-duality, under which Heterotic SO(32) and Type I are dual, and Type IIB is dual to itself.

If one goes to the strong coupling limit of the Type IIA theory, one arrives at an eleven-dimensional theory [3] whose fundamental objects are higher-dimensional branes, and whose low energy effective field theory is eleven-dimensional supergravity [4]. The full eleven-dimensional theory, dubbed M-theory, is largely unknown, but yields in its various compactification limits all five string theories, while T-duality and S-duality become united as components of a larger U-duality web which relates all the string theories [5].

1.1 Duality-invariant theories

We could view this thesis as an exploration of dualities both in themselves, and as the basis for solution-generating techniques in the low energy limit of string theory, and the long-wavelength limit of holography. That this is so was not in fact entirely intentional. However, that we have returned to dualities in various forms serves to illustrate the ubiquitous nature of the duality in modern theoretical physics. We have already encountered the T-, S- and U-dualities of string theory. We look here to illustrate a duality arising in electromagnetism, and from it motivate the *dualityinvariant* approach.

Consider Maxwell's unification of the electric and magnetic fields \mathbf{E} and \mathbf{B} in the

absence of sources

$$\nabla \cdot \mathbf{E} = 0 \qquad \nabla \cdot \mathbf{B} = 0 \tag{1.1}$$

$$\nabla \times \mathbf{E} = -\frac{\partial \mathbf{B}}{\partial t}$$
 $\nabla \times \mathbf{B} = \frac{1}{c^2} \frac{\partial \mathbf{E}}{\partial t}.$ (1.2)

where c is the speed of light. A quick check verifies that the full set of equations are invariant under a simple form of S-duality:

$$(\mathbf{E}/c, \mathbf{B}) \to (\mathbf{B}, -\mathbf{E}/c). \tag{1.3}$$

A system with electric field \mathbf{E}_0 and magnetic field \mathbf{B}_0 is *dual* to another system with electric field $c\mathbf{B}_0$ and magnetic field $-\mathbf{E}_0/c$. They are non-trivially distinguishable from each other, yet exist as dual solutions to Maxwell's equations. The system is also invariant under Lorentz rotations.

We can define a Lorentz-covariant tensor in Minkowski space with coordinates $x^{\mu} = (ct, x^i)$ by

$$F^{i0} = E^i/c \qquad F^{ij} = \epsilon_{ijk}B^k, \tag{1.4}$$

such that the Maxwell equations can now be expressed in a Lorentz-covariant form

$$\nabla_{\mu}F^{\mu\nu} = 0 \qquad \nabla_{\mu}\left(\epsilon^{\mu\nu\sigma\rho}F_{\sigma\rho}\right) = 0, \tag{1.5}$$

where ϵ is the antisymmetric symbol (see app. A). In addition, noting that (1.3) can be generated by

$$F_{\mu\nu} \to \epsilon_{\mu\nu\sigma\rho} F^{\sigma\rho},$$
 (1.6)

we illustrate the possibility of promoting duality (in this case both S-duality and Lorentz duality) to manifest symmetry.

1.1.1 Double field theory

This process is analogous the promotion of T-duality to a manifest symmetry of generalised geometry by Hitchin [6] and Gaultieri [7], which extends the tangent space TM of a d-dimensional theory, whose sections are vectors parametrising diffeomorphisms of a symmetric $g_{\mu\nu}$, to $TM \oplus T^*M$, such that sections additionally include one-forms parametrising gauge transformations of a two-form field $B_{\mu\nu}$. In particular, there exists in this generalised geometry a natural $O(d, d, \mathbb{R})$ -invariant inner product. In double field theory (DFT), with origins in [8–12] (see [13–15] for reviews and [16–84] for further literature) one additionally doubles the coordinate space itself, and allows dependence of the fields on the the full set of coordinates. The extended coordinates are reduced to a physical subset by a *weak constraint* arising from the level matching condition of string theory, which can be supplemented by a *strong constraint* which ensures closure of gauge transformations on the extended space. Various solutions to these constraints yield NS-NS supergravity backgrounds related under T-duality, in addition to gauged supergravities with non-geometric fluxes recovered via Scherk-Schwarz reductions [78–85].

The supergravity fields (including the dilaton) are collected in an $O(d, d, \mathbb{R})/O(d, \mathbb{R}) \times O(d, \mathbb{R})$ coset representative generalised metric and a DFT dilaton. These objects' dynamics are governed by an action whose minima provide the equations of motion. When directions are compactified, T-duality transformations appear as coordinate transformations on the doubled space.

1.1.2 U-duality invariant M-theory

In M-theory's low energy limit of eleven-dimensional supergravity, compactification on T^n reveals an E_n symmetry originating from the M-theory U-duality group [86]. Analogous to how $O(d, d, \mathbb{R})$ is promoted to a manifest symmetry in generalised geometry and DFT, one can promote U-duality to a manifest symmetry, constructing E_d duality-invariant theories without requiring compactified directions. This is achieved in various frameworks: d-dimensional generalised geometry where the tangent bundle is extended to allow for membrane charges [87-93]; extended geometry where the d spacetime dimensions are supplemented with w "wrapping" coordinates dual to membrane charges, and fields are allowed to depend on all coordinates [36,75,94–103]. From the perspective of the parent eleven-dimensional theory, the d-dimensional internal space is decoupled from the (11 - d)-dimensional external space; exceptional field theory where the full eleven dimensional supergravity is supplemented by w wrapping coordinates, and the internal and eternal spaces are no longer decoupled [104–119]. Without a level matching constraint equivalent for the membrane, the dependence on the additional coordinates appearing in extended and exceptional geometry is constrained by closure of gauge transformations on the extended space. Again, gauged supergravities can be recovered from exceptional and extended geometry via Scherk-Schwarz reductions [85, 120–122].

1.2 The solution-generating transformation

Many sets of equations we encounter in physics are intractable as they stand, examples of particular relevance to this work being the Einstein field equations of general relativity and the incompressible Navier-Stokes equations of hydrodynamics. A natural question with solving such notoriously difficult equations, is whether there is some sensible constraint we can place on solutions, such as time independence, which will reduce the equations to a more tractable system. Not only would the more tractable system be easier to solve, it may possess symmetries not present in the full, unconstrained set of equations. Such symmetries would manifest as dualities between solutions. In particular, simple solutions may be dual to more complex ones, and we can use the symmetry groups to generate these additional solutions perhaps more easily than trying to solve any of the equations directly. Let us offer an example from *d*-dimensional static vacuum Einstein gravity, originally derived by Buchdahl [123]. We will present a derivation using a similar method to that we will employ in §4.2. Consider a spacetime static with respect to a coordinate $x^{\mu} = (x^0, x^a)$, the metric can be written

$$g_{\mu\nu} = \begin{pmatrix} F & 0\\ 0 & g_{ab} \end{pmatrix} = \begin{pmatrix} F & 0\\ 0 & F^{-1/(d-3)}\hat{\gamma}_{ab} \end{pmatrix}, \qquad \frac{\partial}{\partial x^0}g_{\mu\nu} = 0, \tag{1.7}$$

where

$$\hat{\gamma}_{\mu\nu} = F^{1/(d-3)}(g_{\mu\nu} - F^{-1}g_{0\mu}g_{0\nu})$$
(1.8)

is a metric on the space orthogonal to the isometry ∂_0 . The vacuum field equations $R_{\mu\nu} = 0$ reduce to

$$(\hat{\gamma}R)_{\mu\nu} = \frac{d-2}{4(d-3)} \frac{1}{F^2} \hat{D}_{\mu} F \hat{D}_{\nu} F \qquad (\hat{\gamma}^{-1})^{\mu\nu} \hat{D}_{\mu} \left(\frac{1}{F} \hat{D}_{\nu} F\right) = 0, \tag{1.9}$$

where \hat{D}_{μ} is the covariant derivative, and $(\hat{\gamma}R)_{\mu\nu}$ the Ricci tensor, with respect to $\hat{\gamma}_{\mu\nu}$. For fixed $\hat{\gamma}_{\mu\nu}$, equations (1.9) are invariant under

$$F \to \frac{1}{F},$$
 (1.10)

and thus we can generate the vacuum solution

$$g'_{\mu\nu} = \begin{pmatrix} F^{-1} & 0\\ 0 & F^{1/(d-3)}\hat{\gamma}_{ab} \end{pmatrix} = \begin{pmatrix} F^{-1} & 0\\ 0 & F^{2/(d-3)}g_{ab} \end{pmatrix}.$$
 (1.11)

We will see throughout this thesis how we can apply generalisations of this technique to the NS-NS fields of supergravity, the generalised metric of DFT, and to the hydrodynamics holographically dual to gravity, to which we now turn our attention.

1.3 The fluid/gravity correspondence

The limit of long wavelengths and long time scales of any interacting field theory at finite temperature is governed by hydrodynamics. This fact, leading to the great plethora of such phenomena, from the atmosphere to the oceans, from plasmas in the RHIC to the sun itself, make the study of hydrodynamics not only fascinating, but virtually unavoidable.

This area was perhaps somewhat unexpectedly connected with string theory by Maldacena's observation [124] that the dynamics on a codimension one hypersurface in a general relativistic anti-de-Sitter bulk spacetime is dual to a conformal field theory. This is an example of holography, where a gravitational theory is dual to a codimension one quantum field theory. It was realised in [125, 126], that if we scale to the hydrodynamic limit of this duality, we find that the long wavelength, long time scale limit of general relativity is holographically dual to hydrodynamics. This is the fluid/gravity correspondence, which has been explored in [1, 127-175](see [176,177] for reviews). One finds that under a hydrodynamic expansion governed by this limit, whilst demanding regularity on the future horizon, the Goldstone mode dynamics of metrics in general relativity are governed by the equations of motion of a codimension one fluid [178–184]. The correspondence has been shown to hold for higher derivative gravity corrections [185–190]. By considering various bulk spacetime backgrounds and stress-energy content, one finds dual hydrodynamics on curved backgrounds [191, 192] and in the presence of forcing terms responsible for e.g. magnetohydrodynamics [193].

1.4 Structure of the thesis

We will begin in chapter 2 with an exploration of DFT, starting with an introduction in §2.1 to the NS-NS sector of supergravity and in §2.2 to T-duality via Buscher's procedure of the string σ -model. Section 2.3 continues with an introduction to $O(d, d, \mathbb{R})$ -invariant double field theory, adapted largely from [13, 15, 194]. In §2.3.1, a novel derivation of the full DFT equations of motion is presented and, using this, we offer in §2.4 a pp-wave type DFT solution which embeds both the fundamental string and the spacetime pp-wave for two different solutions to the section condition. The Goldstone modes of this solution are analysed in §2.4.1 for the string case, where the dynamics are governed by a self-duality relation in agreement with the duality-symmetric string of [10–12]. We proceed in §2.5 with an adaptation of the Buscher procedure for the string σ -model to σ -model actions in DFT.

We move in chapter 3 to an introduction to E_d invariant geometries, in particular the SL(5)-invariant extended geometry in §3.1, including in §3.1.1 a full chain rule derivation of the equations of motion, and in §3.1.2 a wave solution which corresponds to the M2-brane from the spacetime perspective.

Chapter 4 discusses solution generating techniques in general relativity, supergravity and DFT in the presence of an isometry. We detail in §4.2 a (d-1) + 1split of *d*-dimensional spacetime in the presence of one Killing vector. We introduce in §4.2 a formalism due to [195] for determining solution generating symmetries of general relativity with an isometry in the case that the dynamics can be encoded in an effective action of scalar potentials. In §4.3 this is applied to two sectors of static NS-NS supergravity. Lastly, we briefly explore analogous solution-generating techniques in DFT in §4.4.

In chapter 5 we look to extend these solution-generating techniques in the presence of an isometry to the fluid/gravity correspondence. Section 5.1 begins with an introduction, largely adapted from [176], to relativistic hydrodynamics and the non-relativistic limit governed by the incompressible Navier-Stokes equations (INS). Section 5.2 continues with a brief introduction to the fluid/gravity correspondence, leading into a summary of the derivation in [183] of a vacuum metric dual to a codimension one incompressible Navier-Stokes fluid. We proceed in §5.4 to apply the generalised Ehlers group of [196] in the presence of a Killing vector to this vacuum metric to derive transformations of the Navier-Stokes fluid, and find in §5.4.2 and §5.4.4 a set of invariance transformations of the fluid. We discuss how these transformations act from the bulk spacetime perspective in §5.4.5. Section §5.4.6 takes a brief look at similar solution-generating transformations for a metric dual to magnetohydrodynamics.

Finally, chapter 6 concludes with a discussion of our results and their implications, and a look at possible directions of further research. The appendices include notations and useful formulae for differential forms.

Chapter 2

Double field theory

2.1 NS-NS supergravity

In the low energy limit of the five string theories one obtains various supergravity theories, each with fermionic and bosonic contributions. These supergravity theories share a common bosonic NS-NS sector, whose fields derive from purely bosonic modes on the string worldsheet. This sector forms a consistent supersymmetric theory on its own—supersymmetry generators transform the NS-NS fields among themselves. On this basis, it is quite reasonable, for simplicity, to restrict our analysis to the bosonic NS-NS sector of supergravity.

The degrees of freedom are the spacetime metric $g_{\mu\nu}$, two-form Kalb-Ramond field $B_{\mu\nu}$, and dilaton ϕ . These vary as functions of the coordinates x^{μ} , and the action is given by

$$S = \frac{1}{2\kappa^2} \int d^d x \sqrt{-\det(g)} e^{-2\phi} \left(R + 4\partial_\mu \phi \partial^\mu \phi - \frac{1}{12} H_{\mu\sigma\rho} H^{\mu\sigma\rho} \right), \qquad (2.1)$$

where κ is a constant. The three-form field strength of B

$$H = \mathrm{d}B\tag{2.2}$$

additionally satisfies the Bianchi identity

$$\mathrm{d}H = 0. \tag{2.3}$$

The minima of (2.1) are given by the equations of motion

$$R_{\mu\nu} - \frac{1}{4} H_{\mu}{}^{\sigma\rho} H_{\nu\sigma\rho} + 2\nabla_{\mu} \nabla_{\nu} \phi = 0 \qquad (2.4a)$$

$$\nabla_{\mu}(e^{-2\phi}H^{\mu\nu\rho}) = 0 \tag{2.4b}$$

$$\frac{1}{6}H^{\mu\sigma\rho}H_{\mu\sigma\rho} + 2\nabla^{\mu}\nabla_{\mu}\phi - 4\nabla^{\mu}\phi\nabla_{\mu}\phi = 0.$$
(2.4c)

The equations of motion and action are invariant under gauge transformations of the two-form, along with diffeomorphisms whose infinitesimal action is given by the Lie derivative. For one-forms ℓ and infinitesimal vectors k, these are

$$g \to g + \mathcal{L}_k g \qquad B \to B + \mathcal{L}_k B + \mathrm{d}\ell \qquad \phi \to \phi + \mathcal{L}_k \phi,$$
 (2.5)

where the Lie derivative of (zero weight) tensors $T^{\mu_1...\mu_p}{}_{\nu_1...\nu_q}$ with respect to vectors a is

$$(\mathcal{L}_{a}T)^{\mu_{1}...\mu_{p}}{}_{\nu_{1}...\nu_{q}} = a^{\rho}\partial_{\rho}T^{\mu_{1}...\mu_{p}}{}_{\nu_{1}...\nu_{q}}$$

$$-\sum_{s=1}^{p}T^{\mu_{1}...\mu_{s-1}\rho\mu_{s+1}...\mu_{p}}{}_{\nu_{1}...\nu_{q}}\partial_{\rho}a^{\mu_{s}}$$

$$+\sum_{r=1}^{q}T^{\mu_{1}...\mu_{p}}{}_{\nu_{1}...\nu_{r-1}\rho\nu_{r+1}...\nu_{q}}\partial_{\nu_{r}}a^{\rho}.$$
(2.6)

The action exhibits no further symmetries as is, but when the theory is compactified on T^r , the equations of motion (but not the action) are invariant under a solutiongenerating symmetry group $O(r, r, \mathbb{R})$.

2.2 T-duality

The Buscher rules

The extended nature of the string allows it to wrap round non-contractible compact dimensions, resulting in "winding mode" contributions to its dynamics in addition to the momentum mode contributions familiar from particle dynamics. We will present here a type of duality that relates these two modes, called T-duality, which can be traced back to work by [197,198]. A thorough review of the subject is given in [194]. The particular presentation given here was derived by Buscher [199,200] from the string σ -model, whose action is

$$S = \frac{1}{4\pi\alpha'} \int d^2\sigma \sqrt{\det h} \left[h^{ab} g_{\mu\nu} \partial_a x^{\mu} \partial_b x^{\nu} + i\epsilon^{ab} B_{\mu\nu} \partial_a x^{\mu} \partial_b x^{\nu} + \alpha' R^{(2)} \phi(x) \right], \quad (2.7)$$

where the target space (coordinates x^{μ}) has metric $g_{\mu\nu}$, two-form $B_{\mu\nu}$, and dilaton ϕ , and the worldsheet (coordinates σ^a) has metric h_{ab} with scalar curvature $R^{(2)}$, and alternating symbol ϵ_{ab} (see app. A). The worldsheet is two-dimensional and thus the metric h_{ab} can be brought (locally on the worldsheet) to Minkowski form via conformal rescaling and diffeomorphisms.

If one imposes an abelian isometry $\partial/\partial x^0 = 0$ on the target space, where $x^{\mu} = (x^0, x^m)$, then the action

$$S' = \frac{1}{4\pi\alpha'} \int d^2\sigma \sqrt{\det h} \left[h^{ab} (g_{00}V_aV_b + 2g_{0n}V_a\partial_b x^n + g_{mn}\partial_a x^m\partial_b x^n) + i\epsilon^{ab} (2B_{0n}V_a\partial_b x^n + B_{mn}\partial_a x^m\partial_b x^n) + 2i\hat{x}^0\epsilon^{ab}\partial_a V_b + \alpha' R^{(2)}\phi(x) \right]$$
(2.8)

provides equation of motion for Lagrange multiplier \hat{x}^0

$$\epsilon^{ab}\partial_a V_b = 0, \tag{2.9}$$

which on topologically trivial worldsheets fixes $V_a = \partial_a x^0$, which returns the original action (2.7). If one instead integrates by parts the term with \hat{x}^0 , and substitutes in the resulting equations of motion for V_a ,

$$V_{a} = -\frac{1}{g_{00}} (g_{0n} \partial_{a} x^{n} + i h_{ab} \epsilon^{bc} (B_{0n} \partial_{c} x^{n} + \partial_{c} \hat{x}^{0})), \qquad (2.10)$$

one recovers the dual action

$$S' = \frac{1}{4\pi\alpha'} \int d^2\sigma \sqrt{\det h} \left[h^{ab} g'_{\mu\nu} \partial_a \hat{x}^{\mu} \partial_b \hat{x}^{\nu} + i\epsilon^{ab} B'_{\mu\nu} \partial_a \hat{x}^{\mu} \partial_b \hat{x}^{\nu} + \alpha' R^{(2)} \phi'(\hat{x}) \right], \quad (2.11)$$

with dual coordinates $\hat{x}^{\mu} = (\hat{x}^0, x^m)$, background

$$g'_{00} = \frac{1}{g_{00}} \qquad g'_{0n} = \frac{B_{0n}}{g_{00}} \qquad g'_{mn} = g_{mn} - \frac{g_{0m}g_{0n} - B_{0m}B_{0n}}{g_{00}}$$

$$B'_{0m} = \frac{g_{0m}}{g_{00}} \qquad B'_{mn} = B_{mn} - \frac{g_{0m}B_{0n} - B_{0m}g_{0n}}{g_{00}}$$
(2.12)

and dilaton

$$\phi' = \phi - \frac{1}{2} \ln g_{00}. \tag{2.13}$$

The dilaton shift is found by a careful regularisation of determinants [12, 201, 202].

This is called T-duality, and can be generalised to to the case of toroidal T^r compactifications defined by r compact bosonic directions identified as $X^i \sim X^i + 2\pi$, where the full T-duality group is given by $O(r, r, \mathbb{Z})$ acting on the background matrix

$$E_{ij} = g_{ij} + B_{ij} \tag{2.14}$$

and dilaton in the non-linear form

$$E' = \frac{aE+b}{cE+d} \qquad \phi' = \phi + \frac{1}{4} \ln \left[\frac{\det(g')}{\det(g)} \right] \qquad \begin{pmatrix} a & b\\ c & d \end{pmatrix} \in O(r, r, \mathbb{Z}).$$
(2.15)

The $O(r, r, \mathbb{Z})$ group is generated by three elements:

• Large diffeomorphisms of T^r preserving periodicities which produce a $GL(r, \mathbb{Z})$ basis change

$$O_A = \begin{pmatrix} A^T & 0\\ 0 & A^{-1} \end{pmatrix} \quad \text{where} \quad A \in GL(r, \mathbb{Z})$$
 (2.16)

• Integer shifts in the *B* field which enact a shift $2\pi\mathbb{Z}$ in the action and thus do not change the path integral

$$O_{\Omega} = \begin{pmatrix} 1_r & \Omega \\ 0 & 1_r \end{pmatrix} \quad \text{where} \quad \Omega_{ij} = -\Omega_{ji} \in GL(r, \mathbb{Z}) \quad (2.17)$$

• Factorised dualities, which in even dimensions correspond to the generalisation of the radial inversion $g_{00} \rightarrow 1/g_{00}$ in (2.12)

$$O_{e_s} = \begin{pmatrix} 1_r - e_s & e_s \\ e_s & 1_r - e_s \end{pmatrix} \quad \text{where} \quad (e_s)_{ij} = \delta_{sj} \delta_{sk} \tag{2.18}$$

for some $s \in [1, r]$.

In the low energy limit, this symmetry becomes the $O(r, r, \mathbb{R})$ of NS-NS supergravity compactified on T^r .

2.3 The doubled space

There has been a long-standing interest [10–12] in formulating string dynamics in a manifestly T-duality-invariant manner, where T-duality acts linearly on fields. Exactly how to do this was greatly aided by the generalised geometry construction of Hitchin [6] and Gaultieri [7], who studied structures on the tangent space $TM \oplus T^*M$, where sections consist of the sum of vectors and one-forms. This sum can then parametrise gauge transformations of a symmetric tensor g and two-form B. Of particular interest here, is that the tangent space has a natural $O(d, d, \mathbb{R})$ inner product which, in local coordinates such that $A = A^{\mu}\partial_{\mu}$ and $a = a_{\mu}dx^{\mu}$, becomes

$$\langle A+a, B+b \rangle = \frac{1}{2} (A^{\mu} b_{\mu} + B^{\mu} a_{\mu}).$$
 (2.19)

In double field theory (DFT), one doubles not just the tangent space, but the coordinates space itself. We shall follow the expositions in [13–15] for this short introduction to DFT. The dimension of the coordinate space is given by the dimension 2d of the fundamental of $O(d, d, \mathbb{R})$, and the doubled coordinates

$$X^M = (x^\mu, \tilde{x}_\mu) \tag{2.20}$$

then transform linearly as a group representative

$$X^M \to O^M{}_N X^N, \tag{2.21}$$

where O is an $O(d, d, \mathbb{R})$ matrix defined to preserve the inner product

$$O_M{}^P \eta_{PQ} O_N{}^Q = \eta_{MN}, \qquad \eta_{MN} = \begin{pmatrix} 0 & \delta^\mu_\nu \\ \delta^\nu_\mu & 0 \end{pmatrix}$$
(2.22)

and we have introduced the $O(d, d, \mathbb{R})$ metric η . This $O(d, d, \mathbb{R})$ is a global symmetry of DFT.

The supergravity fields g, B and ϕ will now depend on the 2d coordinates of this doubled space. This is clearly undesired from a supergravity perspective. Before we formally introduce how this is remedied, let us illustrate the connection with the $O(d, d, \mathbb{Z})$ of T-duality. Consider a particular case where the fields depend only on the directions $x^{\mu \neq d}$ (and not the \tilde{x}_{μ}), as is the case in supergravity with an isometry $\partial/\partial x^{d} = 0$. Then we are quite free to demand that the physical coordinates of spacetime are the $x^{\mu \neq d}$, plus one of the pair (x^d, \tilde{x}_d) . This freedom corresponds to the $O(1,1,\mathbb{Z}) = \mathbb{Z}_2$ T-duality transformation. In particular, the choice x^d corresponds to the usual spacetime coordinate, whilst the choice \tilde{x}_d corresponds to the T-dual coordinate. In this way, in compactification on tori, one can see the \tilde{x}_{μ} as "winding" coordinates, Fourier dual to the winding momenta in the string context, though this is not their interpretation in general. One can thus see the connection with the T-duality of closed string theory (for a more detailed discussion, see [45]).

In fact, one finds on formulating the level matching constraint in closed string field theory in a $O(d, d, \mathbb{R})$ covariant manner, the condition needed to restrict the dependence of the fields on the number of coordinates [19]. It is termed the *weak constraint* and requires that

$$Y^{MN}{}_{PQ}\partial_M\partial_N\theta = 0 \tag{2.23}$$

annihilates single fields θ , where the DFT Y-tensor¹

$$Y^{MN}{}_{PQ} = \eta^{MN} \eta_{PQ}. \tag{2.24}$$

One can define local gauge transformations on this space in terms of a generalised Lie derivative with respect to a generalised vector U^M , the most general form of which, acting on a tensor $A_M{}^N$ with weight w(A), is

$$(L_U A)_M{}^N = U^P \partial_P A_M{}^N + A_P{}^N \partial_M U^P - A_M{}^P \partial_P U^N + A_M{}^P Y^{QN}{}_{PR} \partial_Q U^R - A_P{}^N Y^{QP}{}_{RM} \partial_Q U^R + w(A) \partial_P U^P A_M{}^N.$$
(2.25)

It is important to note that these generalised coordinate transformations are not conventional coordinate transformations on a doubled space—one cannot form (2.25) from the conventional diffeomorphism group on the 2*d* space with the additional condition that diffeomorphisms respect the $O(d, d, \mathbb{R})$ metric η_{MN} . To ensure that the generalised Lie derivative sends tensors into tensors requires its closure

$$[L_{U_1}, L_{U_2}]V^M = L_{[U_1, U_2]_C}V^M$$
(2.26)

on to the *C*-bracket

$$[U_1, U_2]_C \equiv \frac{1}{2} \left(L_{U_1} U_2 - L_{U_2} U_1 \right), \qquad (2.27)$$

¹We write the weak constraint explicitly in terms of the Y-tensor as this form is also applicable to the E_d -invariant extended geometries we will discuss in chapter 3 (where the form of the Y-tensor depends on the relevant duality group E_d). In DFT, it is frequently written in the literature in the simpler yet equivalent form $\eta^{MN} \partial_M \partial_N \theta = 0$, for single fields θ .

which, for weight zero vectors U_1, U_2, V , requires

$$Y^{RS}{}_{PQ}\left(U_1^P\partial_R U_2^Q\partial_S V^M + 2\partial_R U_1^P\partial_S U_2^M V^Q\right) - U_1 \leftrightarrow U_2 = 0.$$
(2.28)

One solution to this is the *strong constraint* or *section condition*, that in addition to the weak constraint (2.23),

$$Y^{MN}{}_{PQ}\partial_M\theta_1\partial_N\theta_2 = 0 \tag{2.29}$$

annihilates products of fields. In [48] the authors discuss the CFT origin of this constraint. It can be shown that all solutions to the strong constraint are $O(d, d, \mathbb{R})$ rotations of $\tilde{\partial}^{\mu}\theta = 0$ for any field θ . We will refer to the frame where all objects have no dependence on the \tilde{x}_{μ} coordinates as the *supergravity frame*. We note that (2.29) is not the most general solution to (2.28). More general solutions are possible, and are pivotal in the Scherk-Schwarz reduction of DFT to gauged supergravities [78–80, 83–85, 120].

The supergravity fields may be cast into representations of $O(d, d, \mathbb{R})$. One can unify the spacetime metric and two-form into a $O(d, d, \mathbb{R})/O(d, \mathbb{R}) \times O(d, \mathbb{R})$ coset representative² generalised metric

$$H_{MN} = \begin{pmatrix} g_{\mu\nu} - B_{\mu\rho}g^{\rho\sigma}B_{\sigma\nu} & B_{\mu\rho}g^{\rho\nu} \\ -g^{\mu\sigma}B_{\sigma\nu} & g^{\mu\nu} \end{pmatrix}, \qquad (2.30)$$

transforming as an $O(d, d, \mathbb{R})$ tensor:

$$H_{MN} \to O_M{}^P H_{PQ} O_N{}^Q. \tag{2.31}$$

The coset form ensures the metric has inverse $H^{MN} = \eta^{MP} H_{PQ} \eta^{QN}$. Meanwhile, the supergravity dilaton is shifted to define an $O(d, d, \mathbb{R})$ scalar density DFT dilaton d defined by³

$$e^{-2d} = \sqrt{-\det(g)} e^{-2\phi}.$$
 (2.32)

Infinitesimal gauge transformations of the generalised metric and dilaton are given by the generalised Lie derivative with respect to an infinitesimal generalised vector (k^{μ}, ℓ_{μ}) . In the supergravity frame, these generate the spacetime diffeomorphisms

²The $O(d, \mathbb{R}) \times O(d, \mathbb{R})$ forms a local symmetry of the theory, analogous to the Lorentz symmetry of relativity.

³To align with the literature, we use d to mean both the spacetime dimension and the DFT dilaton field, though it should be clear which is implied in each instance.

and gauge transformations (2.5) of the supergravity fields.

2.3.1 Action and equations of motion

One can form an action in the DFT metric and dilaton. The supergravity Lagrangian contains terms up to second order in derivatives, so one also expects two derivatives on the metric H and dilaton d in the DFT Lagrangian⁴. All possible $O(d, d, \mathbb{R})$ covariant terms can be linearly summed over, and the relevant constants fixed on requiring the Lagrangian transforms covariantly under DFT gauge transformations. Meanwhile, the DFT dilaton forms the measure in the action. The action is thus [23, 24]

$$S = \int \mathrm{d}^{2d} X e^{-2d} R, \qquad (2.33)$$

where the $O(d, d, \mathbb{R})$ scalar

$$R = \frac{1}{8} H^{MN} \partial_M H^{KL} \partial_N H_{KL} - \frac{1}{2} H^{MN} \partial_M H^{KL} \partial_K H_{NL}$$

$$+ 4 H^{MN} \partial_M \partial_N d - \partial_M \partial_N H^{MN} - 4 H^{MN} \partial_M d\partial_N d + 4 \partial_M H^{MN} \partial_N d.$$

$$(2.34)$$

We have neglected an additional term (see e.g. [80]) whose contribution vanishes under the strong constraint here and in all further dynamics, but is necessary for the Scherk-Schwarz reduction to gauged supergravity. A treatment of boundary contributions for the DFT action can be found in [75]. It is non-trivial that the action (2.33) reproduces the NS-NS supergravity action (2.1) in the supergravity frame.

We now introduce a novel derivation of the DFT equations of motion. The degrees of freedom are encoded within the metric H and dilaton d. Varying the action with respect to the latter produces

$$\delta_d S = \int \mathrm{d}^{2d} X \left(-2e^{-2d} R \delta d \right) + \text{boundary terms}, \qquad (2.35)$$

yielding the dilaton equation of motion

$$R = 0. \tag{2.36}$$

⁴Higher derivative corrections have been considered in [39, 41, 57].

Meanwhile, varying the action with respect to the metric yields

$$\delta_H S = \int d^{2d} X e^{-2d} \mathcal{K}_{MN} \delta H^{MN} + \text{boundary terms}, \qquad (2.37)$$

where

$$\mathcal{K}_{MN} = \frac{1}{8} \partial_M H^{KL} \partial_N H_{KL} - \frac{1}{2} H^{PK} H^{QL} \partial_L H_{PM} \partial_K H_{NQ}$$

$$+ \frac{1}{4} H^{PQ} H^{KL} \partial_P H_{KM} \partial_Q H_{NL} + 2 \partial_M \partial_N d$$

$$+ (\partial_L - 2(\partial_L d)) \left[H^{LK} \left(\partial_{(M} H_{N)K} - \frac{1}{4} \partial_K H_{MN} \right) \right].$$
(2.38)

However, this does not yield the equations of motion $\mathcal{K}_{MN} = 0$, since $\delta_H S$ is required to vanish only for those δH^{MN} constrained by the $O(d, d, \mathbb{R})/O(d, \mathbb{R}) \times O(d, \mathbb{R})$ coset form of the generalised metric. This was realised by the authors of [24], who derived the full equations of motion by ensuring that the coset form was satisfied through the condition $\eta_{MP} H^{PQ} \eta_{QN} = H_{MN}$. We will take a different route, which while considerably more cumbersome, may be adapted quite naturally to the more demanding case of E_d -covariant extended geometries, where the full equations of motion are largely unknown. We will

- Demand that the metric assumes the coset form (2.30)
- Expand variations of the generalised metric using the chain rule in terms of variations of the field content: the spacetime metric g and two-form B
- Re-express the result in a manifestly duality-covariant form
- Determine the duality-covariant equations of motion which contains precisely the information of the equations of motion for the fields g and B

That is,

$$\mathcal{K}_{MN}\delta H^{MN} = \mathcal{K}_{MN} \left(\frac{\partial H^{MN}}{\partial g_{\mu\nu}} \delta g_{\mu\nu} + \frac{\partial H^{MN}}{\partial B_{\mu\nu}} \delta B_{\mu\nu} \right)$$
(2.39a)

$$= \left[-\mathcal{K}_{\mu\nu}g^{\mu\rho}g^{\sigma\nu} + 2\mathcal{K}_{\mu}{}^{\nu}g^{\mu\rho}g^{\sigma\tau}B_{\tau\nu} + \mathcal{K}^{\mu\nu}\left(\delta^{\rho}\delta^{\sigma} + B_{-}a^{\tau\rho}a^{\sigma\lambda}B_{\lambda}\right) \right]\delta q$$
(2.39b)

$$+ \mathcal{K}^{\nu} \left[\left(\delta^{\rho}_{\mu} \delta^{\rho}_{\nu} + B_{\mu\tau} g^{r} g^{\rho} B_{\lambda\nu} \right) \right] \delta^{\rho}_{\rho\sigma} \\ + \left[-2\mathcal{K}_{\mu}^{\nu} g^{\mu\tau} \delta^{\rho}_{\tau} \delta^{\sigma}_{\nu} - 2\mathcal{K}^{\mu\nu} B_{\mu\tau} g^{\tau\lambda} \delta^{\rho}_{\lambda} \delta^{\sigma}_{\nu} \right] \delta B_{\rho\sigma} \\ = \left[-\mathcal{K}_{\mu\nu} H^{\mu\rho} H^{\sigma\nu} + 2\mathcal{K}_{\mu}^{\nu} H^{\mu\rho} H^{\sigma}_{\nu} + \mathcal{K}^{\mu\nu} \left(\delta^{\rho}_{\mu} \delta^{\sigma}_{\nu} - H_{\mu}^{\rho} H^{\sigma}_{\nu} \right) \right] \delta g_{\rho\sigma}$$

$$(2.39c)$$

$$-\left(\mathcal{K}_{\mu}^{\ \nu}H^{\mu\tau} + \mathcal{K}^{\mu\nu}H_{\mu}^{\ \tau}\right)\left(\delta_{\tau}^{\rho}\delta_{\nu}^{\sigma} - \delta_{\tau}^{\sigma}\delta_{\nu}^{\rho}\right)\delta B_{\rho\sigma}$$
$$-\mathcal{K}_{\tau\tau}\left(n^{K\rho}n^{\sigma L} - H^{K\rho}H^{\sigma L}\right)\delta a \qquad (2.30d)$$

$$= \mathcal{K}_{KL} \left(\eta^{-} \eta^{-} - H^{-} H^{-} \right) \delta g_{\rho\sigma}$$

$$- \mathcal{K}_{KL} \left(H^{KP} \eta_{PM} \eta^{LN} - H^{KP} \delta^{N}_{P} \delta^{L}_{M} \right) \eta^{M\rho} \delta^{\sigma}_{N} \delta B_{\rho\sigma}$$

$$(2.39d)$$

$$= P_{MN}{}^{KL}\mathcal{K}_{KL}\left(\eta^{M\rho}\eta^{\sigma N}\delta g_{\rho\sigma} + \eta^{M\rho}H^{\sigma N}\delta B_{\rho\sigma}\right), \qquad (2.39e)$$

where

$$P_{MN}{}^{KL} = \delta^K_M \delta^L_N - H_{MP} \eta^{PK} \eta_{NQ} H^{QL}.$$
(2.40)

Thus,

$$P_{MN}{}^{KL}\mathcal{K}_{KL} = 0 \tag{2.41}$$

are sufficient covariant equations of motion which correspond to minima of the action (2.33). That they are necessary (i.e. that they do not over-constrain g and B), can be seen from a simple degree of freedom counting. We do this for vanishing two-form field and a Minkowski metric. We find $P_{MN}{}^{KL}$ has a kernel on symmetric objects J_{KL} of dimension d(d + 1) which, of the 2d(2d + 1)/2 degrees of freedom in the symmetric \mathcal{K}_{MN} , leaves d^2 degrees of freedom, i.e. (2.41) contains d^2 equations of motion. Alongside the 1 dilaton equation of motion (2.36), this corresponds to the d(d + 1)/2 + d(d - 1)/2 + 1 degrees of freedom in the supergravity fields⁵ (g, B, ϕ) . Indeed, the equations of motion (2.36), (2.41) are in agreement with those calculated in [24] up to contributions which vanish under the section condition.

 $^{^{5}}$ We note that this degrees of freedom counting, while highly suggestive, is not strictly a full proof that (2.41) are the covariant DFT metric equations of motion. Specifically, we have not shown that the variation of the spacetime metric and two-form do not lie within the kernel of the projector.

2.4 A double field theory wave solution recovers the string

We seek a generalised metric solution for the DFT equations of motion corresponding to a null wave whose momentum is pointing the \tilde{x}_d direction. The ansatz will be analogous to that of a pp-wave in general relativity [203]. That this is a solution of the DFT equations of motion does not follow obviously from the existence of the spacetime pp-wave. As we have seen, the equations of motion of the generalised metric in DFT are certainly not the same as the equations of motion of the metric in relativity. Let us remove any source of confusion. The pp-wave as a solution for g may of course, by construction, be embedded as a solution in DFT by simply inserting the pp-wave solution for g and B into H. However, here we will consider a pp-wave analogue (that is the usual pp-wave ansatz of [203]) not for g but for the DFT metric H itself and then determine its interpretation in terms of the usual metric g and two-form B.

With generalised coordinates written

$$X^{M} = (x^{\mu}, \tilde{x}_{\mu}) = (t, y^{m}, z, \tilde{t}, \tilde{y}_{m}, \tilde{z}), \qquad (2.42)$$

the following metric

$$H_{MN} dX^{M} dX^{N} = (K-2)(dt^{2} - dz^{2}) + \delta_{mn} dy^{m} dy^{n}$$

$$+ 2(K-1)(dt d\tilde{z} + d\tilde{t} dz)$$

$$- K(d\tilde{t}^{2} - d\tilde{z}^{2}) + \delta^{mn} d\tilde{y}_{m} d\tilde{y}_{n}$$

$$(2.43)$$

and constant dilaton d constitute a solution to the DFT equations of motion (2.36), (2.41) provided K satisfies certain conditions we will shortly discuss.

Since it is of the same form as the usual pp-wave solution, the natural interpretation is of a pp-wave in the doubled geometry. One can therefore imagine it propagates with momentum in the \tilde{z} direction. To determine whether it truly carries momentum would require the construction of conserved charges in DFT, which is yet to be done. It would be useful to consider objects like generalised Komar integrals, and other ways one defines charges in general relativity, but now for DFT. Nevertheless, we shall proceed with our interpretation.

From a DFT perspective, it is required (at least naively) that K satisfies the section condition. We take K to be a harmonic function of the usual transverse

coordinates y^m (but not of their duals \tilde{y}_m):

$$K = K(y^m), \qquad {}^y \Box K = 0, \qquad \text{where} \quad {}^y \Box \equiv \delta^{mn} \partial_m \partial_n.$$
 (2.44)

This is sufficient for the solution to satisfy the equations of motion. For illustrative purposes, we will adopt the explicit form

$$K = 1 + \frac{k_0}{|y|^{d-4}}, \qquad |y|^2 = y^m y^n \delta_{mn}, \tag{2.45}$$

where k_0 is a constant. In particular, this gives an asymptotically flat background metric. We note that the precise form of K does not affect the Goldstone mode calculations appearing later in §2.4.1. That H depends only on the y^m and not their duals implies that the wave solution is smeared in the \tilde{y}_m directions. One can think of it as a plane wave front extending along the dual \tilde{y}_m directions but with momentum in the \tilde{z} direction.

The generalised metric as (2.30) corresponds to a Kaluza-Klein ansatz from which we can determine the supergravity metric and two-form:

$$H_{MN} \mathrm{d}X^M \mathrm{d}X^N = (g_{\mu\nu} - B_{\mu\rho}g^{\rho\sigma}B_{\sigma\nu})\mathrm{d}x^\mu\mathrm{d}x^\nu + 2B_{\mu\rho}g^{\rho\nu}\mathrm{d}x^\mu\mathrm{d}\tilde{x}_\nu + g^{\mu\nu}\mathrm{d}\tilde{x}_\mu\mathrm{d}\tilde{x}_\nu,$$
(2.46)

while we can determine the dilaton ϕ from (2.32). If we choose the spacetime coordinates to be the subset $x^{\mu} = (t, y^m, \tilde{z})$, the DFT wave (2.43) reduces via (2.46) to the *spacetime* pp-wave solution

$$g_{\mu\nu}dx^{\mu}x^{\nu} = (K-2)dt^{2} + 2(K-1)dtd\tilde{z} + Kd\tilde{z}^{2} + \delta_{mn}dy^{m}dy^{n}$$

$$B_{\mu\nu} = 0 \qquad e^{-2\phi} = e^{-2d},$$
 (2.47)

with momentum along the \tilde{z} direction. Alternatively, with spacetime coordinates $x^{\mu} = (t, y^m, z)$, we find

$$g_{\mu\nu}dx^{\mu}dx^{\nu} = -K^{-1}(dt^2 - dz^2) + \delta_{mn}dy^m dy^n \qquad e^{-2(\phi-d)} = K$$

$$B_{tz} = 1 - K^{-1} \qquad B_{\mu n} = 0,$$
(2.48)

which is the fundamental string solution extended along the z direction [204]. We have thus shown that the solution (2.43) which carries momentum in the \tilde{z} direction in the doubled space corresponds in the supergravity frame to a string extended along the z direction from the spacetime perspective, and a spacetime pp-wave with momentum in the \tilde{z} direction in a T-dual frame.

Of course this is no surprise from the point of view of T-duality. Momentum and string winding exchange under T-duality. It is precisely as expected that momentum in the dual direction corresponds to a string. What is more surprising is when one views this from the true DFT perspective. There are null wave solutions that can point in any direction. When we analyse these null waves from the reduced theory we see them as fundamental strings in the supergravity frame, or as pp-waves from a dual direction. It is a simple rotation of direction of propagation that takes one solution into the other. This is duality from the DFT perspective.

Moreover, in the doubled formalism the solution is a massless wave with $\mathcal{P}_M \mathcal{P}_N H^{MN} = 0$ (where the \mathcal{P}^M are some generalised momenta), while from the spacetime perspective in the supergravity frame the string has a tension T and charge q which are given by the momenta in the dual directions with a resulting BPS equation T = |q|.

2.4.1 Goldstone modes of the wave solution

In the previous section we presented a solution to the equations of motion of DFT which reduces in the supergravity frame to the fundamental string. It will be interesting to analyse the Goldstone modes of this DFT solution in the supergravity frame. Especially since the advent of M-theory, it was understood that branes are dynamical objects and that when one finds a solution of the low energy effective action one can learn about the theory by examining the dynamics of the Goldstone modes. For D-branes in string theory this was done in [205] and for the membrane and fivebrane in M-theory, where such an analysis was really the only way of describing brane dynamics, this was done in [205, 206]. We will follow the excellent exposition and the method described in [205] as closely as possible.

We will consider small perturbations h_{MN} and λ of the generalised metric and dilaton respectively, each generated by the generalised Lie derivative (2.25) with respect to a generalised vector ζ^M ,

$$h_{MN} = (L_{\zeta}H)_{MN} \tag{2.49}$$

$$\lambda = L_{\zeta} d = \zeta^M \partial_M d - \frac{1}{2} \partial_M \zeta^M.$$
(2.50)

Note that (2.50) contains a weight term for the dilaton (whose exponential e^{-2d} has weight 1).

The wave solutions are extended objects and therefore sweep out a worldvolume in space, spanned by the coordinates (t, z). The solution clearly breaks translation symmetry and so one naturally expects worldvolume-scalar zero-modes. One immediate puzzle would be to ask about the number of degrees of freedom of the Goldstone modes. Given that the space is now doubled one might imagine that any solution which may be interpreted as a string would have 2d - 2 degrees of freedom rather than the expected d - 2. We will answer this question and show how the Goldstone modes have the correct number of degrees of freedom despite the solution living in a 2d dimensional space. The projected form of the equations of motion are crucial in making this work out.

To carry out the analysis it will be useful to split up the space into parts longitudinal and transverse to the string. We will do this using an alternative coordinate notation for this section $X^M = (x^{\mu}, \tilde{x}^{\bar{\mu}})$, so as not to confuse the inverse metric and dual coordinates. We collect the coordinates into $x^a = (t, z)$, $\tilde{x}^{\bar{a}} = (\tilde{t}, \tilde{z})$ such that the generalised coordinates are $X^M = (x^a, y^m, \tilde{x}^{\bar{a}}, \tilde{y}^{\bar{m}})$. We will refer to the $(\tilde{x}^{\bar{a}})$ x^a as the (dual) worldsheet coordinates, and the $(\tilde{y}^{\bar{m}}) y^m$ as the (dual) transverse coordinates. In this notation, the non-zero components of the metric and its inverse are

$$H_{ab} = (2 - K)\mathbb{I}_{ab} \qquad H^{ab} = K\mathbb{I}^{ab}$$

$$H_{\bar{a}\bar{b}} = K\mathbb{I}_{\bar{a}\bar{b}} \qquad H^{\bar{a}\bar{b}} = (2 - K)\mathbb{I}^{\bar{a}\bar{b}}$$

$$H_{a\bar{b}} = (K - 1)\mathbb{J}_{a\bar{b}} \qquad H^{a\bar{b}} = (K - 1)\mathbb{J}^{a\bar{b}} \qquad (2.51)$$

$$H_{mn} = \delta_{mn} \qquad H^{mn} = \delta^{mn}$$

$$H_{\bar{m}\bar{n}} = \delta_{\bar{m}\bar{n}} \qquad H^{\bar{m}\bar{n}} = \delta^{\bar{m}\bar{n}}$$

where $\mathbb{I}_{ab} = \mathbb{I}^{ab}$ $(\mathbb{I}_{\bar{a}\bar{b}} = \mathbb{I}^{\bar{a}\bar{b}})$ is the Minkowski space metric $\mathbb{I} = \text{diag}(-1, 1)$ in the (dual) worldsheet coordinates, and the 2 × 2 matrix

$$\mathbb{J} = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}. \tag{2.52}$$

We now choose a transformation parameter ζ^M with components along the transverse directions (and their duals),

$$\zeta^M = (0, K^\alpha \varphi^m, 0, K^\beta \tilde{\varphi}^{\bar{m}}). \tag{2.53}$$

Here, φ^m and $\tilde{\varphi}^{\bar{m}}$ are coordinate-independent vectors that will become the Goldstone modes once we allow them to depend on the worldsheet coordinates, and α and β are

constants that will be determined by demanding that the Goldstone modes become normalisable. From (2.49), the *non-zero* components of the metric perturbation are

$$h_{ab} = -\varphi^{m} K^{\alpha} \partial_{m} K \mathbb{I}_{ab} \quad h_{mn} = 2\varphi^{q} \delta_{q(m} \delta_{n)}{}^{p} \partial_{p} K^{\alpha}$$

$$h_{\bar{a}\bar{b}} = \varphi^{m} K^{\alpha} \partial_{m} K \mathbb{I}_{\bar{a}\bar{b}} \qquad h_{\bar{m}\bar{n}} = -2\varphi^{q} \delta_{q(\bar{m}} \delta_{\bar{n}}){}^{p} \partial_{p} K^{\alpha} \qquad (2.54)$$

$$h_{a\bar{b}} = \varphi^{m} K^{\alpha} \partial_{m} K \mathbb{J}_{a\bar{b}} \qquad h_{m\bar{n}} = -2\tilde{\varphi}^{\bar{q}} \delta_{\bar{q}[m} \delta_{\bar{n}}]{}^{p} \partial_{p} K^{\beta},$$

while the dilaton perturbation (2.50) is

$$\lambda = -\frac{1}{2}\varphi^m \partial_m K^\alpha. \tag{2.55}$$

We now promote the moduli to depend on the worldsheet coordinates (but not their duals, so as to satisfy the section condition),

$$\varphi^m \to \varphi^m(x^a), \qquad \tilde{\varphi}^{\bar{m}} \to \tilde{\varphi}^{\bar{m}}(x^a).$$
 (2.56)

These are the Goldstone modes, which are really the normalisable modes corresponding to broken gauge symmetry. Whereas for gravity gauge transformations are ordinary diffeomorphisms, in the case of DFT they are generated by the generalised Lie derivative.

The equations of motion of the modes are determined by those of the perturbed DFT metric H + h and dilaton $d + \lambda$. We will determine them to linear order in h and λ (it would certainly be interesting to move beyond this expansion and compare with a Nambu-Goto type action but we will not do so here). In approaching the equations of motion to linear order, we will simplify our calculation with the fact that both the background and gauge transformations of the background (which were calculated for *constant* φ^m , $\tilde{\varphi}^{\bar{m}}$) will satisfy the equations of motion. Thus, terms will only contribute to the equations of motion when there are x^a derivatives acting on modes.

For the metric equations of motion, we note that the background satisfies $\mathcal{K} = 0$. Therefore, in the linear order equations of motion, \mathcal{K} will only contain terms linear in the modes, while P will contain no contribution from the modes—it will be given purely by the background metric. With the projector only in terms of the background metric, the equations of motion automatically satisfy

$$(P\mathcal{K})_{MN} = -H_{MK}Y^{KP}{}_{NL}H^{LQ}(P\mathcal{K})_{PQ}, \qquad (2.57)$$

which we can use to determine the independent components, which are as follows.

We begin with the transverse-worldsheet cross components (and their duals), $(P\mathcal{K})_{ma}$ and $(P\mathcal{K})_{m\bar{a}}$, where

$$(P\mathcal{K})_{ma} + \mathbb{I}_{ab} \mathbb{J}^{b\bar{a}} (P\mathcal{K})_{m\bar{a}} = -\frac{1}{2} \left(\delta^p_m \delta^r_q - \delta_{mq} \delta^{pr} \right) \\ \times \left[\partial_a \varphi^q \partial_r \partial_p K^\alpha - \mathbb{I}_{ab} \epsilon^{bc} \partial_c \tilde{\varphi}^{\bar{q}} \delta^{\bar{q}}_{\bar{q}} \partial_r \partial_p K^\beta \right],$$

$$(2.58)$$

with antisymmetric symbol ϵ^{ab} (see app. A). So as to obtain an equation of motion for the modes independent of the background function K, we choose $\alpha = \beta$, where the above vanishes if the modes satisfy a duality relation

$$\partial_a \varphi^q = \mathbb{I}_{ab} \epsilon^{bc} \partial_c \tilde{\varphi}^{\bar{q}} \delta^q_{\bar{q}}, \qquad (2.59)$$

which in turn implies the wave equations

$$\Box \varphi^q = \Box \varphi^{\bar{q}} \delta^q_{\bar{q}} = 0, \quad \text{where} \quad \Box \equiv \mathbb{I}^{ab} \partial_a \partial_b. \tag{2.60}$$

Employing $\alpha = \beta$ and the duality relation (2.59), we also have the contribution

$$\mathbb{I}_{ab}\mathbb{J}^{b\bar{a}}(P\mathcal{K})_{m\bar{a}} = \frac{1}{2} \left(\delta^p_m \delta^r_q - \delta_{mq} \delta^{pr} \right) \left[\partial_a \varphi^q \partial_r (K^\alpha \partial_p K) + \mathbb{I}_{ab} \epsilon^{bc} \partial_c \tilde{\varphi}^{\bar{q}} \delta^q_{\bar{q}} \partial_r (K \partial_p K^\beta) \right] \\
= \frac{1}{2} (\alpha + 1) \left(\delta^p_m \delta^r_q - \delta_{mq} \delta^{pr} \right) \partial_a \varphi^q \partial_r (K^\alpha \partial_p K),$$
(2.61)

which vanishes for $\alpha = -1$. Indeed, this is the value derived in [205] for the Goldstone dynamics of branes. In particular, for the D3, M2 and M5 brane, $\alpha = -1$ provided normalisable modes. The independent transverse and worldsheet components, along with the scalar R of the dilaton equation (2.36) are

$$(P\mathcal{K})_{mn} = -K \Box \varphi^q \delta_{q(m} \partial_{n)} K^\alpha \tag{2.62a}$$

$$(P\mathcal{K})_{m\bar{n}} = K \Box \tilde{\varphi}^{\bar{q}} \delta_{\bar{q}[m} \delta^{p}_{\bar{n}]} \partial_{p} K^{\beta}$$
(2.62b)

$$(P\mathcal{K})_{ab} = f_1(K,\alpha)\mathbb{I}_{ab}\Box\varphi^q\partial_q K + (\alpha+1)(K-2)K^\alpha\partial_a\partial_b\varphi^q\partial_q K \qquad (2.62c)$$

$$\mathbb{I}_{bc}\mathbb{J}^{cb}(P\mathcal{K})_{a\bar{b}} = f_2(K,\alpha)\mathbb{I}_{ab}\Box\varphi^q\partial_q K + (\alpha+1)(1-K)K^\alpha\partial_a\partial_b\varphi^q\partial_q K \qquad (2.62d)$$

$$R = -(2\alpha + 1)\Box\varphi^q K^\alpha \partial_q K, \qquad (2.62e)$$

where f_1 and f_2 are functions whose form is unimportant for this analysis. These all vanish by means of the wave equations (2.60) and $\alpha = -1$.

Returning to the duality relation (2.59), we can combine the Goldstone modes into a generalised vector $\Phi^M = (0, \varphi^m, 0, \tilde{\varphi}^{\bar{m}})$, where we find

$$H_{MN}\partial_a \Phi^M = \eta_{MN} \mathbb{I}_{ab} \epsilon^{bc} \partial_c \Phi^N, \qquad (2.63)$$

in agreement with Duff's result in [10] for the duality symmetric string. Alternatively, expressed in terms of the (anti-)chiral combinations

$$\psi^m_{\pm} = \varphi^m \pm \tilde{\varphi}^{\bar{n}} \delta^m_{\bar{n}}, \qquad (2.64)$$

equation (2.59) describes the familiar (anti-)self-dual left- and right-movers

$$\partial_a \psi^m_{\pm} = \pm \mathbb{I}_{ab} \epsilon^{bc} \partial_c \psi^m_{\pm} \tag{2.65}$$

of the Tseytlin-string [11,12]. Thus the dynamics of the Goldstone modes of the wave solution reproduce the duality symmetric string in doubled space. The number of physical degrees of freedom are not doubled but just become rearranged in terms of chiral and anti-chiral modes on the worldsheet.

2.4.2 Comparison with the σ -model evaluated in the string or wave background

The equations of motion derived in the previous section recover the equations of motion of the Tseytlin string. A natural question would be to ask what background is the string in? Is the target space of the doubled solution the combination of the fundamental string with the spacetime pp-wave background? The answer to this question can be seen immediately from the Goldstone mode analysis which gives the equations of motion of the free string i.e. that of the σ -model in a flat background.

To understand this it is worth understanding what the Goldstone mode analysis provides us with in other cases where this has been carried out in a more conventional setting. In [205] the Goldstone mode analysis for the D3, M2 and M5 branes was completed and used to determine the effective equations of motion for each of those objects. In each case the analysis gave the description of the brane in a flat background—as one would expect—the Goldstone mode analysis must give the equations of motion of the brane in a flat background since the solution for which one is determining the moduli is that of the brane in a flat background. The same argument applies here—the string dynamics must be those of the string in a flat background.

A string solution in the background of other strings i.e. a string σ -model in a string background would be a different solution and as such obey a different set of equations of motion. Describing this more technically, to find the σ -model in a non-trivial background one must find the back-reacted wave solution not for asymptotically flat space but for one with asymptotically switched on NS-fluxes and then determine its moduli and their equations of motion. Once one has determined the effective equations of motion through a Goldstone mode analysis, one can then proceed to covariantize these equations (in terms of the geometry of moduli space) to determine the general equations of motion. In terms of the doubled string above, this would imply just replacing the flat target space generalised metric with the generalised metric of an arbitrary background.

2.5 Buscher-type transformation in the double field theory σ -model

We have seen in §2.2 how Buscher recovered T-duality transformations from the string σ -model with an isometry in the target space. We have also seen in §2.4 how string dynamics can be approached in an $O(d, d, \mathbb{R})$ -covariant manner via DFT. In this section, we look at manifestly $O(d, d, \mathbb{R})$ -covariant string σ -models in DFT, and apply a Buscher procedure to these. A Floreanini-Jackiw-style [207] string σ -model was derived for $O(d, d, \mathbb{R})$ -invariant field theory initially in [11, 12], which takes the form

$$S = \frac{1}{2} \int d^2 \sigma (-H_{MN} \partial_1 X^M \partial_1 X^N + \eta_{MN} \partial_1 X^M \partial_0 X^N), \qquad (2.66)$$

where indices on the partial derivatives correspond to worldsheet directions (as opposed to the doubled target space indices M, N...). This is not, however, manifestly worldsheet covariant—covariance is recovered on-shell. A manifestly worldsheetcovariant action was proposed by Hull [58] (which has been shown to be equivalent to (2.66) [66], see also [61,68]), which can be gauged by a derivative-index-valued connection $A_M \sim \psi \partial_M \vartheta$, to produce the gauged action of [60]

$$S = \int d^2 \sigma \frac{1}{2} H_{MN} D_a X^M D_b X^N h^{ab} - \epsilon^{ab} D_a X^M A_{bM}.$$
 (2.67)

Here, $D_a X^M = \partial_a X^M - A_a^M$, and target space indices are here raised and lowered with η . We will apply the Buscher procedure with one Lagrange multiplier to the

ungauged version of the covariant action (2.67), and with two Lagrange multipliers to the non-covariant action (2.66). For both cases, we will split the coordinates as $\mu = (0, m)$, where x^0 need not be the time coordinate.

A single Buscher transformation

The action

$$S' = \frac{1}{2} \int d^2 \sigma \left[(H_{00}C_aC_b + 2H_{0m}C_a\partial_bX^m + H_{mn}\partial_aX^m\partial_bX^n + 2H_0^{\nu}C_a\partial_bX_{\nu} + 2H_m^{\nu}\partial_aX^m\partial_bX_{\nu} + H^{\mu\nu}\partial_aX_{\mu}\partial_bX_{\nu})h^{ab} + \alpha\partial_aC_b\epsilon^{ab} \right]$$
(2.68)

has equations of motion for the Lagrange multiplier α

$$\partial_{[a}C_{b]} = 0, \tag{2.69}$$

which on identifying $C_a = \partial_a X^0$ recovers the covariant action (2.67) for $A_M = 0$. If instead we integrate by parts the Lagrange multiplier term, we find (up to boundary contributions)

$$S' = \frac{1}{2} \int d^2 \sigma \left[(H_{00}C_aC_b + 2H_{0m}C_a\partial_b X^m + H_{mn}\partial_a X^m\partial_b X^n + 2H_0^{\nu}C_a\partial_b X_{\nu} + 2H_m^{\nu}\partial_a X^m\partial_b X_{\nu} + H^{\mu\nu}\partial_a X_{\mu}\partial_b X_{\nu})h^{ab} - \partial_a \alpha C_b \epsilon^{ab} \right],$$
(2.70)

which has equations of motion

$$h^{ab} \left(2H_{00}C_b + 2H_{0m}\partial_b X^m + 2H_0{}^{\nu}\partial_b X_{\nu} \right) - \epsilon^{ab}\partial_b \alpha = 0.$$
 (2.71)

If we insert this into the action (2.70), we find

$$S' = \int d^2 \sigma \frac{1}{2} H'_{MN} \partial_a (X')^M \partial_b (X')^N h^{ab} + \epsilon^{ab} \partial_a (X')^M \partial_b (X')^0 \frac{H_{0M}}{H_{00}}, \qquad (2.72)$$

where $(X')^M = (\alpha/2, X^m, X_\mu)$ and

$$H'_{00} = \frac{1}{H_{00}} \quad H'_{mn} = H_{mn} - \frac{H_{0m}H_{0n}}{H_{00}}$$

$$H'_{0n} = 0 \qquad H'_{m}{}^{\nu} = H_{m}{}^{\nu} - \frac{H_{0m}H_{0}{}^{\nu}}{H_{00}}$$

$$H'_{0}{}^{\mu} = 0 \qquad H'^{\mu\nu} = H^{\mu\nu} - \frac{H_{0}{}^{\mu}H_{0}{}^{\nu}}{H_{00}}.$$
(2.73)

While it may appear that (2.72) takes the form of the gauged σ -model (2.67), we recall that A_M is required to be derivative-index-valued. With this in mind, we do not find that (2.72) corresponds to a gauged DFT σ -model. That is, we find no solution-generating transformations for one Lagrange multiplier.

A double Buscher transformation

We now extend the analysis to include an additional Lagrange multiplier term in the dual coordinates, but here for the action (2.66). Consider the action

$$S' = \frac{1}{2} \int d^2 \sigma (-H_{00} A_1^2 - 2H_{0m} A_1 \partial_1 X^m - 2H_0^0 A_1 B_1 - 2H_0^m A_1 \partial_1 X_m - H_{mn} \partial_1 X^m \partial_1 X^n - 2H_m^0 \partial_1 X^m B_1 - H^{mn} \partial_1 X_m \partial_1 X_n - 2H_m^n \partial_1 X^m \partial_1 X_n - H^{00} B_1^2 - 2H^{0m} B_1 \partial_1 X_m + A_1 B_0 + A_0 B_1 + \partial_1 X^m \partial_0 X_m + \partial_1 X_m \partial_0 X^m - \alpha (\partial_1 A_0 - \partial_0 A_1) - \beta (\partial_0 B_1 - \partial_1 B_0)).$$
(2.74)

The Lagrange multipliers α and β have resulting equations of motion

$$\partial_1 A_0 - \partial_0 A_1 = 0 \qquad \partial_0 B_1 - \partial_1 B_0 = 0,$$
 (2.75)

solutions of which are $A_a = \partial_a C$, $B_a = \partial_a D$. When we reinsert these solutions in to the action (2.74) we recover the action (2.66) with the identification $C = X^0$, $D = X_0$.

If instead we integrate (2.74) by parts so that the derivatives act on the Lagrange multipliers, we find, up to total derivatives,

$$S' = \frac{1}{2} \int d^2 \sigma (-H_{00} A_1^2 - 2H_{0m} A_1 \partial_1 X^m - 2H_0^0 A_1 B_1 - 2H_0^m A_1 \partial_1 X_m \qquad (2.76)$$
$$-H_{mn} \partial_1 X^m \partial_1 X^n - 2H_m^0 \partial_1 X^m B_1 - H^{mn} \partial_1 X_m \partial_1 X_n$$
$$-2H_m^n \partial_1 X^m \partial_1 X^n - H^{00} B_1^2 - 2H^{0m} B_1 \partial_1 X_m + A_1 B_0 + A_0 B_1$$
$$+ \partial_1 X^m \partial_1 X_m + \partial_1 X_m \partial_0 X^m + A_0 \partial_1 \alpha - A_1 \partial_0 \alpha - B_1 \partial_0 \beta + B_0 \partial_1 \beta).$$

The equations of motion for A_a and B_a are now

$$A_1 + \partial_1 \beta = 0 \qquad (2.77a)$$

$$-2H_0^0 A_1 - 2H_m^0 \partial_1 X^m - 2H^{00} B_1 - 2H^{0m} \partial_1 X_m + A_0 - \partial_0 \beta = 0 \qquad (2.77b)$$

$$B_1 + \partial_1 \alpha = 0 \qquad (2.77c)$$

$$-2H_{00}A_1 - 2H_{0m}\partial_1 X^m - 2H_0{}^0B_1 - 2H_0{}^m\partial_1 X_m + B_0 - \partial_0\alpha = 0.$$
(2.77d)

Inserting these into (2.76), we recover the action

$$S' = \frac{1}{2} \int d^2 \sigma (-H_{MN} \partial_1 (X')^M \partial_1 (X')^N + \eta_{MN} \partial_1 (X')^M \partial_0 (X')^N),$$

where $(X')^M = (-\beta, X^m, -\alpha, X_m).$ (2.78)

We observe that the Lagrange multiplier α for the A equation of motion (which was used to replace X^0) becomes the dual coordinate X_0 , likewise for β and B. That is, the transformation sends $x^0 \leftrightarrow \tilde{x}_0$. One can see this is the T-duality invariance of the theory manifesting if instead we equivalently write

$$S' = \frac{1}{2} \int d^2 \sigma (-H'_{MN} \partial_1 (X')^M \partial_1 (X')^N + \eta_{MN} \partial_1 (X')^M \partial_0 (X')^N),$$

where $(X')^M = (-\alpha, X^m, -\beta, X_m)$ (2.79)

where we find the corresponding spacetime metric and two-form field transformations are given by the Buscher rules (2.12).
Chapter 3

U-duality invariant M-theory

The strong coupling limit of type IIA string theory is an eleven dimensional theory, termed M-theory, which has as its fundamental objects not strings, but membranes and five-branes. Various compactification limits of M-theory yield all five string theories, while T-duality and S-duality are united as part of a larger web of U-dualities. Though our understanding of the full theory remains rather incomplete, we can look to the low energy limit, eleven dimensional supergravity, discovered by Cremmer, Julia, and Scherk [4], and whose action in the bosonic sector is given by

$$S_{11} = \frac{1}{2\kappa^2} \int_{M_{11}} \sqrt{\det(g)} \left(R - \frac{1}{48} F_4^2 \right) - \frac{1}{4} F_4 \wedge F_4 \wedge C_3, \tag{3.1}$$

where $F_4 = dC_3$ is the field strength of the three-form gauge field.

In 1980, Julia [86] demonstrated that compactifications of eleven dimensional supergravity on T^n exhibit the E_n symmetries listed in table 3.1. These symmetries are inherited from the U-duality group of M-theory. We will here on use the name U-duality interchangeably to mean both the continuous hidden symmetry group of supergravity and the parent M-theory duality group.

It has been proposed that these U-duality groups are not only present on compactification, but are symmetries of the full M-theory [208]—when compactifying on T^n , an E_n subset of the full E_{11} symmetry group is revealed, acting on the compactified directions. In the supergravity theory, the symmetries act on the metric and gauge fields on equal footing, and so one may ask whether there is some way of unifying these objects in a manifestly duality-invariant form. However, since the symmetry is not present in the supergravity theory as it stands, some kind of extension is required. This is the role of the exceptional geometries.

11 - n	n	G	H
10	1	SO(1,1)	1
9	2	SL(2)	SO(2)
8	3	$SL(3) \times SL(2)$	$SO(3) \times SO(2)$
7	4	SL(5)	SO(5)
6	5	SO(5,5)	$SO(5) \times SO(5)$
5	6	E_6	USp(8)
4	$\overline{7}$	E_7	SU(8)
3	8	E_8	SO(16)

Table 3.1: The symmetry groups G and their maximally compact subgroups H of 11-dimensional supergravity compactified on T^n .

n	$G = E_d$	Tangent space	D
4	SL(5)	$TM\oplus\wedge^2T^*M$	10
5	Spin(5,5)	$TM\oplus\wedge^2T^*M\oplus\wedge^5T^*M$	16
6	E_6	$TM \oplus \wedge^2 T^*M \oplus \wedge^5 T^*M$	27
7	E_7	$TM \oplus \wedge^2 T^*M \oplus \wedge^5 T^*M \oplus \wedge^6 T^*M$	56

Table 3.2: The tangent space and coordinate representation D of E_d duality-invariant extended geometries

Recall that in $O(d, d, \mathbb{R})$ -invariant generalised geometry, the duality group of compactified NS-NS supergravity is made manifest without compactification by extending the tangent space to include one-forms which can parametrise gauge transformations of the two-form field. Analogously, Hull [209], and later Pacheco and Waldram [210] made manifest the E_d duality group in *exceptional generalised geometry* by extending the tangent space to include forms which can parametrise gauge transformations of the form field content inherited from the parent membrane theory. The tangent spaces are given for d = 4-7 in table 3.2. One can think of the tangent space contributions as arising with three-form gauge transformations ($\in \wedge^2 T^*M$), fivebrane-modes ($\in \wedge^5 T^*M$) for $d \ge 5$, and six-form (dual to C_3) gauge transformations ($\in \wedge^6 T^*M$) for $d \ge 7$. Meanwhile, the form fields and metric are unified into a G/H coset representative which encodes precisely the

$$d(d+1)/2 + {d \choose 3} + {d \choose 6} = \dim(G/H)$$
 (3.2)

degrees of freedom in g, C_3 and C_6 (these have been constructed for d = 3, 4 [94], d = 5-7 [97] and d = 8 [211]).

As DFT is to generalised geometry, where one extends not just the tangent space, but the coordinate space itself, [36,75,94–103] constructed extended geometries where the d spacetime dimensions are supplemented with w "wrapping" coordinates which in the supergravity picture correspond to membrane charges. The E_d duality group now acts as a manifest symmetry of a (d + w)-dimensional space. Again, there has been no compactification, unlike in the standard torus compactification of elevendimensional supergravity. That E_d can act on the space is precisely due to the presence of the additional wrapping coordinates¹. The tangent spaces of the extended geometries are the same as those of their corresponding exceptional generalised geometry, given in table 3.2. Again, the crucial difference between generalised and extended geometries is that, in the latter, one allows dependence of all the fields on the additional coordinates, while imposing a section condition which determines the physical subset. Different solutions to this section condition are related by E_d rotations.

Extended geometry is in fact a truncation of exceptional field theory [104–119]. There, the eleven-dimensional supergravity coordinates are split 11 = (11 - d) + dwithout dimensional reduction, and then supplemented with the w wrapping coordinates. While the extended geometries can be seen as (a truncation of) an elevendimensional theory which exhibits manifest $E_d \times GL(11 - d, \mathbb{R})$ invariance, and where the internal and external sectors are decoupled, in exceptional field theory these sectors are no longer decoupled.

In this chapter, we focus on the four-dimensional SL(5)-invariant extended geometry constructed in [94–96, 98, 99], whose corresponding four-dimensional supergravity has action (3.1) (short of the Chern-Simons terms which can only be defined in eleven dimensions). The picture we illustrate in the next section is similar to that in d = 5-7.

3.1 The SL(5)-invariant theory

In SL(5)-invariant M-theory, one supplements the four-dimensional spacetime coordinates with the six coordinates dual to wrapping configurations of M2-branes in four dimensions. These can be combined into an SL(5) coordinate on the full space

¹More accurately, the extended geometries exhibit manifest $E_d(\mathbb{R})$ covariance, while compactification on T^r reveals an $E_r(\mathbb{Z})$ symmetry group, analogous to how $O(d, d, \mathbb{R})$ -covariant DFT compactified on T^r reveals an $O(r, r, \mathbb{Z})$ T-duality group.

via the decomposition²

$$X^{M} = X^{[mm']} = \begin{cases} X^{\mu 5} = x^{\mu} \\ X^{\mu \nu} = \frac{1}{2} \epsilon^{\mu \nu \rho \sigma 5} y_{\rho \sigma} \end{cases}$$
(3.3)

where $\mu \in \{1, \ldots, 4\}$. Here, capital Roman indices run $M \in \{1, \ldots, 10\}$ and correspond to an antisymmetric pair of later-alphabet lower Roman indices $m, \ldots, z \in \{1, \ldots, 5\}$. Fields are allowed to depend on all X^M .

For the SL(5) geometry, the Y-tensor takes the form

$$Y^{MN}{}_{KL} = \frac{1}{4} \epsilon^{aMN} \epsilon_{aKL}, \qquad (3.4)$$

where $\epsilon^{aMN} = \epsilon^{amm'nn'}$ is the five-dimensional antisymmetric symbol. In terms of this Y-tensor, the reduction to the physical set of coordinates is once again achieved by a section condition of the form (2.23) with (2.29). This additionally guarantees closure of gauge transformations on the extended space, which are given by the corresponding generalised Lie derivative (2.25) (in this comparison, indices $M, N \dots$ should of course be understood here as those of the SL(5) representation (3.3)). Solving the section condition by demanding independence from the dual coordinates $y_{\rho\sigma}$ recovers the supergravity frame, while other solutions are SL(5) rotations of this. One may relax the section condition to allow Scherk-Schwarz reductions to gauged supergravities [85, 120, 121].

The metric and form field C_3 are unified into the 10×10 symmetric coset generalised metric

$$M_{MN} = g^{1/5} \begin{pmatrix} g_{\mu\nu} + \frac{1}{2} C_{\mu}^{\rho\sigma} C_{\nu\rho\sigma} & -\frac{1}{2\sqrt{2}} C_{\mu}^{\rho\sigma} \epsilon_{\rho\sigma\lambda\lambda'} \\ -\frac{1}{2\sqrt{2}} C_{\nu}^{\rho\sigma} \epsilon_{\rho\sigma\eta\eta'} & (\det g)^{-1} g_{\eta\eta',\lambda\lambda'} \end{pmatrix},$$
(3.5)

where $g_{\eta\eta',\lambda\lambda'} \equiv g_{\eta[\lambda}g_{\lambda']\eta'}$ (also $g^{\eta\eta',\lambda\lambda'} \equiv g^{\eta[\lambda}g^{\lambda']\eta'}$) and $C = C_3$ is the three-form

 $^2\mathrm{Note}$ that with our summation convention, we have

$$A_M dx^M = A_{\mu 5} dX^{\mu 5} + A_{5\mu} dX^{5\mu} + A_{\mu\nu} dX^{\mu\nu} = 2A_{\mu 5} dx^{\mu} + \frac{1}{2} A_{\mu\nu} \epsilon^{\mu\nu\sigma\rho} dy_{\sigma\rho}$$

In particular, we note the factor of 2 for the dx^{μ} contribution.

gauge field. The inverse³ metric is

$$M^{MN} = g^{-1/5} \begin{pmatrix} \frac{1}{4} g^{\mu\nu} & \frac{1}{4\sqrt{2}} g^{\mu\kappa} C_{\kappa\omega\omega'} \epsilon^{\omega\omega'\lambda\lambda'} \\ \frac{1}{4\sqrt{2}} g^{\nu\kappa} C_{\kappa\omega\omega'} \epsilon^{\omega\omega'\sigma\sigma'} & (\det g) g^{\sigma\sigma',\lambda\lambda'} + \frac{1}{8} g^{\alpha\beta} C_{\alpha\omega\omega'} \epsilon^{\omega\omega'\sigma\sigma'} C_{\beta\chi\chi'} \epsilon^{\chi\chi'\lambda\lambda'} \end{pmatrix}$$
(3.6)

There is further field content which can be seen to arise from the truncation of the eleven dimensional theory to four dimensions. In this process, the E_{11} decomposes as $SL(5) \times GL(7)$, and the eleven-dimensional metric as $g_{11} = \text{diag}(g, g_7)$, where g_7 is the metric on the seven-dimensional space. Extending the four-dimensional spacetime with six wrapping coordinates, the resulting metric on the (4 + 6 + 7)-dimensional space is [94]

$$\mathcal{M} = |\det(g_{11})|^{-1/2} \begin{pmatrix} \det(g)^{-1/5}M & 0\\ 0 & g_7 \end{pmatrix}.$$
 (3.7)

If we assume the diagonal form⁴ $g_7 = \det(g_7)^{1/7} \delta_7$, we can use $\det(g_{11}) = \det(g) \det(g_7)$ to find

$$\mathcal{M} = \begin{pmatrix} e^{-\Delta}M & 0\\ 0 & e^{-5\Delta/7}\delta_7 \end{pmatrix},\tag{3.8}$$

where

$$e^{\Delta} = \det(g_7)^{1/2} \det(g)^{7/10}$$
(3.9)

defines the additional degree of freedom, the volume factor scalar density Δ .

3.1.1 Action and equations of motion

We can now apply the method used in §2.3.1 to derive the full equations of motion for SL(5)-invariant M-theory. The action (up to boundary contributions, treated in [75]) is

$$S = \int \mathrm{d}^{10} X e^{\Delta} R, \qquad (3.10)$$

³Note that the identity here takes the form

$$\delta^N_M = \begin{pmatrix} \delta^\nu_{[\mu} \delta^5_{5]} & \delta^\nu_{[\mu} \delta^{\mu_{\mu'}]} \\ \delta^\nu_{[\mu} \delta^5_{5]} & \delta^\nu_{[\mu} \delta^{\nu_{\mu'}]} \end{pmatrix} = \begin{pmatrix} \frac{1}{2} \delta^\nu_\mu & 0 \\ 0 & \delta^\nu_{[\mu} \delta^{\nu'_{\mu'}} \end{pmatrix}$$

⁴As would arise on compactification on T^7 where the tori have equal radii det $(g_7)^{1/7}$.

where the SL(5) scalar

$$R = \frac{1}{12} M^{MN} \partial_M M^{KL} \partial_N M_{KL} - \frac{1}{2} M^{MN} \partial_M M^{KL} \partial_L M_{KN} - \partial_M \partial_N M^{MN} \quad (3.11)$$
$$- \frac{1}{7} M^{MN} \partial_M \Delta \partial_N \Delta - \frac{2}{7} \partial_M M^{MN} \partial_N \Delta - \frac{2}{7} M^{MN} \partial_M \partial_N \Delta.$$

The first three terms reproduce the four-dimensional supergravity action in the supergravity frame, whilst the latter three are kinetic terms for Δ .

Varying the action with respect to the volume factor yields the equation of motion

$$R = 0, \tag{3.12}$$

while varying the action with respect to the metric yields

$$\delta_M S = \int \mathrm{d}^{10} X e^{\Delta} \mathcal{K}_{MN} \delta M^{MN}, \qquad (3.13)$$

where

$$\mathcal{K}_{MN} = \frac{1}{12} \partial_M M^{KL} \partial_N M_{KL} - \frac{1}{2} M^{PK} M^{QL} \partial_L M_{PM} \partial_K M_{QN} \qquad (3.14)
+ \frac{1}{6} M^{PQ} M^{KL} \partial_P M_{KM} \partial_Q M_{LN} - \partial_M \partial_N \Delta - \frac{6}{7} \partial_M \Delta \partial_N \Delta
+ (\partial_L + (\partial_L \Delta)) \left[M^{LK} \left(\partial_{(M} M_{N)K} - \frac{1}{6} \partial_K M_{MN} \right) \right].$$

Again, this does not require $\mathcal{K}_{MN} = 0$, since δM^{MN} is constrained by its SL(5)/SO(5) coset structure (3.5). As shown in [98], this particular coset structure allows one to write the metric and volume factor in full generality in terms of a "little metric" m_{ab} where $M_{ab,cd} = m_{a[c}m_{d]b}$. From this, equations of motion were derived via construction of a semi-covariant derivative and curvature tensor analogous to the Riemann tensor. We will determine the metric equations of motion by the chain rule method of §2.3.1. For this purpose, it is convenient to define a rescaled metric

 $\tilde{M}^{MN} = g^{1/5} M^{MN}$. Thus,

$$g^{1/5} \mathcal{K}_{MN} \delta M^{MN} \tag{3.15a}$$

$$=g^{1/5}\mathcal{K}_{MN}\left(g^{-1/5}\delta\tilde{M}^{MN}+\delta g^{-1/5}\tilde{M}^{MN}\right)$$
(3.15b)

$$= \mathcal{K}_{MN} \left[\frac{\partial \tilde{M}^{MN}}{\partial g_{\mu\nu}} \delta g_{\mu\nu} + \frac{\partial \tilde{M}^{MN}}{\partial C_{\mu\nu\rho}} \delta C_{\mu\nu\rho} + \delta \ln g^{-1/5} \tilde{M}^{MN} \right]$$
(3.15c)

$$= \left[-g^{\mu\sigma}g^{\nu\rho}\mathcal{K}_{\mu5\nu5} - \frac{1}{\sqrt{2}}g^{\mu\sigma}g^{\rho\kappa}C_{\kappa\omega\omega'}\epsilon^{\omega\omega'\nu\nu'}\mathcal{K}_{\mu5\nu\nu'} + \left(gg^{\sigma\rho}g^{\mu\mu',\nu\nu'} - gg^{\mu\sigma}g^{\rho[\nu}g^{\nu']\mu'} - gg^{\mu[\nu}g^{\nu']\sigma}g^{\rho\mu'} - \frac{1}{8}g^{\sigma\alpha}g^{\rho\beta}C_{\alpha\omega\omega'}\epsilon^{\omega\omega'\mu\mu'}C_{\beta\chi\chi'}\epsilon^{\chi\chi'\nu\nu'}\right)\mathcal{K}_{\mu\mu'\nu\nu'} - \frac{1}{5}g^{\rho\sigma}\mathcal{K}_{MN}\tilde{M}^{MN} \right]\delta g_{\sigma\rho} + \left[\frac{1}{\sqrt{2}}g^{\mu\alpha}\epsilon^{\beta\gamma\nu\nu'}\mathcal{K}_{\mu5\nu\nu'} + \frac{1}{4}g^{\sigma\alpha}\epsilon^{\beta\gamma\mu\mu'}C_{\sigma\chi\chi'}\epsilon^{\chi\chi'\nu\nu'}\mathcal{K}_{\mu\mu'\nu\nu'}\right]\delta C_{\alpha\beta\gamma}$$
(3.15d)

$$= \left[-4\tilde{M}^{\sigma 5M}\tilde{M}^{\rho 5N}\mathcal{K}_{MN} + \tilde{M}^{\rho 5N}\epsilon^{\mu\sigma 5P}\epsilon_{\mu NR}\tilde{M}^{RQ}\mathcal{K}_{PQ} \right]$$
(3.15e)
$$-\frac{4}{5}\tilde{M}^{\sigma 5\rho 5}\mathcal{K}_{MN}\tilde{M}^{MN} \delta g_{\rho\sigma} + \sqrt{2}M^{M\alpha 5}\epsilon^{\beta\gamma\nu\nu'}\mathcal{K}_{M\nu\nu'}\delta C_{\alpha\beta\gamma}$$
$$= 4\tilde{M}^{\sigma 5M}\tilde{M}^{\rho 5N} \left(-\delta_{M}^{P}\delta_{N}^{Q} + M_{ML}Y^{LP}{}_{NR}M^{RQ} - \frac{1}{5}M_{MN}M^{PQ} \right) \mathcal{K}_{PQ}\delta g_{\rho\sigma}$$
$$+ \frac{\sqrt{2}}{1+2x}M^{M\sigma 5} \left(\delta_{M}^{P}\delta_{N}^{Q} - xM_{ML}Y^{LP}{}_{NR}M^{RQ} - yM_{MN}M^{PQ} \right) \\ \times \mathcal{K}_{PQ}\delta C_{\sigma\omega\omega'}\epsilon^{\omega\omega'N5},$$
(3.15f)

where the constants $x \neq -1/2$ and y are otherwise arbitrary. Sufficient equations of motion are thus

$$P_{MN}{}^{PQ}\mathcal{K}_{PQ} = 0, (3.16)$$

where the projector

$$P_{MN}{}^{PQ} = \delta_M^P \delta_N^Q - M_{ML} Y^{LP}{}_{NR} M^{RQ} + \frac{1}{5} M_{MN} M^{PQ}$$

$$= \left(\delta_M^R \delta_N^S - \frac{1}{10} M_{MN} M^{RS} \right) \left(\delta_R^P \delta_S^Q - M_{RK} Y^{KP}{}_{SL} M^{LQ} \right).$$
(3.17)

As in §2.3.1, one can determine if (3.16) are *necessary* by counting the dimension of the symmetric kernel of P for Minkowski metric and vanishing C_3 . We find, of the

10(10+1)/2 = 55 degrees of freedom in \mathcal{K}_{MN} , that 41 are projected out, leaving precisely the $\binom{4}{3}$ and 4(4+1)/2 degrees of freedom in C_3 and g respectively⁵. Further work includes verifying that these equations agree with those of [98].

In (3.17), the first bracket projects out terms proportional to the metric. This can be seen to arise from the relation

$$(\mathcal{K}_{PQ} + M_{PQ})\delta M^{PQ} = \mathcal{K}_{PQ}\delta M^{PQ} - \delta \ln \det M = \mathcal{K}_{PQ}\delta M^{PQ}, \qquad (3.18)$$

where we have used the fact that det(M) = 1. Such a term does not explicitly arise in the DFT projector simply because the metric is already projected out by the projector (2.40). Thus, both projectors are of the form

$$P_{MN}{}^{PQ} = \left(\delta_M^R \delta_N^S - \frac{1}{D} M_{MN} M^{RS}\right) \left(\delta_R^P \delta_S^Q - M_{RK} Y^{KP}{}_{SL} M^{LQ}\right), \qquad (3.19)$$

where D is the dimension of the extended geometry. A natural question is whether the equations of motion in extended geometries in higher dimensions are also given by (3.16) with projector (3.19), where metrics/vielbein have been constructed in [94] (d = 3, 4), [97] (d = 5-7), [211] (d = 8), and the corresponding Y-tensors in [102] for d = 3-7.

3.1.2 An SL(5) wave solution recovers the M2-brane

In this section, we will redefine many of the symbols used in chapter 2 to keep analogy to the DFT case clear. A wave solution for the SL(5) duality invariant theory is given by the generalised metric M_{MN} with line element

$$M_{MN} dX^{M} dX^{N} = (K-2)[(dx^{1})^{2} - (dx^{2})^{2} - (dx^{3})^{2}] + (dx^{4})^{2} + 2(K-1)[dx^{1}dy_{23} + dx^{2}dy_{13} - dx^{3}dy_{12}] - K[(dy_{13})^{2} + (dy_{12})^{2} - (dy_{23})^{2}] + (dy_{34})^{2} + (dy_{24})^{2} - (dy_{14})^{2}, (3.20)$$

and constant volume factor Δ . It can be interpreted as a pp-wave in the extended geometry which carries momentum in the directions dual to x^2 and x^3 , i.e. combinations of y_{12}, y_{13} and y_{23} (though as for the DFT case in §2.4, we have not constructed the conserved charges). In the pp-wave interpretation, it has no mass or charge and

⁵Again, we note that this degrees of freedom counting, while highly suggestive, is not strictly a full proof that (3.16) are the SL(5) covariant metric equations of motion. Specifically, we have not shown that the variation of the spacetime metric and three-form do not lie within the kernel of the projector.

the solution is pure metric, i.e. there is no form field it couples to. K is a harmonic function of the transverse radial coordinate $|x^4|$: $K = 1 + k_0 |x^4|$, where k_0 is a constant. The wave is smeared in the remaining dual directions.

So that we may interpret this solution from the perspective of objects originating in eleven-dimensional supergravity, we use a Kaluza-Klein reduction from eleven to four dimensions: $\mathcal{M}_{11} = \mathcal{M}_4 \times \mathcal{M}_7$, followed by an augmentation by the relevant winding coordinates to a ten-dimensional extended geometry. We do this assuming the supergravity frame, with spacetime coordinates x^{μ} . The metric ansatz is thus

$$M_{MN} dX^{M} dX^{N} = \left(g_{\mu\nu} + e^{2w} C_{\mu}{}^{\rho\sigma} C_{\rho\sigma\nu}\right) dx^{\mu} dx^{\nu}$$

$$+ 2e^{2w} C_{\mu}{}^{\rho\sigma} dx^{\mu} dy_{\rho\sigma} + e^{2w} g^{\lambda\tau,\rho\sigma} dy_{\lambda\tau} dy_{\rho\sigma},$$
(3.21)

where the function $e^{2w(x^{\mu})}$ is a scale factor inherited from the KK reduction from eleven to four dimensions. Comparing (3.21) with (3.20), we find

$$g_{\mu\nu} dx^{\mu} dx^{\nu} = -K^{-1} \left[(dx^{1})^{2} - (dx^{2})^{2} - (dx^{3})^{2} \right] + (dx^{4})^{2}$$

$$C_{123} = 1 - K^{-1} \qquad C_{\mu\nu4} = 0$$
(3.22)

with $e^{2w} = K^{-1}$. We now view this solution in the Einstein frame related to (3.22) by

$$\mathfrak{g}_{\mu\nu} = \sqrt{|\det(e^{2w}g^{\mu\nu,\rho\sigma})|} \, g_{\mu\nu} = K^{3/2}g_{\mu\nu}, \qquad (3.23)$$

where the determinant is calculated by considering $g^{\mu\nu,\rho\sigma}$ as a 6×6 matrix with rows given by $\{\mu\nu\} = \{12, 13, 14, 23, 24, 34\}$, and similarly for columns. The C-field is not transformed (only its field strength obtains a different factor in the action) and we find

$$\mathfrak{g}_{\mu\nu} \mathrm{d}x^{\mu} \mathrm{d}x^{\nu} = -K^{1/2} \left[(\mathrm{d}x^{1})^{2} - (\mathrm{d}x^{2})^{2} - (\mathrm{d}x^{3})^{2} \right] + K^{3/2} (\mathrm{d}x^{4})^{2}$$

$$C_{123} = 1 - K^{-1} \qquad C_{\mu\nu4} = 0,$$
(3.24)

which is the four-dimensional M2-brane in the Einstein frame, extended in the $x^2 - x^3$ plane. We have thus shown that the solution (3.20) which carries momentum in the directions dual to x^2 and x^3 in the extended geometry corresponds to a membrane stretched along these directions from a spacetime perspective. By similar arguments as in the string case, the mass and charge of the M2-brane are given by the momenta in the dual directions.

It would be interesting to see a Goldstone mode analysis for the SL(5) wave

analogous to that achieved for the DFT wave in §2.4.1. One would consider perturbations of the metric and volume factor given by the generalised Lie derivative with respect to a generalised vector $\sim K^{\alpha}\varphi\partial/\partial x^4 + K^{\beta}\tilde{\varphi}_{\mu}\partial/\partial y_{\mu4}$, which is constant on the worldvolume. Promoting the constants φ and $\tilde{\varphi}_{\mu}$ to functions of the worldvolume coordinates, one could then determine the linear equations of motion, which we expect would produce the duality relation of Duff [212] for the membrane.

Chapter 4

Solution-generating transformations in NS-NS supergravity and double field theory

In 1954, Buchdahl [123] found a symmetry group of the vacuum Einstein equations for static vacuum metrics. Metrics are defined to be static if they possess a Killing vector ξ which is additionally hypersurface orthogonal. This is true if and only if the *twist*, proportional to $\iota_{\xi} \star d\xi_{\flat}$, vanishes. In adapted coordinates $\xi = \partial_0$ (where $x^{\mu} = (x^0, x^a)$), this is equivalent to the statement that the metric takes the block diagonal form $g = \text{diag}(g_{00}, g_{ab})$.

Buchdahl's symmetry group was extended significantly in (3 + 1) dimensions by the work of Ehlers [213] and Geroch [214] to an $SL(2, \mathbb{R})$ symmetry group of the vacuum Einstein equations in the presence of a non-null Killing vector¹. Harrison [215] extended their work to generate maps between solutions to Einstein-Maxwell theory and vacuum Einstein gravity. These symmetry groups were found by considering a split of the metric and electromagnetic field into two parts:

• Those defining a codimension one *reduced* metric orthogonal to Killing direction, for example g_{ab} when $\xi = \partial_0$. The symmetry groups preserve a conformal transformation of this metric.

¹Mars [196] showed that their analysis was also valid for null Killing vectors.

• All remaining contributions, which are decomposed into a set of scalar functions or *potentials*, whose invariance transformations determine the symmetry group.

The symmetry group was completed by Neugebauer and Kramer [195] in Einstein-Maxwell theory for any non-null Killing vector.

In this chapter, we look at how we might find similar solution-generating symmetries in NS-NS supergravity and DFT (see [216–221] for similar studies in supergravity). We have seen that NS-NS supergravity with an isometry has an $O(1, 1, \mathbb{R})$ symmetry group, a remnant of string theory T-duality, and how this is made manifest in DFT. However, we also look for symmetries which lie outside this manifest symmetry group, and how they might appear within the DFT context. We begin our exploration with a reformulation of general relativity with an isometry, decomposing the metric into a reduced metric and *p*-form potentials, and then apply this to NS-NS supergravity before we approach DFT.

4.1 Reduction of Einstein gravity with respect to an isometry

The space V_{d-1} orthogonal to a Killing vector

Consider a gravitational theory with arbitrary field content. Define a *d*-dimensional manifold V_d with metric $g_{\mu\nu}$ which exhibits some isometry

$$(\mathcal{L}_{\xi}g)_{\mu\nu} = 2\nabla_{(\mu}\xi_{\nu)} = 0 \tag{4.1}$$

with respect to a Killing vector ξ . We can define a (d-1)-dimensional hypersurface V_{d-1} in V_d where objects $T_{\mu_1...\mu_p}{}^{\nu_1...\nu_q}$ are defined to live on V_{d-1} if they satisfy

$$(\mathcal{L}_{\xi}T)_{\mu_{1}...\mu_{p}}{}^{\nu_{1}...\nu_{q}} = 0 \tag{4.2a}$$

$$\xi^{\mu_s} T_{\mu_1 \dots \mu_s \dots \mu_p}{}^{\nu_1 \dots \nu_q} = 0 \qquad \forall s \in \{1, \dots, p\}$$
(4.2b)

$$\xi_{\nu_s} T_{\mu_1 \dots \mu_p}{}^{\nu_1 \dots \nu_s \dots \nu_q} = 0 \qquad \forall s \in \{1, \dots, q\}.$$
(4.2c)

The reduced metric on V_{d-1} is

$$\gamma_{\mu\nu} = g_{\mu\nu} - \frac{1}{F} \xi_{\mu} \xi_{\nu} \qquad F = \xi^{\mu} \xi_{\mu},$$
(4.3)

where the norm F, which lives on V_{d-1} , is required to be non-vanishing throughout the geometry² (ξ may be spacelike or timelike). The reduced metric acts in V_d as a projector on to V_{d-1} . The derivative on V_d of the Killing vector can be written in terms of objects on V_{d-1} as

$$\nabla_{\mu}\xi_{\nu} = \frac{1}{F} \left(-\frac{1}{2(d-3)!} \xi^{\rho} \hat{\epsilon}_{\sigma_{1}...\sigma_{d-3}\rho\mu\nu} \omega^{\sigma_{1}...\sigma_{d-3}} + \xi_{[\nu} \nabla_{\mu]} F \right), \tag{4.4}$$

where $\hat{\epsilon}_{\mu_1...\mu_d}$ is the alternating tensor (see app. A), and we have introduced the *twist*

$$\omega^{\sigma_1\dots\sigma_{d-3}} \equiv \hat{\epsilon}^{\sigma_1\dots\sigma_{d-3}\alpha\beta\gamma}\xi_\alpha \nabla_\beta\xi_\gamma. \tag{4.5}$$

which lives on V_{d-1} .

We can define the unique metric-compatible derivative operator on V_{d-1} [222, p. 257] by

$$D_{\mu}T^{\sigma_{1}\dots\sigma_{k}}{}_{\rho_{1}\dots\rho_{l}} = \gamma^{\sigma_{1}}_{\alpha_{1}}\dots\gamma^{\sigma_{k}}_{\alpha_{k}}\gamma^{\beta_{1}}_{\rho_{1}}\dots\gamma^{\beta_{l}}_{\rho_{l}}\gamma^{\nu}_{\mu}\nabla_{\nu}T^{\alpha_{1}\dots\alpha_{k}}{}_{\beta_{1}\dots\beta_{l}}, \tag{4.6}$$

in terms of which the twist and norm satisfy the following relations which will prove useful:

$$D_{\mu}\left(F^{-3/2}\omega^{\sigma_1\dots\sigma_{d-4}\mu}\right) = 0 \tag{4.7}$$

$$\nabla_{[\nu}\omega_{\mu_1\dots\mu_{d-3}]} = -2\xi^{\chi}\hat{\epsilon}_{\chi\nu\mu_1\dots\mu_{d-3}\lambda}R^{\lambda}{}_{\eta}\xi^{\eta}$$

$$\tag{4.8}$$

$$D^{\mu}D_{\mu}F = \frac{1}{2F}D^{\mu}FD_{\mu}F - \frac{1}{(d-3)!F}\omega^{\sigma_{1}...\sigma_{d-3}}\omega_{\sigma_{1}...\sigma_{d-3}} - 2\xi^{\mu}\xi^{\nu}R_{\mu\nu}.$$
(4.9)

In particular, while the twist is *not* one of the potentials which we use to define our solution-generating transformations, we will find the necessary potential via (4.8)—for example, consider vacuum spacetimes where $d\omega = 0$ implies $\omega = d\chi$.

We can define a reduced determinant of the reduced metric γ as follows. First, we note that γ has a zero eigenvalue and thus zero determinant

$$\gamma_{\mu\nu}\xi^{\nu} = 0 \quad \Rightarrow \quad \det(\gamma_{\mu\nu}) = 0.$$
 (4.10)

²The results of this formalism may in fact remain valid for null ξ , but we have not verified as such. For example, in [196], the author presents the (3 + 1)-dimensional case in a manner valid for null Killing vectors.

Writing the determinant of a $p \times p$ matrix A_{ab} as

$$\det(A_{ab}) = \frac{1}{p!} \epsilon^{c_1 \dots c_p} \epsilon^{d_1 \dots d_p} A_{c_1 d_1} \dots A_{c_p d_p},$$
(4.11)

where lower Roman indices here run 1 to p (note that we use the alternating symbol, rather than the tensor), we can decompose (4.10) to derive

$$\det(g) = F\gamma_{\xi},\tag{4.12}$$

where we have defined, for an arbitrary $d \times d$ matrix $J_{\mu\nu}$, a reduced determinant

$$J_{\xi} \equiv \frac{1}{(d-1)!F^2} \xi_{\alpha} \epsilon^{\alpha \sigma_1 \dots \sigma_{d-1}} \xi_{\beta} \epsilon^{\beta \phi_1 \dots \phi_{d-1}} J_{\sigma_1 \phi_1} \dots J_{\sigma_{d-1} \phi_{d-1}}, \qquad (4.13)$$

which is normalised such that for $\xi = \partial_0$, with $x^{\mu} = (x^0, x^a)$, we have

$$\gamma_{\xi} = \det(\gamma_{ab}). \tag{4.14}$$

It is central to our construction that for a given metric γ , twist ω , and norm F, all living on V_{d-1} , and satisfying (4.7), in addition to the Killing vector ξ and reduced metric determinant γ_{ξ} , we can reconstruct entirely the original metric g (up to diffeomorphisms) as follows. We can rewrite equation (4.4) in the form

$$\nabla_{[\mu}(F^{-1}\xi_{\nu]}) = -\frac{1}{2(d-3)!F^2} \sqrt{-F\gamma_{\xi}} \,\xi^{\rho} \epsilon_{\sigma_1\dots\sigma_{d-3}\rho\mu\nu} \omega^{\sigma_1\dots\sigma_{d-3}}, \tag{4.15}$$

which has a solution $F^{-1}\xi_{\mu}$ if the right hand side of (4.15) has vanishing exterior derivative

$$\nabla_{[\eta} \nabla_{\mu} (F^{-1} \xi_{\nu]}) = -\frac{1}{3! (d-4)!} \frac{1}{F^{1/2}} \xi^{\delta} \hat{\epsilon}_{\delta \eta \mu \nu \sigma_1 \dots \sigma_{d-4}} D_{\lambda} \left(F^{-3/2} \omega^{\sigma_1 \dots \sigma_{d-4} \lambda} \right).$$
(4.16)

By contracting with $\iota_{\xi}\star$, we find that the RHS of (4.16) vanishes if and only if (4.7) holds. The physical contribution to the gauge freedom in $F^{-1}\xi_{\mu}$ from (4.15) is fully constrained by the definition of the norm $\xi^{\mu}F^{-1}\xi_{\mu} = 1$. The remaining gauge freedom

$$F^{-1}\xi_{\mu} \to F^{-1}\xi_{\mu} + \partial_{\mu}z(x^{\nu}) \qquad \xi^{\mu}\partial_{\mu}z = 0$$
(4.17)

corresponds merely to a coordinate transformation which, in the coordinate frame where $\xi = \partial_0 (x^0 \text{ need not be the time coordinate})$ is $x^0 \to x^0 + z(x^{\mu \neq 0})$.

Curvature and conformal transformations on V_{d-1}

The Riemann tensor on V_{d-1} is defined in terms of a one-form u on V_{d-1} by

$$({}^{\gamma}R)_{\mu\nu\rho}{}^{\sigma}u_{\sigma} = (D_{\mu}D_{\nu} - D_{\nu}D_{\mu})u_{\rho}.$$
 (4.18)

One can show that the Riemann tensors on V_d and V_{d-1} are related by

$$({}^{\gamma}R)_{\mu\nu\rho}{}^{\eta} = \gamma^{\phi}_{[\mu}\gamma^{\sigma}_{\nu]}\gamma^{\beta}_{\rho}\gamma^{\eta}_{\lambda} \left[R_{\phi\sigma\beta}{}^{\lambda} + \frac{2}{F}\gamma^{\lambda\chi}(\nabla_{\phi}\xi_{\beta}\nabla_{\sigma}\xi_{\chi} + \nabla_{\phi}\xi_{\sigma}\nabla_{\beta}\xi_{\chi}) \right], \qquad (4.19)$$

where we have used the relation [223, p. 100]

$$\nabla_{\mu}\nabla_{\nu}\xi_{\sigma} = R_{\sigma\nu\mu\lambda}\xi^{\lambda}.$$
(4.20)

Substituting the expression (4.4) for the derivative of the Killing vector in to the Riemann curvature (4.19) on V_{d-1} , the Ricci tensor on V_{d-1} is then

$$({}^{\gamma}R)_{\mu\rho} = ({}^{\gamma}R)_{\mu\nu\rho}{}^{\nu} = \frac{1}{2F}D_{\mu}D_{\rho}F - \frac{1}{4F^{2}}D_{\rho}FD_{\mu}F + \frac{1}{2F^{2}}\frac{1}{(d-3)!}[(d-3)\omega^{\sigma_{1}...\sigma_{d-4}}{}_{\rho}\omega_{\sigma_{1}...\sigma_{d-4}\mu} - \gamma_{\mu\rho}\omega^{\sigma_{1}...\sigma_{d-3}}\omega_{\sigma_{1}...\sigma_{d-3}}] + \gamma_{\mu}^{\sigma}\gamma_{\rho}^{\nu}R_{\sigma\nu}.$$
(4.21)

We note at this point that all components of the Ricci tensor on V_d are encoded in the equations (4.8), (4.9), (4.21). With these equations interpreted as equations of motion for F, ω and γ , one can then reconstruct g from these objects provided the integrability condition (4.7) is satisfied. That is, these four equations fully determine the dynamics of the metric g—they are the full metric equations of motion.

It is useful for solution-generating transformations to consider a conformal rescaling of V_{d-1} with respect to a scalar κ on V_{d-1} . Define the space \hat{V}_{d-1} with metric and "inverse"

$$\hat{\gamma}_{\mu\nu} = \kappa \gamma_{\mu\nu} \qquad (\hat{\gamma}^{-1})^{\mu\nu} = \kappa^{-1} \gamma^{\mu\nu} \qquad \mathcal{L}_{\xi} \kappa = 0.$$
(4.22)

where the inverse is defined such that for tensors T on \hat{V}_{d-1} (equivalently on V_{d-1}), $\hat{\gamma}^{-1}\hat{\gamma}T = T$ (in particular, $\hat{\gamma}^{\mu\nu} = \kappa^2(\hat{\gamma}^{-1})^{\mu\nu}$). If we write the covariant derivatives on V_{d-1} and \hat{V}_{d-1} (acting on a objects on these spaces) as

$$D_{\mu}u_{\lambda} = \partial_{\mu}u_{\lambda} - (^{\gamma}\Gamma)^{\rho}{}_{\mu\lambda}u_{\rho} \tag{4.23}$$

$$\hat{D}_{\mu}u_{\lambda} = \partial_{\mu}u_{\lambda} - (\hat{\gamma}\Gamma)^{\rho}{}_{\mu\lambda}u_{\rho} \tag{4.24}$$

respectively, we can use metric compatibility on V_{d-1} : $D_{\mu}\gamma_{\nu\rho} = 0$, and demand metric compatibility on \hat{V}_{d-1} : $\hat{D}_{\mu}\hat{\gamma}_{\nu\rho} = 0$ to determine

$$\left(\left({}^{\hat{\gamma}} \Gamma \right)^{\sigma}{}_{\rho\nu} - \left({}^{\gamma} \Gamma \right)^{\sigma}{}_{\rho\nu} \right) \gamma_{\sigma}^{\lambda} = -\frac{1}{2} \gamma^{\lambda\mu} \left(\gamma_{\nu\rho} \partial_{\mu} \ln \kappa - \gamma_{\nu\mu} \partial_{\rho} \ln \kappa - \gamma_{\mu\rho} \partial_{\nu} \ln \kappa \right).$$
(4.25)

This is sufficient to relate the Ricci tensors on \hat{V}_{d-1} and V_{d-1} by

$$({}^{\hat{\gamma}}R)_{\mu\rho} = ({}^{\gamma}R)_{\mu\rho} - \frac{1}{2}\gamma_{\mu\rho}\gamma^{\sigma\nu} \left[\hat{D}_{\sigma}\hat{D}_{\nu}\ln\kappa - \frac{1}{2}(d-3)\hat{D}_{\sigma}\ln\kappa\hat{D}_{\nu}\ln\kappa \right]$$

$$- \frac{1}{2}(d-3)\hat{D}_{\mu}\hat{D}_{\rho}\ln\kappa - \frac{1}{4}(d-3)\hat{D}_{\mu}\ln\kappa\hat{D}_{\rho}\ln\kappa.$$

$$(4.26)$$

4.2 Solution-generating symmetries from an effective action

We look now to construct solution-generating transformations deriving from isometries, which preserve the conformally rescaled reduced metric $\hat{\gamma}$. Transformed quantities will be denoted with primes. Suppose the full equations of motion of the theory can be recovered by variation of an action

$$\int_{\hat{V}_{d-1}} \sqrt{|\hat{\gamma}_{\xi}|} \left(\hat{\gamma}_{R} + L_{p} \right) \quad \text{where} \quad \delta \hat{\gamma}_{\xi} = \hat{\gamma}_{\mu\nu} \delta(\hat{\gamma}^{-1})^{\mu\nu}, \quad (4.27)$$

we define $\hat{\gamma}R = (\hat{\gamma}^{-1})^{\mu\nu} (\hat{\gamma}R)_{\mu\nu}$, and L_p denotes all contributions from remaining fields, both metric and non-metric.

That L_p is the only part of the action that transforms is seen as follows. Firstly, the scalar curvature $\hat{\gamma}R$ is conserved. That the reduced determinant $\hat{\gamma}_{\xi}$ is invariant is shown as follows. Expanding $\hat{\gamma}' = \hat{\gamma}$, the transformed metric g' is

$$g'_{\mu\nu} = \frac{\kappa}{\kappa'} g_{\mu\nu} - \frac{\kappa}{\kappa'} \frac{1}{F} \xi_{\mu} \xi_{\nu} + \frac{1}{F'} \xi'_{\mu} \xi'_{\nu}, \qquad (4.28)$$

where $\xi'_{\mu} \equiv \xi^{\nu} g'_{\mu\nu}$. We can then use the following equation valid for any invertible $d \times d$ symmetric matrix M_{ij} [196]

$$\det(a_1 M_{ij} + a_2 b_i b_j + a_3 c_i c_j) = a_1^{d-2} \det(M_{ij}) \left[(a_1 + a_2 b^2) (a_1 + a_3 c^2) - a_2 a_3 (b \cdot c)^2 \right]$$

$$b^2 \equiv b_i b_j (M^{-1})^{ij} \qquad c^2 \equiv c_i c_j (M^{-1})^{ij} \qquad b \cdot c \equiv b_i c_j (M^{-1})^{ij}, \tag{4.29}$$

where $a_1 \neq 0$, a_2 and a_3 are all real constants and b_i , c_i are arbitrary d-column

vectors. Applying (4.29) to (4.28) we find

$$\frac{(\kappa')^{d-1}}{F'}\det(g') = \frac{\kappa^{d-1}}{F}\det(g),$$
(4.30)

or, using (4.12) and $\hat{\gamma}_{\xi} = \kappa^{d-1} \gamma_{\xi}$ from (4.13), we arrive at our result

$$\hat{\gamma}'_{\xi} = \hat{\gamma}_{\xi}. \tag{4.31}$$

Invariance transformations of L_p can in simple cases be found by observation or comparison with previous known results. However, in the case L_p forms a scalar sigma model, [195] developed a methodical procedure to find all resulting transformations.

Suppose all potentials are given by N scalars φ^A , $A = 1, \ldots, N$, and their contribution to the action (4.27) can be written

$$L_p = G_{AB}(\varphi^C)(\hat{\gamma}^{-1})^{\mu\nu}\partial_\mu\varphi^A\partial_\nu\varphi^B, \qquad (4.32)$$

where $det(G_{AB}) \neq 0$. One can consider the object G_{AB} as a metric on a *potential* space with coordinates φ^A , whose line element is

$$G_{AB}(\varphi^C) \mathrm{d}\varphi^A \mathrm{d}\varphi^B. \tag{4.33}$$

Infinitesimal invariance transformations of L_p are then generated by m Killing vectors $\Phi^{(m)} = \Phi^{(m)A} \partial/\partial \varphi_A$ on this potential space satisfying

$$(\mathcal{L}_{\Phi}G)_{AB} = 0, \tag{4.34}$$

as

$$\varphi^A \to \varphi^A + \lambda^{(m)} \Phi^{(m)A}, \tag{4.35}$$

for infinitesimal parameters $\lambda^{(m)}$. Finite transformations can be found by exponentiation, while discrete transformations can be derived by taking appropriate limits of the finite continuous transformations.

4.2.1 An example: (3+1)-d vacuum Einstein gravity with an isometry

It will be instructive for our exploration of solution-generating transformations in supergravity to briefly review the corresponding transformations in (3+1)-dimensional vacuum Einstein gravity. The Einstein equations

$$R_{\mu\nu} = 0 \tag{4.36}$$

can be reduced in the presence of a Killing vector ξ in terms of the metric $F\gamma$ (i.e. $\kappa = F$) and the scalars F and χ , where $\partial_{\mu}\chi = \omega_{\mu}$ and the existence of this scalar is guaranteed by (4.8) in vacuum. The full equations of motion are

$$(\hat{\gamma}R)_{\mu\rho} = \frac{1}{2F^2}\hat{D}_{\mu}F\hat{D}_{\rho}F + \frac{1}{2F^2}\hat{D}_{\rho}\chi\hat{D}_{\mu}\chi$$
 (4.37a)

$$\hat{D}^{\mu}\hat{D}_{\mu}\chi = \frac{2}{F}\hat{D}^{\mu}F\hat{D}_{\mu}\chi \tag{4.37b}$$

$$\hat{D}^{\mu}\hat{D}_{\mu}F = \frac{1}{F}\hat{D}^{\mu}F\hat{D}_{\mu}F - \frac{1}{F}\hat{D}^{\mu}\chi\hat{D}_{\mu}\chi$$
(4.37c)

where indices are raised with $\hat{\gamma}^{-1}$, and can be found by variation of the action

$$\int_{\hat{V}_{d-1}} \sqrt{-\hat{\gamma}_{\xi}} \left[\hat{\gamma}_{R} - \frac{1}{2F^{2}} \left(\hat{D}^{\mu} F \hat{D}_{\mu} F + \hat{D}^{\mu} \chi \hat{D}_{\mu} \chi \right) \right].$$
(4.38)

Following §4.2, one can then form the potential space line element

$$dS^{2} = \frac{1}{2F^{2}}(dF^{2} + d\chi^{2}), \qquad (4.39)$$

whose Killing vectors generate the transformations with infinitesimal real parameters k, p, q,

$$F \to F + kF\chi + pF$$
 $\chi \to \chi + \frac{1}{2}k(\chi^2 - F^2) + p\chi + q.$ (4.40)

The finite counterparts are the finite invariance transformations of the (3 + 1)dimensional vacuum Einstein equations in the presence of an isometry, which can be combined into the Möbius map

$$\sigma = F + i\chi \qquad \sigma \to \frac{\alpha \sigma - i\beta}{i\gamma \sigma + \delta} \qquad \begin{pmatrix} \alpha & \beta \\ \gamma & \delta \end{pmatrix} \in SL(2, \mathbb{R})$$
(4.41)

for constant α , β , γ , δ .

4.3 Application to NS-NS supergravity

We now apply our analysis to d-dimensional NS-NS supergravity with equations of motion

$$R_{\mu\nu} - \frac{1}{4} H_{\mu}{}^{\sigma\rho} H_{\nu\sigma\rho} + 2\nabla_{\mu} \nabla_{\nu} \phi = 0 \qquad (4.42a)$$

$$d \star (e^{-2\phi}H) = 0$$
 (4.42b)

$$H \wedge \star H - e^{2\phi} \mathbf{d} \star \mathbf{d} e^{-2\phi} = 0, \qquad (4.42c)$$

and Bianchi identity

$$\mathrm{d}H = 0, \tag{4.43}$$

for the case that the field content satisfies 3

$$\mathcal{L}_{\xi}H = 0 \qquad \mathcal{L}_{\xi}\phi = 0 \qquad \mathcal{L}_{\xi}g = 0. \tag{4.44}$$

We begin by constructing the potentials of theory, then proceeding to express the equations of motion in terms of these potentials. The field strength can be expressed under an "electric-magnetic" decomposition (see app. A.1)

$$FH = -\left((-1)^d (\iota_{\xi} \star)^2 + (\star \iota_{\xi})^2\right) H$$
$$= -\iota_{\xi} \star \beta - \star \iota_{\xi} \star \alpha, \qquad (4.45)$$

where the forms

$$\alpha = \iota_{\xi} H \qquad \beta = (-1)^d \iota_{\xi} \star H \tag{4.46}$$

satisfy

$$d\alpha = \mathcal{L}_{\xi} H - \iota_{\xi} dH \tag{4.47}$$

$$\mathcal{L}_{\xi}\alpha = \iota_{\xi}\mathcal{L}_{\xi}H \tag{4.48}$$

$$d\beta = (-1)^d (\mathcal{L}_{\xi} \star H - \iota_{\xi} d \star H)$$
(4.49)

$$\mathcal{L}_{\xi}\beta = (-1)^d \iota_{\xi} \mathcal{L}_{\xi} \star H. \tag{4.50}$$

We will refer to α and β as the electric and magnetic contributions respectively, for

³Owing to the gauge freedom in B, we do not require $\mathcal{L}_{\xi}B = 0$.

any ξ . Using the equation of motion (4.42b), equation (4.49) becomes

$$d\left(e^{-2\phi}\beta\right) = (-1)^d \mathcal{L}_{\xi}\left(e^{-2\phi} \star H\right).$$
(4.51)

Thus we have, at least locally, for all (g, H, ϕ) satisfying the equations of motion and Bianchi identity, and annihilated by \mathcal{L}_{ξ} , that we can write

$$\alpha = \mathrm{d}a \qquad \beta = e^{2\phi}\mathrm{d}b,\tag{4.52}$$

where da and db live on \hat{V}_{d-1} . Additionally, under isometry the exterior derivative of the twist (4.8) becomes

$$d\left(e^{-2\phi}\omega\right) = -(-1)^d 4d(a \wedge db), \qquad (4.53)$$

which ensures that we can define a potential χ by

$$\omega = e^{2\phi} (\mathrm{d}\chi - 4(-1)^d \mathrm{d}b \wedge a). \tag{4.54}$$

Our potentials are thus (F, χ, ϕ, a, b) .

We now restrict to the scalar potential space, by demanding that all higher form potentials vanish. There are multiple cases where this is true. These include static Einstein-dilaton theory in arbitrary dimension, and the magnetic sector of five-dimensional static NS-NS supergravity (for the latter, we refer the reader to the closely related analysis [221]).

4.3.1 Static Einstein-dilaton theory

We first examine static Einstein-dilaton theory in arbitrary dimension, with vanishing H and ω . The equations of motion are (raising indices with $\hat{\gamma}^{-1}$)

$$(\hat{\gamma}R)_{\mu\nu} = \frac{1}{2F} \hat{D}_{\mu} \hat{D}_{\rho} F - \frac{1}{4F} \hat{\gamma}_{\mu\rho} \hat{D}^{\sigma} \ln \kappa \hat{D}_{\sigma} F + \frac{1}{2F} \hat{D}_{(\mu} \ln \kappa \hat{D}_{\rho)} F - \frac{1}{4F^2} \hat{D}_{\mu} F \hat{D}_{\rho} F - 2 \hat{D}_{\mu} \hat{D}_{\rho} \phi + \hat{\gamma}_{\mu\rho} \hat{D}^{\sigma} \ln \kappa \hat{D}_{\sigma} \phi - 2 \hat{D}_{(\mu} \ln \kappa \hat{D}_{\rho)} \phi - \frac{1}{2} \hat{\gamma}_{\mu\rho} \left(\hat{D}^{\sigma} \hat{D}_{\sigma} \ln \kappa - \frac{1}{2} (d-3) \hat{D}^{\sigma} \ln \kappa \hat{D}_{\sigma} \ln \kappa \right) - \frac{1}{2} (d-3) \hat{D}_{\mu} \hat{D}_{\rho} \ln \kappa - \frac{1}{4} (d-3) \hat{D}_{\mu} \ln \kappa \hat{D}_{\rho} \ln \kappa$$

$$(4.55a)$$

$$\hat{D}^{\mu}\hat{D}_{\mu}F = \frac{1}{2F}\hat{D}^{\mu}F\hat{D}_{\mu}F + 2\hat{D}^{\mu}F\hat{D}_{\mu}\phi + \frac{1}{2}(d-3)\hat{D}^{\mu}\ln\kappa\hat{D}_{\mu}F$$
(4.55b)

$$\hat{D}^{\mu}\hat{D}_{\mu}\phi = 2\hat{D}^{\mu}\phi\hat{D}_{\mu}\phi - \frac{1}{F}\hat{D}^{\mu}F\hat{D}_{\mu}\phi + \frac{1}{2}(d-3)\hat{D}^{\mu}\ln\kappa\hat{D}_{\mu}\phi.$$
(4.55c)

We employ the conformal rescaling⁴

$$\kappa = \left(|F|e^{-4\phi}\right)^{1/(d-3)},\tag{4.56}$$

where the equations of motion simplify to

$$(\hat{\gamma}R)_{\mu\nu} = \frac{1}{4F^2}\hat{D}_{\mu}F\hat{D}_{\rho}F + \frac{1}{d-3}\left[\frac{1}{4F^2}\hat{D}_{\mu}F\hat{D}_{\rho}F - \frac{2}{F}\hat{D}_{(\mu}\ln\kappa\hat{D}_{\rho)}F - 4\hat{D}_{\mu}\phi\hat{D}_{\rho}\phi\right]$$
(4.57a)

$$\hat{D}^{\mu}\hat{D}_{\mu}F = \frac{1}{F}\hat{D}^{\mu}F\hat{D}_{\mu}F$$
(4.57b)

$$\hat{D}^{\mu}\hat{D}_{\mu}\phi = 0, \tag{4.57c}$$

which can be derived by variation of the following action with potentials F, κ

$$\int_{\hat{V}_{d-1}} \sqrt{|\hat{\gamma}_{\xi}|} \left[\hat{\gamma} R - (\hat{\gamma}^{-1})^{\mu\nu} \left(\frac{1}{4F^2} \hat{D}_{\mu} F \hat{D}_{\nu} F + \frac{d-3}{4\kappa^2} \hat{D}_{\mu} \kappa \hat{D}_{\nu} \kappa \right) \right].$$
(4.58)

⁴This is chosen as it is a simple example—in particular to render the equations of motion first order in derivatives. It would be interesting to find if other rescalings allowed different symmetry groups.

This action exhibits a $\mathbb{Z}_2\times\mathbb{R}\times\mathbb{Z}_2\times\mathbb{R}$ invariance group with finite generators

$$F \to \frac{1}{F}$$
 & $\phi \to \phi - \frac{1}{2} \ln |F|$ (4.59a)

$$F \to qF$$
 & $\phi \to \phi$ $q \in \mathbb{R}$ (4.59b)

$$F \to F$$
 & $\phi \to -\phi + \frac{1}{2} \ln |F|$ (4.59c)

$$F \to F$$
 & $\phi \to \phi - p$ $p \in \mathbb{R}$ (4.59d)

for constant p and q.

4.3.2 5-d static magnetic NS-NS supergravity

Here we consider the magnetic sector of static five-dimensional supergravity, where α and ω vanish. The equations of motion are

$$(\hat{\gamma}R)_{\mu\nu} = \frac{1}{2F}\hat{D}_{\mu}\hat{D}_{\rho}F - \frac{1}{4F}\hat{\gamma}_{\mu\rho}\hat{D}^{\sigma}\ln\kappa\hat{D}_{\sigma}F + \frac{1}{2F}\hat{D}_{(\mu}\ln\kappa\hat{D}_{\rho)}F - \frac{1}{4F^{2}}\hat{D}_{\mu}F\hat{D}_{\rho}F + \frac{e^{4\phi}}{F}\left(\hat{D}_{\mu}b\hat{D}_{\rho}b - \hat{\gamma}_{\mu\rho}\hat{D}^{\sigma}b\hat{D}_{\sigma}b\right) - 2\left(\hat{D}_{\mu}\hat{D}_{\rho}\phi - \frac{1}{2}\left(\hat{\gamma}_{\mu\rho}\hat{D}^{\sigma}\ln\kappa\hat{D}_{\sigma}\phi - 2\hat{D}_{(\mu}\ln\kappa\hat{D}_{\rho)}\phi\right)\right) - \frac{1}{2}\hat{\gamma}_{\mu\rho}\left(\hat{D}^{\sigma}\hat{D}_{\sigma}\ln\kappa - \hat{D}^{\sigma}\ln\kappa\hat{D}_{\sigma}\ln\kappa\right) - \hat{D}_{\mu}\hat{D}_{\rho}\ln\kappa - \frac{1}{2}\hat{D}_{\mu}\ln\kappa\hat{D}_{\rho}\ln\kappa$$

$$(4.60a)$$

$$\hat{D}^{\mu}\hat{D}_{\mu}F = \hat{D}^{\mu}\ln\kappa\hat{D}_{\mu}F + \frac{1}{2F}\hat{D}^{\mu}F\hat{D}_{\mu}F + 2\hat{D}_{\mu}F\hat{D}^{\mu}\phi$$
(4.60b)

$$\hat{D}^{\mu}\hat{D}_{\mu}\phi = \hat{D}^{\mu}\ln\kappa\hat{D}_{\mu}\phi + 2\hat{D}^{\mu}\phi\hat{D}_{\mu}\phi - \frac{1}{2F}\hat{D}^{\mu}F\hat{D}_{\mu}\phi + \frac{e^{4\phi}}{2F}\hat{D}^{\mu}b\hat{D}_{\mu}b$$
(4.60c)

$$\hat{D}^{\mu}\hat{D}_{\mu}b = \hat{D}^{\mu}\ln\kappa\hat{D}_{\mu}b + \frac{1}{2F}\hat{D}^{\mu}F\hat{D}_{\mu}b - 2\hat{D}^{\mu}\phi\hat{D}_{\mu}b.$$
(4.60d)

We employ the conformal factor (4.56)

$$\kappa = |F|^{1/2} e^{-2\phi},\tag{4.61}$$

where the equations of motion become

$$(\hat{\gamma}R)_{\mu\nu} = \frac{3}{8F^2} \hat{D}_{\mu}F\hat{D}_{\rho}F + 2\hat{D}_{\mu}\phi\hat{D}_{\rho}\phi - \frac{1}{F}\hat{D}_{(\mu}\phi\hat{D}_{\rho)}F + \frac{e^{4\phi}}{2F}\hat{D}_{\mu}b\hat{D}_{\rho}b \qquad (4.62a)$$

$$\hat{D}^{\mu}\hat{D}_{\mu}F = \frac{1}{F}\hat{D}^{\mu}F\hat{D}_{\mu}F$$
(4.62b)

$$\hat{D}^{\mu}\hat{D}_{\mu}\phi = \frac{e^{4\phi}}{2F}\hat{D}^{\mu}b\hat{D}_{\mu}b \tag{4.62c}$$

$$\hat{D}^{\mu}\hat{D}_{\mu}b = \frac{1}{F}\hat{D}^{\mu}F\hat{D}_{\mu}b - 4\hat{D}^{\mu}\phi\hat{D}_{\mu}b, \qquad (4.62d)$$

which can be derived by variation of an action, here written in terms of κ (as (4.61)), F and b, as

$$\int_{\hat{V}_{d-1}} \sqrt{|\hat{\gamma}_{\xi}|} \left[{}^{\hat{\gamma}}R - (\hat{\gamma}^{-1})^{\mu\nu} \left(\frac{1}{4F^2} \hat{D}_{\mu}F \hat{D}_{\nu}F + \frac{1}{2\kappa^2} \hat{D}_{\mu}\kappa \hat{D}_{\nu}\kappa + \frac{1}{\kappa^2} \hat{D}_{\mu}b \hat{D}_{\nu}b \right) \right].$$
(4.63)

This action exhibits an $SL(2,\mathbb{R}) \times \mathbb{Z}_2 \times \mathbb{R}$ invariance group with finite generators

$$F \to \frac{1}{F} \qquad \& \qquad \phi \to \phi - \frac{1}{2} \ln |F|$$

$$\tag{4.64a}$$

$$F \to qF$$
 & $\phi \to \phi - \frac{1}{4} \ln |q|$ $q \in \mathbb{R}$ (4.64b)

$$\sigma \to \frac{\alpha \sigma - i\beta}{i\gamma \sigma + \delta} \qquad \begin{pmatrix} \alpha & \beta \\ \gamma & \delta \end{pmatrix} \in SL(2, \mathbb{R}) \qquad \sigma = |F|^{1/2} e^{-2\phi} + ib. \tag{4.64c}$$

for constant q, α , β , γ and δ .

4.4 Extension to double field theory

In this section, we offer a few thoughts on extensions of the solution-generating techniques presented in this chapter to DFT. We will use the coordinate notation $X^M = (x^{\mu}, \tilde{x}^{\bar{\mu}})$ throughout. To do this, we will take inspiration from our analysis of Einstein-dilaton theory and static magnetic five-dimensional NS-NS supergravity in §4.3. It is the generalised metric H which is dynamical, and not the metric η which defines inner products, so it is the former, along with the DFT dilaton, that we will generate solutions of. We will choose an isometry $\partial_0 = 0$, here with respect to a generalised vector $V = \partial_0$ with Killing equations

$$\partial_0 H_{MN} = 0 \qquad \partial_0 d = 0. \tag{4.65}$$

We begin by noting that vanishing α and ω implies, for a given gauge of B,

$$B_{0m} = g_{0m} = 0. (4.66)$$

This can be seen from the DFT perspective as some kind of staticity requirement on the generalised metric with respect to $X^0 = x^0$, that

$$H_{0M} = \delta_M^0 H_{00} \qquad H_{\bar{0}M} = \delta_M^0 H_{\bar{0}\bar{0}}. \tag{4.67}$$

With regard to solution-generating techniques, one can easily see that preserving a reduced metric orthogonal to ∂_0 (as we did in supergravity) would be far too restrictive: all of the metric data is contained within this choice of reduced metric, and so preserving it allows only the dilaton to transform. Which quantity is it then that we wish to preserve? In our analysis of 5-d static magnetic supergravity, which can be found from DFT in the supergravity frame, where one imposes the section condition $\tilde{\partial}_{\bar{\mu}} = 0$, we preserved a conformally rescaled metric proportional to the $H_{\bar{m}\bar{n}}$ part of the generalised metric. Perhaps a natural generalisation is as follows

- Solve the section condition by requiring the fields depend on spacetime coordinates X^s , where s runs over some d of $(1, \ldots, 2d)$.
- Determine the metric components along these X^s directions, but excluding the direction of the isometry (and its dual).
- Preserve a conformal rescaling of these components.

For example, we could have

- 1. For Killing vector ∂_0 and section $\tilde{\partial}_{\bar{\mu}} = 0$, we preserve $\propto H_{\bar{m}\bar{n}}$
- 2. For Killing vector $\tilde{\partial}_{\bar{0}}$ and section $\partial_{\mu} = 0$, we preserve $\propto H_{mn}$.

We note however that since the metric is not an arbitrary symmetric object, but takes a coset form, there will be other metric components which must also be preserved by this transformation. It remains unclear how we determine which other components to preserve in general, and how precisely we do this. However, we circumvent this issue here by studying the simple case of vanishing two-form field (equivalently zero $H_{m\bar{n}}$), corresponding to $O(d, d, \mathbb{R})$ -covariant Einstein-dilaton theory, where the procedure becomes quite straightforward. We will impose the supergravity frame. Recalling that the metric is static (see (4.67)) with respect to x^0 , it takes the diagonal form

$$H = \text{diag}(H_{00}, H_{mn}, H_{\bar{0}\bar{0}}, H_{\bar{m}\bar{n}}).$$
(4.68)

We will look for transformations which preserve a conformal rescaling of the metric components H_{mn} . From the coset form of the metric, we can determine that this must also preserve appropriate rescalings of $H_{\bar{m}\bar{n}}$, H^{mn} and $H^{\bar{m}\bar{n}}$.

The equations of motion which do not vanish identically are

$$R = 0 \tag{4.69a}$$

$$\mathcal{K}_{00} = (H^{\bar{0}\bar{0}})^2 \mathcal{K}_{\bar{0}\bar{0}} \tag{4.69b}$$

$$\mathcal{K}_{mn} = H_{mr} \eta^{r\bar{s}} H_{nl} \eta^{l\bar{u}} \mathcal{K}_{\bar{s}\bar{u}} \tag{4.69c}$$

where R and the non-vanishing components of \mathcal{K} are written explicitly as

$$R = \frac{1}{4} H^{mn} \partial_m H^{qr} \partial_n H_{qr} - \frac{1}{2} H^{mn} \partial_m H^{ql} \partial_q H_{nl}$$
(4.70a)
+ $4 H^{mn} \partial_m \partial_n d - \partial_m \partial_n H^{mn} - 4 H^{mn} \partial_m d\partial_n d + 4 \partial_m H^{mn} \partial_n d$
+ $\frac{1}{8} H^{mn} \left(\partial_m H^{00} \partial_n H_{00} + \partial_m H^{\bar{0}\bar{0}} \partial_n H_{\bar{0}\bar{0}} \right)$
 $\mathcal{K}_{00} = \frac{1}{4} H^{mn} H^{00} \partial_m H_{00} \partial_n H_{00} - \frac{1}{4} (\partial_l - 2\partial_l d) (H^{lk} \partial_k H_{00})$ (4.70b)
 $\mathcal{K}_{\bar{0}\bar{0}} = \frac{1}{4} H^{mn} H^{\bar{0}\bar{0}} \partial_m H_{\bar{0}\bar{0}} \partial_n H_{\bar{0}\bar{0}} - \frac{1}{4} (\partial_l - 2\partial_l d) (H^{lk} \partial_k H_{\bar{0}\bar{0}})$ (4.70c)

$$\mathcal{K}_{mn} = \frac{1}{8} \left(\partial_m H^{00} \partial_n H_{00} + \partial_m H^{00} \partial_n H_{\bar{0}\bar{0}} \right) + \frac{1}{8} \partial_m H^{qr} \partial_n H_{qr} + \frac{1}{8} \partial_m H^{\bar{q}\bar{r}} \partial_n H_{\bar{q}\bar{r}} - \frac{1}{2} H^{rs} H^{ql} \partial_l H_{rm} \partial_s H_{nq} + \frac{1}{4} H^{rs} H^{ql} \partial_r H_{qm} \partial_s H_{nl} - \frac{1}{4} (\partial_l - 2\partial_l d) [H^{lq} (4\partial_{(m} H_{n)q} - \partial_q H_{mn})] + 2\partial_m \partial_n d \right]$$

$$(4.70d)$$

$$\mathcal{K}_{\bar{m}\bar{n}} = \frac{1}{4} H^{rs} H^{\bar{l}\bar{q}} \partial_r H_{\bar{l}\bar{m}} \partial_s H_{\bar{n}\bar{q}} - \frac{1}{4} (\partial_l - 2\partial_l d) (H^{lq} \partial_q H_{\bar{m}\bar{n}}).$$
(4.70e)

Noting that $H^{\bar{0}\bar{0}} = 1/H^{00} = H_{00} = 1/H_{\bar{0}\bar{0}}$, and fixing d and H_{mn} etc. (employing a trivial conformal rescaling of 1), the equations are clearly invariant under the

transformations

$$H_{00} \to 1/H_{00}$$
 (4.71a)

$$H_{00} \to q H_{00} \qquad q \in \mathbb{R}. \tag{4.71b}$$

So that we may compare this with the Einstein-dilaton analysis of §4.3.1, we note the relations

$$\gamma_{mn} = H_{mn}, \qquad F = H_{00}, \qquad \kappa^{d-3} = \frac{1}{|\gamma_{\xi}|} e^{-4d}.$$
 (4.72)

From these equalities, we see that fixed d and H_{mn} corresponds to fixed κ , and that (4.71) are the $\mathbb{Z}_2 \times \mathbb{R}$ transformations (4.59a) and (4.59b). The first constitutes a T-duality transformation, while the second a trivial coordinate rescaling. The remaining symmetries in (4.59) are not manifest, since they do not belong to T-duality. Retrieving these, and any further possible invariance transformations from these equations, are avenues for future research. It would indeed be interesting to see more general constructions for finding solution-generating symmetries in DFT, perhaps utilising the constructions of DFT covariant derivatives and curvature tensors in [69–74].

Chapter 5

Solution-generating transformations in the fluid/gravity correspondence

5.1 Hydrodynamics

The study of hydrodynamics is fundamental to vast areas of physics and engineering, owing to its origin as the long-wavelength limit of any interacting field theory at finite temperature. Such a limit needs a consistent definition. Consider a quantum field theory where quanta interact with a characteristic length scale $\ell_{\rm corr}$, the correlation length. The long-wavelength limit simply requires that fluctuations of the thermodynamic quantities of the system vary with a length scale L much greater than $\ell_{\rm corr}$, parametrized by the dimensionless Knudsen number

$$K_n = \frac{\ell_{\rm corr}}{L}.\tag{5.1}$$

For a fluid description to be useful in non-equilibrium states, we naturally require that L remain small compared to the size of the system. This is usually satisfied trivially by considering systems of infinite size.

The long-wavelength limit allows the definition of a particle as an element of the macroscopic fluid, infinitesimal with respect to the size of the system, yet containing a sufficiently large number of microscopic quanta. One mole contains an Avogadro's number of molecules, for example. Each particle defines a local patch of the fluid in thermal equilibrium, that is, thermodynamic quantities do not vary within the

particle. Away from global equilibrium quantities vary between particles as function of time τ and spatial coordinates x^i , combined as $x^a = (\tau, x^i)$. The evolution of particles in the fluid is parametrized by a relativistic velocity $u^b(x^a)$, which refers to the velocity of the fluid at x^a . It is well known [224] that the thermodynamic quantities, such as the temperature $T(x^a)$ and the local energy density $\rho(x^a)$, are determined by the value of any two of them, along with the equation of state. The evolution of the system is then specified by the equations of hydrodynamics in terms of a set of transport coefficients, whose values depend on the fluid in question.

Fluid flow is in general relativistic in that the systems it describes are constrained by local Lorentz invariance, and velocities may take any physical values below the speed of light. Applications at relativistic velocities are multitudinous: the dust clouds in galaxy and star formation; the flow of plasmas and gases in stars supporting fusion; the superfluid cores of neutron stars; the near-horizon dynamics of black holes are all described by hydrodynamics. Modelling black holes (and black branes in M/string theory) with hydrodynamics has now developed into a fundamental correspondence of central importance to our present study, as discussed in §5.2. Quark-gluon plasmas behave as nearly ideal fluids and are expected to have formed after the inflationary epoch of the big bang, and are reproduced in collisions at the RHIC and LHC. Non-relativistic fluids are equally ubiquitous, somewhat more familiar, and constitute an endless list of phenomena from the atmosphere to the oceans.

5.1.1 The fluid equations

We begin with a discussion, adapted from [176], of the relativistic fluid described by the stress energy tensor T^{ab} and a set of conserved currents J_I^a where I indexes the corresponding conserved charge. The dynamical equations of the (d-1)-spacetime dimensional fluid are

$$\nabla_a T^{ab} = 0 \tag{5.2a}$$

$$\nabla_a J_I^a = 0. \tag{5.2b}$$

For an ideal fluid, with no dissipation, the energy-momentum tensor and currents may be expressed in a local rest frame in the form

$$T^{ab} = \rho u^a u^b + p(g^{ab} + u^a u^b)$$
(5.3a)

$$J_I^a = q_I u^a \tag{5.3b}$$

where p is the pressure, q_I are the conserved charges and g_{ab} is the metric of the space on which the fluid propagates. The velocity is normalised to $u^a u_a = -1$. The entropy current is given by (5.3b) with the charge q being given by the local entropy density. The conservation of the entropy current illustrates the non-dissipative nature intrinsic to zero entropy production.

In a dissipative fluid, there are corrections to (5.3). We must first take into account the interrelation between mass and energy to define the velocity field more rigorously. This is achieved by using the Landau gauge, which requires that the velocity be an eigenvector of the stress-energy tensor with eigenvalue the local energy density of the fluid. This is satisfied when dissipative terms are orthogonal to the velocity (it is satisfied for the ideal fluid by the normalisation of the velocity). If the stress energy tensor gains a dissipative term Π^{ab} , and the current a term Υ^a_I , this reads

$$\Pi^{ab}u_a = 0 \qquad \Upsilon^a_I u_a = 0. \tag{5.4}$$

Dissipative corrections to the stress tensor are constructed in a derivative expansion of the velocity field and thermodynamic variables, where derivatives implicitly scale with the infinitesimal Knudsen number (5.1). Recalling that the equations of motion for the ideal fluid are composed of relations between these gradients, we may express Π^{ab} purely in terms of the derivative of the velocity. This can be iterated to all orders in the expansion (except when charges are present in which case this is only true to to first order). Now, the derivative of the velocity may be decomposed using the acceleration A^a , divergence ϑ , a symmetric traceless shear σ^{ab} , and the antisymmetric vorticity w^{ab} into the form

$$\nabla^b u^a = -A^a u^b + \sigma^{ab} + w^{ab} + \frac{1}{d-2} \vartheta P^{ab}, \qquad (5.5)$$

where

$$\vartheta = \nabla_a u^a \tag{5.6}$$

$$A^a = u^b \nabla_b u^a \tag{5.7}$$

$$\sigma^{ab} = P^{ac} P^{bd} \nabla_{(c} u_{d)} - \frac{1}{d-2} \theta P^{ab}$$
(5.8)

$$w^{ab} = P^{bc} P^{ad} \nabla_{[c} u_{d]}. \tag{5.9}$$

and $P^{ab} = g^{ab} + u^a u^b$ is a projection operator on to spatial directions. In the Landau frame, only the divergence and shear can contribute to the first-order stress-energy tensor. A similar analysis for the charge current retains the acceleration, and if one includes the parity-violating pseudo-vector contribution

$$\ell^a = \hat{\epsilon}_{bcd}{}^a u^b \nabla^c u^d, \tag{5.10}$$

the leading order dissipative equations of motion for a relativistic fluid are (5.2) with

$$T^{ab} = \rho u^a u^b + p P^{ab} - 2\eta \sigma^{ab} - \zeta \vartheta P^{ab}$$
(5.11a)

$$J_I^a = q_I u^a - \chi_{IJ} P^{ab} \nabla_b q_J - \Theta_I \ell^a - \gamma_I P^{ab} \nabla_b T, \qquad (5.11b)$$

where η and ζ are the shear¹ and bulk viscosities respectively, χ_{IJ} is the matrix of charge diffusion coefficients, γ_I indicates the contribution of the temperature gradients and Θ_I the pseudo-vector transport coefficients. The transport coefficients have been calculated in the weakly coupled QFT in perturbation theory, whereas in the strongly coupled theory, a dual holographic description may be employed, see e.g. [129].

5.1.2 Non-relativistic Navier-Stokes fluids

We now take the non-relativistic limit, defined by long distances, long times and low velocity and pressure amplitudes, of the relativistic fluid on a spacetime lightly perturbed in the following manner [131]. Consider an arbitrary perturbation

$$G_{ab} = g_{ab} + H_{ab} \tag{5.12}$$

¹We use the traditional notation of [128, 131, 224] rather than that of [182, 183]. We introduce the kinematical viscosity ν in §5.1.2.

of the background metric

$$g_{ab}\mathrm{d}x^a\mathrm{d}x^b = -\mathrm{d}\tau^2 + g_{ij}\mathrm{d}x^i\mathrm{d}x^j \tag{5.13}$$

(this form is purely a choice of coordinate system for a large class of metrics), where

$$H_{\tau\tau} = \epsilon^2 h_{\tau\tau}(\epsilon^2 \tau, \epsilon x^i)$$

$$H_{\tau i} = \epsilon \hat{a}_i(\epsilon^2 \tau, \epsilon x^i)$$

$$H_{ij} = \epsilon^2 h_{ij}(\epsilon^2 \tau, \epsilon x^i),$$
(5.14)

for infinitesimal scaling parameter ϵ . Treating the fluid on G_{ab} as an effective forced flow of that on g_{ab} , whilst preserving the normalisation $u_a u^a = -1$, the velocity

$$u^{a} = \frac{1}{\sqrt{1 - g_{ij}V^{i}V^{j}}}(1, V^{i}), \qquad (5.15)$$

density and pressure may be expanded about an equilibrium configuration (ρ_0, V_0^i, p_0) of a stationary, uniform fluid at rest², as

$$V^{i} = \epsilon v^{i}(\epsilon^{2}\tau, \epsilon x^{i})$$

$$\frac{p - p_{0}}{\rho_{0} + p_{0}} = \epsilon^{2} p_{e}(\epsilon^{2}\tau, \epsilon x^{i}).$$
(5.16)

The scaling of the density is fixed by the equation of state. The scaling of time τ and space x^i are implicit: time and space derivatives acting on the velocity and pressure will draw out, via the chain rule, the corresponding power of ϵ . In this limit, the temporal component of the relativistic Navier-Stokes equations becomes

$$\nabla_a T^{a\tau} = \epsilon^2 (\rho_0 + p_0) \nabla_i v^i + \mathcal{O}(\epsilon^4), \qquad (5.17)$$

where ∇_i is the covariant derivative with respect to the spatial metric g_{ij} . Separating the gauge field \hat{a}_i into its pure curl and divergence parts

$$\hat{a}_i = a_i + \nabla_i \chi, \tag{5.18}$$

²We assume the background allows such stationary, uniform solutions. In particular, one would not expect generic non-static metrics g_{ab} to allow stationary fluid solutions. On the other hand, stationary, uniform solutions will of course exist in flat space, for example.

such that $\nabla_i a^i = 0$, and defining an effective pressure

$$P = p_e - \frac{1}{2}h_{\tau\tau} + \frac{\partial\chi}{\partial\tau},\tag{5.19}$$

the spatial components yield

$$\nabla_a T^{ai} = \epsilon^3 (\rho_0 + p_0) [\nabla_i P + \partial_\tau v_i + v^j \nabla_j v_i - \nu (\nabla^j \nabla_j v_i + R_{ij} v^j) + \partial_\tau a_i + v^j f_{ji}] + \mathcal{O}(\epsilon^5),$$
(5.20)

where

$$\nu = \eta / (\rho_0 + p_0) \tag{5.21}$$

is the kinematical viscosity, $f_{ij} \equiv \partial_i a_j - \partial_j a_i$ is the field strength of a_i (equivalently \hat{a}_i) and R_{ij} is the Ricci tensor of g_{ij} . If a fluid carries conserved charges, equation (5.2b) will also yield incompressibility (5.17). Taking the hydrodynamic limit $\epsilon \to 0$, these give the non-relativistic incompressible Navier-Stokes (INS)³ equations with a forcing function due to an external electromagnetic field,

$$\partial_{\tau} v_i - \nu (\partial^2 v_i + R_{ij} v^j) + \partial_i P + v^j \partial_j v_i = -\partial_{\tau} a_i - v^j f_{ji}$$
(5.22a)

$$\partial_i v^i = 0. \tag{5.22b}$$

Ideal fluids are described by Euler's equations, obtained by setting $\nu = 0$. We will mostly be concerned with fluid flow on flat space $(R_{ij} = 0)$ in the absence of external forces, where H_{ab} , thus a_i , are zero.

5.1.3 Open problems in the Navier-Stokes equations

Despite extensive and successful application in physics and engineering, and two hundred years of intense mathematical study, the Navier-Stokes equations still obscure a great number of secrets. In particular, the questions of existence, uniqueness and regularity, see [225].

Consider a solution $v^j(\tau, x^i)$, $P(\tau, x^i) \in C^{\infty}(\mathbb{R}^d)$ to the incompressible Navier-Stokes equations in some domain $\Gamma \in \mathbb{R}^d$, initially at time $\tau = \tau_0$. Existence refers to the condition that the kinetic energy $v^2/2$ remains bounded for all time. In fact, we expect for finite mean dissipation, solutions to decay to equilibrium for $\tau \to \infty$. Uniqueness refers to the question of whether $v^j(\tau, x^i)$ is uniquely determined by

³INS will refer to the non-relativistic incompressible Navier-Stokes equations without forcing terms, i.e. when $a_i = 0$ in (5.22).

 $v^{j}(\tau_{0}, x^{i})$ for all $\tau \geq \tau_{0}$, and regularity is concerned with, for an isolated system, the most general conditions for smooth solutions to exist, which may be expressed in terms of the mean-square of the vorticity $\omega_{ij} = \partial_{i}v_{j} - \partial_{j}v_{i}$, as whether

$$\int_{\tau_0}^{\tau_1 \ge \tau_0} \mathrm{d}\tau \left[\int_{\Gamma} |\omega(\tau, x^i)|^2 \right]^2 < \infty.$$
(5.23)

For d = 2, the vorticity is self-parallel, and resulting conservation laws fix the vorticity to remain finite, so solutions are smooth. Moreover, it is known that solutions are unique [226].

5.2 The fluid/gravity correspondence

In 1974, Damour [227], and later in 1986, Thorne et al. [228], considered an observer outside a black hole, interacting with (perturbing) the event horizon. Surprisingly, they found that the observer will experience perturbations of the "stretched" horizon (no observer can exist at the horizon itself) described by modes of a viscous fluid possessing electric charge and conductivity. While this model is just that, a model, it did inspire investigation into whether a more concrete correspondence could be found. This began when Policastro et al. [125] related the shear viscosity of $\mathcal{N} = 4$ super Yang-Mills theory to the absorption of energy by a black brane (a further set of coefficients were subsequently similarly derived in [129]). This was carried out in the context of the anti-de Sitter/conformal field theory correspondence, an equivalence between a theory of gravity in anti-de Sitter space and a conformal field theory in one dimension less [124].

This signalled a major emergence in holography, the correspondence between gravitational theories on *d*-dimensional manifolds and (d-1)-dimensional quantum field theories dual to the the dynamics on a hypersurface within. The correspondence holds independent of the coupling of the QFT, though in the strong coupling limit, the gravitational dual becomes classical. Motion of the hypersurface along the bulk dimension parametrizes the renormalisation group (RG) flow, or energy scaling of the dual QFT.

What was now understood was that this holographic correspondence can be viewed in the long-wavelength limit of the dual QFT, where one recovers a dual hydrodynamics. In particular, the long-wavelength limit can be expressed as the set of hydrodynamic solutions varying slowly on a scale set by the extrinsic curvature of the hypersurface. This motivated what is now known as the fluid/gravity correspondence. The gravitational dual to the thermal state is given by the structure of the spacetime, for example in AdS space by an AdS black hole. As one expects from the UV/IR connection [229], where infrared effects in the bulk emerge as ultraviolet effects in the boundary theory, the relativistic hydrodynamics of perturbations about this thermal state determine the bulk gravity solution in a gradient expansion off the hypersurface. The corresponding metric construction was developed initially in [178].

5.2.1 The Navier-Stokes fluid on a Rindler boundary

A metric dual to the INS equations was first developed in [182] on the Rindler wedge, up to third order in the non-relativistic, small amplitude expansion detailed later in this section. An algorithm for generalising this metric to all orders was subsequently developed in [183], though terms calculated beyond third order are not universal. They receive corrections from quadratic curvature in Gauss-Bonnet gravity [184]. The construction of general metrics dual to relativistic fluids is set up in [181]. We summarise the construction in [183] here.

Consider the (d-1)-dimensional surface Σ_c with induced metric

$$y_{ab}\mathrm{d}x^a\mathrm{d}x^b = -r_c\mathrm{d}\tau^2 + \mathrm{d}x_i\mathrm{d}x^i,\tag{5.24}$$

where the parameter r_c is an arbitrary constant. One metric embedding this surface is

$$\mathrm{d}s^2 = -r\mathrm{d}\tau^2 + 2\mathrm{d}\tau\mathrm{d}r + \mathrm{d}x_i\mathrm{d}x^i,\tag{5.25}$$

which describes flat space (fig. 5.1) in ingoing Rindler coordinates $x^{\mu} = (\tau, x^i, r)$, defined in terms of the Cartesian chart (t, x^i, z) by

$$z^2 - t^2 = 4r \qquad z + t = e^{\tau/2}.$$
(5.26)

The hypersurface Σ_c is defined by $r = r_c$ where r is the coordinate into the bulk. Allowing for a family of equilibrium configurations, consider diffeomorphisms satisfying the three conditions

- i) The induced metric on the hypersurface Σ_c takes the form (5.24).
- ii) The stress tensor on Σ_c describes a perfect fluid.
- iii) Diffeomorphisms return metrics stationary and homogeneous in (τ, x^i) .



Figure 5.1: The past \mathcal{H}^- and future \mathcal{H}^+ horizons define the boundary of the Rindler wedge. Grey lines demonstrate lines of constant r (curved) and τ (straight). Long-wavelength perturbations of the hypersurface Σ_c are described by the equations of hydrodynamics (image from [152]).

The allowed set is given by the following boost, shift and rescaling of x^{μ} . First, a constant boost β_i ,

$$\sqrt{r_c}\tau \to \gamma(\sqrt{r_c}\tau - \beta_i x^i), \qquad x^i \to x^i - \gamma\beta^i \sqrt{r_c}\tau + (\gamma - 1)\frac{\beta^i \beta_j}{\beta^2} x^j,$$
 (5.27)

where $\gamma = (1 - \beta^2)^{-1/2}$ and $\beta_i \equiv r_c^{-1/2} v_i$. Second, a shift in r and a rescaling of τ ,

$$r \to r - r_h, \qquad \tau \to (1 - r_h/r_c)^{-1/2} \tau.$$
 (5.28)

These yield the flat space metric in rather complicated coordinates,

$$ds^{2} = \frac{d\tau^{2}}{1 - v^{2}/r_{c}} \left(v^{2} - \frac{r - r_{h}}{1 - r_{h}/r_{c}} \right) - \frac{2\gamma}{\sqrt{1 - r_{h}/r_{c}}} d\tau dr - \frac{2\gamma v_{i}}{r_{c}\sqrt{1 - r_{h}/r_{c}}} dx^{i} dr + \frac{2v_{i}}{1 - v^{2}/r_{c}} \left(\frac{r - r_{c}}{r_{c} - r_{h}} \right) dx^{i} d\tau + \left(\delta_{ij} - \frac{v_{i}v_{j}}{r_{c}^{2}(1 - v^{2}/r_{c})} \left(\frac{r - r_{c}}{1 - r_{h}/r_{c}} \right) \right) dx^{i} dx^{j}.$$
(5.29)

The Brown-York stress tensor on Σ_c (in units where $16\pi G = 1$) is given by

$$T_{ab} = 2(Ky_{ab} - K_{ab}), (5.30)$$

where

$$K_{ab} = \frac{1}{2} (\mathcal{L}_n y)_{ab}, \qquad K = K^a{}_a,$$
 (5.31)

are the extrinsic curvature and its mean, and n^{μ} is the spacelike unit normal to the hypersurface.

By imposing that the Brown-York stress tensor on Σ_c gives that of the stressenergy tensor of a fluid we can identify the parameters of the metric (5.29) with the density ρ , pressure p and four-velocity u^a of a fluid, as follows:

$$\rho = 0, \qquad p = \frac{1}{\sqrt{r_c - r_h}}, \qquad u^a = \frac{1}{\sqrt{r_c - v^2}}(1, v_i).$$
(5.32)

The Hamiltonian constraint

$$R_{\mu\nu}n^{\mu}n^{\nu} = 0 \tag{5.33}$$

on Σ_c yields a constraint on the Brown-York stress tensor

$$(d+2)T_{ab}T^{ab} = (T_a^a)^2. (5.34)$$

When this constraint is applied to the equilibrium configurations described above, one finds the equation of state is $\rho = 0$ (as above), or $\rho = -2(d+2)(d+1)p$ which occurs for a fluid on the Taub geometry [180].

Promoting v_i and p to slowly varying functions of the coordinates x^a , and regarding $v^i(x^a)$ and $p = r_c^{-1/2} + r_c^{-3/2} P(x^a)$ as small perturbations scaling as

$$v_i \sim \epsilon, \qquad P \sim \epsilon^2,$$
 (5.35)

about equilibrium, yields the metric

$$ds^{2} = -rd\tau^{2} + 2d\tau dr + dx^{i}dx_{i}$$

$$-2\left(1 - \frac{r}{r_{c}}\right)v_{i}dx^{i}d\tau - 2\frac{v_{i}}{r_{c}}dx^{i}dr$$

$$+\left(1 - \frac{r}{r_{c}}\right)\left[(v^{2} + 2P)d\tau^{2} + \frac{v_{i}v_{j}}{r_{c}}dx^{i}dx^{j}\right] + \left(\frac{v^{2} + 2P}{r_{c}}\right)d\tau dr + \mathcal{O}(\epsilon^{3})$$
(5.36)

which satisfies the Einstein's equations to $\mathcal{O}(\epsilon^2)$ if v_i satisfies incompressibility, $\partial_i v^i = \mathcal{O}(\epsilon^3)$. Corrections appear in powers of ϵ^2 , so this is the complete metric to second order.
The metric may now be built up order by order in the hydrodynamic scaling. Assume one has the metric at order ϵ^{n-1} , where the first non-vanishing component $\hat{R}^{(n)}_{\mu\nu}$ of the Ricci tensor appears at order n. By adding a correction term $g^{(n)}_{\mu\nu}$ to the metric at order n, resulting in a shift in the Ricci tensor $\delta R^{(n)}_{\mu\nu}$, we can ensure the vanishing of the Ricci tensor is guaranteed to order n if

$$\hat{R}^{(n)}_{\mu\nu} + \delta R^{(n)}_{\mu\nu} = 0.$$
(5.37)

Recalling that, in the hydrodynamic scaling, derivatives scale as

$$\partial_r \sim \epsilon^0, \qquad \partial_i \sim \epsilon^1, \qquad \partial_\tau \sim \epsilon^2,$$
 (5.38)

one sees that corrections $\delta R^{(n)}_{\mu\nu}$ at order *n* will appear only as *r* derivatives of $g^{(n)}_{\mu\nu}$. It is shown in [183] that, using the Bianchi identity and the Gauss-Codacci relations, integrability of the set of differential equations (5.37) defining $\delta R^{(n)}_{\mu\nu}$ in terms of $g^{(n)}_{\mu\nu}$ is given by imposing the momentum constraint, equivalent to the conservation of the stress tensor on Σ_c ,

$$R_{a\mu}n^{\mu} = \nabla_a T^{ab}|_{\Sigma_c}^{(n)} = 0, \qquad (5.39)$$

which is precisely the fluid equations of motion, to order n.

The perturbation scheme contains several degrees of freedom. The gauge freedom of the infinitesimal perturbations

$$g_{\mu\nu}^{(n)} \to g_{\mu\nu}^{(n)} + \partial_{\mu}\varphi_{\nu}^{(n)} + \partial_{\nu}\varphi_{\mu}^{(n)} \tag{5.40}$$

for some arbitrary vector $\varphi^{\mu(n)}(\tau, \vec{x}, r)$ at order ϵ^n , which may be fixed by demanding that $g_{r\mu}$ is that of the seed metric to all orders in ϵ . The x^a -dependent functions of integration from equation (5.37) may be fixed by imposing the boundary form (5.24) of the metric on Σ_c , and also requiring regularity of the metric at r = 0, which in this construction translates to the absence of logarithmic terms in r. Corrections to the bulk metric under these conditions then become

$$\begin{split} g_{r\mu}^{(n)} &= 0\\ g_{\tau\tau}^{(n)} &= (1 - r/r_c) F_{\tau}^{(n)}(x^a) + \int_r^{r_c} \mathrm{d}r' \int_{r'}^{r_c} \mathrm{d}r''(\hat{R}_{ii}^{(n)} - r\hat{R}_{rr}^{(n)} - 2\hat{R}_{r\tau}^{(n)})\\ g_{\tau i}^{(n)} &= (1 - r/r_c) F_i^{(n)}(x^a) - 2 \int_r^{r_c} \mathrm{d}r' \int_{r'}^{r_c} \mathrm{d}r'' \hat{R}_{ri}^{(n)}\\ g_{ij}^{(n)} &= -2 \int_r^{r_c} \mathrm{d}r' \frac{1}{r'} \int_0^{r'} \mathrm{d}r'' \hat{R}_{ij}^{(n)}, \end{split}$$
(5.41)

where the $F_a^{(n)}(x^b)$ comprise of the remaining integration functions, and the final degree of freedom; field redefinitions $\delta v_i^{(n)}$ and $\delta P^{(n)}$ at order ϵ^n . $F_i^{(n)}$ is related to redefinitions of the fluid velocity and is fixed by the isotropic gauge condition $P_a^b T_{bc} u^c = 0$. $F_{\tau}^{(n)}$ is related to redefinitions of the pressure and is fixed by defining the isotropic part of T_{ij} to be

$$T_{ij}^{\text{isotropic}} = \left(\frac{1}{\sqrt{r_c}} + \frac{P}{r_c^{3/2}}\right)\delta_{ij}$$
(5.42)

to all orders.

Applying the perturbation scheme to the seed metric yields to third order

$$ds^{2} = -rd\tau^{2} + 2d\tau dr + dx^{i}dx_{i} - 2\left(1 - \frac{r}{r_{c}}\right)v_{i}dx^{i}d\tau - 2\frac{v_{i}}{r_{c}}dx^{i}dr + \left(1 - \frac{r}{r_{c}}\right)\left[(v^{2} + 2P)d\tau^{2} + \frac{v_{i}v_{j}}{r_{c}}dx^{i}dx^{j}\right] + \left(\frac{v^{2} + 2P}{r_{c}}\right)d\tau dr \qquad (5.43) - \left[\frac{(r^{2} - r_{c}^{2})}{r_{c}}\partial^{2}v_{i} + \left(1 - \frac{r}{r_{c}}\right)\left(\frac{v^{2} + 2P}{r_{c}}\right)v_{i}\right]dx^{i}d\tau + \mathcal{O}(\epsilon^{4}),$$

which satisfies the vacuum Einstein equations if

$$r_c^{3/2} \nabla^a T_{ai}|_{\Sigma_c} = \partial_\tau v_i - r_c \partial^2 v_i + \partial_i P + v^j \partial_j v_i = \mathcal{O}(\epsilon^5), \qquad (5.44)$$

which are the Navier-Stokes equations with kinematical viscosity

$$\nu = r_c. \tag{5.45}$$

We will refer to (5.43) as the *fluid metric*, not to be confused with the metric (5.24) of the space on which the fluid evolves.

Higher order corrections to the Navier-Stokes equations follow from conservation of the stress tensor on Σ_c . Vector and scalar quantities are at odd and even orders respectively in the scaling ϵ . Accordingly, corrections to the scalar incompressibility equation appear at even orders, and to the vector Navier-Stokes equations at odd orders.

5.3 Duality in the context of holography

The defining equations in general relativity are the Einstein field equations, and in the non-relativistic limit of hydrodynamics, the Navier-Stokes equations (5.22). Each is a set of non-linear partial differential equations whose solutions exhibit fantastically varied phenomenology. When approaching any complex physical system with a view to finding solutions, it is often advantageous to consider the symmetries, intensively studied in both of these systems since their conceptions. Beyond diffeomorphisms, the search in gravity has in general been somewhat limited [223, 230], however in the presence of a spacetime isometry, the symmetry group becomes remarkably large [231], particularly for vacuum spacetimes. For symmetries of the Navier-Stokes equations see [232], and with regards to the conformal group [131,233]. In light of the fluid/gravity correspondence, one may ask whether the symmetries of these systems are linked.

In [234–241], the authors apply known symmetries of the Einstein equations to spacetimes with perfect fluid sources in the presence of one or two Killing vectors, constructing new spacetimes with fluids possessing the same equation of state. Our approach will differ from theirs in that the fluid we will consider is no longer a perfect fluid evolving in the *d*-dimensional bulk spacetime, but is a holographically dual dissipative fluid on a flat (d - 1)-dimensional hypersurface. Meanwhile, rather than employing the symmetries of Einstein gravity with a perfect fluid source, we will employ invariance transformations of the vacuum Einstein equations and holographically project these to the dual fluid to find invariance transformations of the Navier-Stokes fluid. It is in this way that we look for dualities between solutions to the Navier-Stokes equations arising from dualities between solutions to the vacuum Einstein equations: dual metrics yield dual fluid configurations.

To do this, we will will impose an isometry on the fluid metric, and then act on the metric with the generalised Ehlers group defined in terms of this isometry, which does not in general generate vacuum metrics from vacuum metrics, but does contain such transformations. At the same time, we will preserve certain aspects of the fluid metric (5.43) such as the embedded flat background on which the fluid evolves. Despite the fact that we will not have explicitly demanded that the transformed metric is vacuum, we will find that it is regardless, and that no additional constraints are required for the transformed fluid to satisfy the incompressible Navier-Stokes equations. We will discuss this unexpected result in §5.4.5.

5.4 The generalised Ehlers transformation in the fluid/gravity correspondence

The generalised Ehlers group

We will use a solution-generating technique detailed in [196], which is similar to that in chapter 4, in that it preserves the conformally rescaled reduced metric $F\gamma$ defined in (4.3), orthogonal to a Killing vector ξ . However, it has the crucial difference that it allows the use of null Killing vectors. For Killing vector⁴ $\xi = \xi^{\mu} \partial_{\mu}$ and one-form $W = W_{\mu} dx^{\mu}$, the generalised Ehlers transformation is given by

$$g_{\mu\nu} \to h_{\mu\nu}(\xi, W, g) = \Omega^2 g_{\mu\nu} - 2\xi_{(\mu}W_{\nu)} + \frac{F}{\Omega^2}W_{\mu}W_{\nu},$$
 (5.46)

where $\Omega^2 = 1 + \xi^{\mu} W_{\mu} \ge 1$, and the transformation defines a group. As such, the group does not in general send vacuum metrics to vacuum metrics, but does contain such transformations.

5.4.1 Introducing the transformed fluid metric ansatz

The Rindler metric (5.43) dual to the incompressible Navier-Stokes fluid is defined in terms of the fluid velocity, pressure, and the hypersurface position within the bulk, $(v_i(x^a), P(x^a), r_c)$. Let us consider those transformed metrics $h(\xi, W, g)$ which have the same functional form as (5.43), but now in terms of a transformed set $(v'_i(x^a), P'(x^a), r'_c)$. On satisfying the vacuum Einstein equations on Σ'_c , now at $r = r'_c$ in the new geometry, the metric h will yield the incompressible Navier-Stokes

⁴The generalised Ehlers transformation as a group is actually defined in [196] for any vector ξ . However, we will in our analysis restrict to the assumption it is a Killing vector. We discuss this choice in §5.4.5.

equations in the transformed set

$$\partial_i v_i' = 0 \tag{5.47a}$$

$$\partial_{\tau}v'_i + \partial_i P' + v'_k \partial_k v'_i - r'_c \partial^2 v'_i = 0.$$
(5.47b)

Crucially, if (v_i, P) satisfy the Navier-Stokes equations with viscosity $\nu = r_c$, and if the metric transformation is vacuum to vacuum, the transformed velocity and pressure (v'_i, P') represent a new set of solutions for viscosity $\nu = r'_c$. We will confine our search for vacuum to vacuum metric transformations to those that lie within the generalised Ehlers map. Thus, we look for a subset of the generalised Ehlers transformation acting on the fluid metric (5.43), obeying some Killing isometry, which corresponds to solution-generating transformations of the velocity and pressure, and RG flow parametrised by r_c , of an incompressible Navier-Stokes fluid in arbitrary dimension.

Now the Rindler metric is just one fluid metric supporting flat background geometries on the boundary. Rather than demanding we preserve the *full* form of the fluid metric, we therefore retain only some common features of such metrics; the metric gauge $g_{\mu r}$, and the flat boundary metric of the form (5.24). This data is sufficient to define both the fluid to all orders and the background upon which it evolves. The equation we wish to solve is thus

$$g(v_i, P, r_c)_{\mu\nu} \to h(\xi, W, g)_{\mu\nu} = g'(v'_i, P', r'_c)_{\mu\nu},$$
 (5.48)

where

$$g'_{\tau r} = 1 + \frac{(v')^2 + 2P'}{2r'_c}, \qquad g'_{ir} = -\frac{v'_i}{r'_c}, \qquad g'_{rr} = 0,$$
 (5.49a)

$$g'_{ab}|_{r'_c} = y'_{ab}, \quad \text{where} \quad y'_{\tau\tau} = -r'_c, \quad y'_{ai} = y_{ai}. \quad (5.49b)$$

5.4.2 Deriving the transformation on the fluid

Equation (5.49) provides us with sufficient information to derive the possible fluid transformations via the form of the one-form W. The parameters of the transformed fluid are determined, to all orders in ϵ , by $g'_{ar} = g'_{ar}|_{r'_c}$. Consequently, the fluid transformations will be given by the transformation of these components. Evaluation at r'_c is necessary in order to circumvent the ambiguity in the dual metric and provide

explicit fluid transformations. Before we specialise to the cases of null and non-null Killing vectors, we note the relations

$$\frac{F}{\Omega^2}W_{\mu} = \xi^{\nu}(g_{\mu\nu} - g'_{\mu\nu}), \qquad (5.50)$$

$$F/\Omega^2 = \xi^{\mu} \xi^{\nu} g'_{\mu\nu}, \tag{5.51}$$

valid for all Killing vectors, found by contraction of (5.46) once and twice respectively with the Killing vector.

Now, for non-null Killing vectors, the transformation (5.46) for the vanishing components of the metric $g'_{rr} = g_{rr} = 0$ fixes

$$W_r = 2\alpha \xi_r \Omega^2 / F$$
 where $\alpha = 0, 1.$ (5.52)

Also, one may obtain an expression for W_a by contracting (5.46) with the boundary indices (a, b, ...) of the Killing vector⁵:

$$W_a = \frac{\Omega^2 \xi^r (g_{ar} - \xi_a \xi_r / F) + \xi^b g'_{ab}}{-F / \Omega^2 + (1 - 2\alpha) \xi^r \xi_r} + \frac{\Omega^2 \xi_a}{F}.$$
(5.53)

Note: this expression is uniquely defined only at the hypersurface Σ'_c of the transformed geometry, where we have defined the form of g'_{ab} and W_a becomes independent of the transformed fluid velocity and pressure. We also have

$$\xi^a(g'_{ar} - (1 - 2\alpha)g_{ar}) = 0, \qquad (5.54)$$

derived by comparing (5.50) and (5.52). Inserting W_r (5.52) and W_a (5.53) into the generalised Ehlers transformation (5.46) and employing (5.54) and (5.51), one finds, on evaluating at r'_c ,

$$g'_{ar} = \left[\frac{Fg_{ar} + \xi_r((1-2\alpha)\xi^b y'_{ab} - \xi_a)}{\xi^c \xi^d y'_{cd} + (1-2\alpha)\xi^r \xi_r}\right]_{r'_c}.$$
(5.55)

Meanwhile, for Killing vectors null at r'_c : $F|_{r'_c} = 0$, we find (5.50) gives

$$g'_{ar} = g_{ar} + \left[\frac{\xi^b (g_{ab} - y'_{ab})}{\xi^r}\right]_{r'_c},$$
(5.56)

⁵Here and in what follows $\xi_{\mu} = g_{\mu\nu}\xi^{\nu}$, i.e. indices are raised and lowered with the metric g and never with g'.

alongside the requirement the Killing vector remains null in the transformed geometry $\xi^{\mu}\xi^{\nu}g'_{\mu\nu} = 0.$

5.4.3 Energy scaling invariance from an isometry into the bulk

We begin with an example of a (null) Killing vector into the bulk, $\xi = \xi^r(x^\mu)\partial_r$. The Killing equations require the integrability condition

$$\partial_{[i}v_{j]} = \mathcal{O}(\epsilon^4), \tag{5.57}$$

while transformation (5.56) yields $g'_{ar} = g_{ar}$, or

$$v'_{i} = \frac{r'_{c}}{r_{c}}v_{i} \qquad \frac{1}{2}(v')^{2} + P' = \frac{r'_{c}}{r_{c}}\left(\frac{1}{2}v^{2} + P\right), \qquad (5.58)$$

which is exact to all orders in ϵ . Since rescaling r_c allows for renormalisation group flow of the hypersurface position into the bulk, it is unsurprising that this results in a rescaling of the fluid energy $v^2/2+P$. That this rescaling is exact to all order appears however to be non-trivial. If one were to calculate the integrability condition for the Killing vector order by order in ϵ , then solutions to these equations along with the incompressible Navier-Stokes equations to the corresponding order, will satisfy the exact rescaling invariance (5.58) to all orders.

For example, at leading order, the general solution to the integrability condition and INS equations is potential flow defined in terms of a potential $q(x^a)$,

$$v_i = \partial_i q, \qquad P = -\partial_\tau q - \frac{1}{2}(\partial_i q)(\partial_i q) + P_0(\tau), \tag{5.59}$$

where $P_0(\tau)$ is an arbitrary function of τ , and the INS equations reduce to

$$\partial^2 q = 0. \tag{5.60}$$

Meanwhile, the fluid transformation (5.58) becomes

$$q \to \frac{r_c'}{r_c} q \qquad P_0 \to \frac{r_c'}{r_c} P_0,$$
 (5.61)

and thus it is trivial that the pair (v'_i, P') satisfy the fluid equations with if (v_i, P) do so due to the scale invariance of the Laplace equation (5.60) in q. It is interesting to consider the problems of existence, uniqueness and regularity of the Navier-Stokes in this case. Vanishing mean square vorticity ensures the class of solutions (v_i, P) generated by (5.61) are regular. With respect to existence, the kinetic energy v^2 scales by a factor $(r'_c/r_c)^2$ and thus is bounded for finite r'_c and smooth q.

The timelike Killing vector

One might expect, in the presence of a timelike Killing vector $\xi = \partial_{\tau}$ (it is sufficient for this discussion to consider stationary solutions) a duality of the form

$$v'_i = -v_i, \qquad P' = P, \qquad r'_c = -r_c$$
 (5.62)

enacting time-reversal of the fluid, but this is not the case. This is explained by noting that time-reversal is enacted by redefining the viscosity by $\nu = \pm r_c$, see [182], rather than by changing r_c itself. This is because sending $r_c \rightarrow -r_c$ brings the fluid outside the causal region of the spacetime.

5.4.4 Fixed viscosity \mathbb{Z}_2 transformations

We turn to fixed boundary (viscosity) transformations, where $r'_c = r_c$. If $\alpha = 0$, or for Killing vectors null at the dual boundary, one recovers the identity. For non-null Killing vectors with $\alpha = 1$, one finds

$$g'_{ar} = g_{ar} - 2\xi_r \left[\frac{\xi^b y_{ab} - \xi^r g_{ar}}{\xi^c \xi^d y_{cd} - \xi^r \xi_r} \right]_{r_c},$$
(5.63)

which defines a \mathbb{Z}_2 group.

Spacelike Killing vectors

Consider a generic space-like Killing vector $\xi = \xi^k \partial_k$. Under (5.63), the pressure is preserved, while the velocity transforms as

$$v_{i}' = v_{i} - 2 \left[\xi_{i} \frac{\sum_{k} \xi^{k} v_{k}}{\sum_{j} (\xi^{j})^{2}} \right]_{r_{c}}, \qquad (5.64)$$

which is a reflection in the hyperplane normal to the Killing vector (evaluated at the hypersurface) and containing the point x^i at which the velocity is defined.

Translational isometry

Consider $\xi = c_k \partial_k$ where the constants c_k are normalised to $\sum_k c_k^2 = 1$. The corresponding isometries are $c_k \partial_k v_i = 0$ and $c_k \partial_k P = 0$. The pressure is preserved, while the velocity transforms as

$$v_i' = v_i - 2c_i c_k v_k. (5.65)$$

The incompressibility condition

$$\partial_i v_i' = \partial_i v_i - 2c_i c_k \partial_i v_k = 0 \tag{5.66}$$

and Navier-Stokes equations

$$\partial_{\tau}v'_{i} + \partial_{i}P' + v'_{k}\partial_{k}v'_{i} - r_{c}\partial^{2}v'_{i} = (\delta_{ik} - 2c_{i}c_{k})(\partial_{\tau}v_{k} + \partial_{k}P + v^{j}\partial_{j}v_{k} - r_{c}\partial^{2}v_{k}) + 2c_{i}c_{k}\partial_{k}P - 2c_{j}v_{j}c_{k}\partial_{k}(v_{i} - 2c_{i}c_{l}v_{l}) = 0$$

$$(5.67)$$

are satisfied by the INS equations in the original fluid parameters along with the isometries.

Rotational isometry

Consider a Killing vector $\xi = -x_2\partial_1 + x_1\partial_2$ corresponding to a rotational isometry in the fluid. In cylindrical coordinates

$$x_1 = \rho \cos \theta, \qquad x_2 = \rho \sin \theta \qquad x_k = x_k \quad \forall k > 2,$$
 (5.68)

where the Killing vector becomes $\xi = \partial_{\theta}$, the isometries are

$$\partial_{\theta} v_1 = -v_2 \qquad \partial_{\theta} v_2 = v_1 \qquad \partial_{\theta} v_k = 0 \quad \forall k > 2 \qquad \partial_{\theta} P = 0. \tag{5.69}$$

Solutions satisfy

$$v_1 = x_1 \mu - x_2 \eta \qquad v_2 = x_2 \mu + x_1 \eta, \tag{5.70}$$

where $\mu = \mu(\tau, x_{k>2}, \rho), \eta = \eta(\tau, x_{k>2}, \rho)$, with transformed velocities

$$v_1' = x_1 \mu + x_2 \eta$$
 $v_2' = x_2 \mu - x_1 \eta.$ (5.71)

That is, the transformation sends $\eta \to -\eta$ (equivalently $\theta \to -\theta$). The incompressible Navier-Stokes equations for the original fluid may be expressed as

$$0 = (2 + \rho \partial_{\rho})\mu + \sum_{k>2} \partial_k v_k \tag{5.72a}$$

$$0 = \partial_{\tau}\mu + \rho^{-1}\partial_{\rho}P + \mu^{2} - \eta^{2} + \mu\rho\partial_{\rho}\mu + \sum_{k>2} v_{k}\partial_{k}\mu$$
(5.72b)

$$-r_c \left(3\rho^{-1}\partial_\rho \mu + \partial_\rho^2 \mu + \sum_{k>2} \partial^k \partial_k \mu \right)$$

$$0 = \partial_{\tau}\eta + 2\mu\eta + \mu\rho\partial_{\rho}\eta + \sum_{k>2} v_k\partial_k\eta - r_c\left(3\rho^{-1}\partial_{\rho}\eta + \partial_{\rho}^2\eta + \sum_{k>2}\partial^k\partial_k\eta\right)$$
(5.72c)

$$0 = \partial_{\tau} v_j + \partial_j P + \mu \rho \partial_{\rho} v_j + \sum_{k>2} v_k \partial_k v_j - r_c \rho^{-1} \partial_{\rho} (\rho \partial_{\rho} v_j) \qquad \forall j > 2,$$
(5.72d)

It is clear from the parity of these equations in η that if there exists a fluid solution defined in terms of a pair (μ, η) by (5.70), then there also exists a solution parametrized by the pair $(\mu, -\eta)$. That is, the dual fluid satisfies the INS equations.

We provide an example with the three-dimensional fluid solution

$$v_1 = A\left(x_1 - x_2 e^{-2A(\tau - \tau_0)}\right) \qquad v_2 = A\left(x_2 + x_1 e^{-2A(\tau - \tau_0)}\right) \tag{5.73a}$$

$$v_3 = B \exp\left(4A(\tau - \tau_0) + \frac{A\rho^2}{2r_c}\right) - e^{2A\tau} \int^{\tau} d\tau' \, q(\tau') e^{-2A\tau'} - 2Ax_3 \qquad (5.73b)$$

$$P = \frac{1}{2}A^2\rho^2(e^{-4A(\tau-\tau_0)} - 1) + q(\tau)x_3 - 2A^2x_3^2$$
(5.73c)

which satisfies the isometries (5.69) (note that this is in general an unphysical solution, presented primarily for illustrative purposes). Here, A, B and τ_0 are arbitrary non-vanishing constants and $q(\tau)$ is an arbitrary function of time. The duality is equivalent to sending $\tau_0 \rightarrow \tau_0 + i\pi/2A$.

5.4.5 Are the metric transformations vacuum to vacuum?

We have found solution-generating transformations of an INS fluid by applying the generalised Ehlers transformation with respect to a Killing vector, while requiring a particular ansatz for the the transformed metric at the hypersurface. However, that this procedure has provided us with solution-generating transformations of the dual fluid is somewhat unexpected. After all, we have not yet demanded that the metric transformation takes vacuum metrics to vacuum metrics. Here, we review the metric transformations and whether they are in fact vacuum to vacuum. We do this in the case of four bulk dimensions by considering the potential formalism in 4 (in particular §4.2.1).

However, before we do this, we recall that the generalised Ehlers transformation (5.46) was defined in the literature for arbitrary ξ , in particular that it need not be a Killing vector. We briefly discuss whether ξ need be a Killing vector for our fluid transformations to yield valid INS fluid solutions.

Spacelike Killing vector $\xi^i \partial_i$

For the case $\xi = c_k \partial_k$, we can determine whether ξ need be a Killing vector from the equations of motion (5.66) and (5.67) in the transformed variables. A simple calculation in the frame where $c_k = \delta_k^1$ determines that the INS in the transformed variables are satisfied only if v_i and P are independent of x^1 . Since there is no other dependence of the fluid metric on x^1 , this implies that ∂_1 must be a Killing vector in the bulk geometry. We have not completed the corresponding calculation for the rotational vector $\xi = -x_2\partial_1 + x_1\partial_2$, since our construction of fluid solutions, along with our presentation of the INS equations, relies on $-x_2\partial_1 + x_1\partial_2$ being a Killing vector. However, we expect an equivalent result: that the vector ξ must be a Killing vector in the bulk geometry for the transformed fluids to satisfy the INS equations. We now discuss whether the transformed metric is vacuum.

In this case, the norm F and twist ω are not defined unambiguously for the transformed metric, and so we cannot determine whether the transformations are vacuum to vacuum using these. However, the fluid transformations acts as a reflection in the isometry direction—e.g. $\xi = \partial_1$ sends $v_1 \to -v_1$. It is quite reasonable to expect that this is also the case for the metric—simply a coordinate reflection and therefore a (somewhat trivial) symmetry of the vacuum Einstein equations.

A Killing vector into the bulk $\xi = \xi^r \partial_r$

Again, we first determine whether ξ need be a Killing vector for (5.58) to be a valid fluid transformation. One finds that to leading order in ϵ , the transformed fluid satisfies the INS equations only if $\partial_{[i}v_{j]}\partial_{[i}v_{j]}$ vanishes (assuming finite r_c). This is true if and only if $\partial_{[i}v_{j]}$ vanishes, which is the integrability condition for ξ being a Killing vector. That is, the transformed fluid is once again a fluid solution only if ξ is a Killing vector. We now discuss whether the transformed metric is vacuum.

The transformed potentials are in this case determined unambiguously for the

transformed metric. The Killing vector is null to all orders for both the metric and the transformed metric by the gauge condition $g_{rr} = 0$. The twist can also be determined by noting

$$0 = \xi^{\mu}\omega_{\mu} = \xi^{\mu}\partial_{\mu}\chi = \xi^{r}\partial_{r}\chi \qquad \Rightarrow \qquad \chi = \chi|_{r_{c}}$$
(5.74)

and similarly $\chi' = \chi'|_{r'_c}$. To leading order, a direct calculation yields $\partial_{\mu}\chi = \partial_{\mu}\chi' = 0$. Meanwhile, the metric transformation (5.46) preserves the conformally rescaled reduced metric $\hat{\gamma} = F\gamma$, which is here

$$\hat{\gamma}_{\mu\nu} = -g_{\mu r}g_{\nu r}.\tag{5.75}$$

Collecting these results, the transformation we encounter in $\S5.4.3$, corresponds to⁶

$$g'_{\mu r} = g_{\mu r} \qquad F' = F \qquad \partial_{\mu} \chi' = \partial_{\mu} \chi, \tag{5.76}$$

which corresponds to a trivial invariance transformation of the vacuum field equations. However, we saw in §4.1 that for non-null Killing vectors, trivial invariance transformations correspond to redefinitions of the coordinate defined by the Killing vector, e.g. τ for $\xi = \partial_{\tau}$. We therefore expect that, under the constraint (5.57), the transformation (5.58) applied to the fluid metric is equivalent to a redefinition of the radial coordinate r, though we do not offer a proof here. Whether these results hold true in greater than four bulk dimensions has not been determined.

Our brief analysis therefore suggests that all the fluid transformations we have encountered, even if non-trivial from the fluid perspective, are trivial from the gravity perspective.

5.4.6 Extension to magnetohydrodynamics

The fluid/gravity correspondence has been extended to include various fields and matter content in the bulk spacetime. We look now to applying the solutiongenerating technique of this section to Einstein-Maxwell theory dual to magnetohydrodynamics (MHD). We use the metric as given in [193], where they use units with $8\pi G = 1$. The bulk dynamics are governed by the Einstein-Maxwell equations

⁶One may ask why we have not uncovered the map $g'_{\mu r} = -g_{\mu r}$. This is simply because this map would change the metric signature, while the generalised Ehlers transformation is shown in [196] to preserve metric signature.

of motion

$$R_{\mu\nu} = \frac{1}{4\pi} \left(F_{\mu\rho} F_{\nu}{}^{\rho} - \frac{1}{2(d-2)} g_{\mu\nu} F_{\sigma\rho} F^{\sigma\rho} \right) \qquad \nabla^{\mu} F_{\mu\nu} = 0.$$
(5.77)

The field strength additionally satisfies the Bianchi identity

$$\mathrm{d}F = 0. \tag{5.78}$$

Meanwhile, the MHD equations are given, for magnetic conductivity $1/4\pi r_c$ and viscosity r_c , by

$$\partial_{\tau} v_i + v^j \partial_j v_i + \partial_i \left(P - \frac{1}{16\pi} \frac{d}{d-2} f^2 \right) - r_c \partial^2 v_i = -\partial^j \pi_{ji}$$
$$\pi_{ji} = \frac{1}{4\pi} \left(f_{jl} f_{il} - \frac{1}{4} f^2 \delta_{ji} \right) \qquad f_{\tau i} = -r_c \partial_j f_{ij} - v^j f_{ji} \qquad \partial_{[a} f_{bc]} = 0 \qquad \partial_i v_i = 0.$$
(5.79)

A similar metric expansion to that outlined in $\S5.2.1$ is employed here, which uses the invariance of the equations under the hydrodynamic scaling (5.38), (5.35) with

$$f_{ij} \sim \epsilon \qquad f_{\tau i} \sim \epsilon^2,$$
 (5.80)

and the resulting metric and field strength are given by

$$g_{\mu\nu}dx^{\mu}dx^{\nu} = -rd\tau^{2} + 2d\tau dr + dx^{i}dx_{i} - 2\left(1 - \frac{r}{r_{c}}\right)v_{i}dx^{i}d\tau - 2\frac{v_{i}}{r_{c}}dx^{i}dr + \left(1 - \frac{r}{r_{c}}\right)\left[(v^{2} + 2P)d\tau^{2} + \frac{v_{i}v_{j}}{r_{c}}dx^{i}dx^{j}\right] + \left(\frac{v^{2}}{r_{c}} + \frac{2P}{r_{c}}\right)d\tau dr - \frac{1}{16\pi(d-2)}\left(1 - \frac{r}{r_{c}}\right)^{2}f^{2}d\tau^{2} + \frac{1}{2\pi r_{c}}\left(1 - \frac{r}{r_{c}}\right)\pi_{ij}dx^{i}dx^{j} - \frac{(r^{2} - r_{c}^{2})}{r_{c}}\partial^{2}v_{i}dx^{i}d\tau + \mathcal{O}(\epsilon^{3})$$
(5.81)

$$r_c F = \frac{1}{2} f_{ij} \mathrm{d}x^i \wedge \mathrm{d}x^j + f_{i\tau} \mathrm{d}x^i \wedge \mathrm{d}\tau - \partial_j f_{ij} \mathrm{d}x^i \wedge \mathrm{d}r + \mathcal{O}(\epsilon^3), \qquad (5.82)$$

which satisfy the Einstein-Maxwell equations to order $\mathcal{O}(\epsilon^3)$ provided (5.79) hold.

Bulk Killing vector

The Killing equations for a Killing vector $\xi^r \partial_r$ are highly restrictive. In particular, they demand that f_{ij} vanishes. The MHD equations then implies that $f_{\tau i}$ vanishes. We therefore recover the unforced INS equations, and the scaling invariance of §5.4.3.

Fixed viscosity \mathbb{Z}_2 transformation

Transformation (5.63) with the choice $\xi = \partial_1$ again yields the simple reflection of the velocity (5.65) with $c_k = \delta_k^1$:

$$v_1 \to -v_1 \qquad v_i \to v_i, \tag{5.83}$$

where hatted indices omit 1. The pressure is preserved. Rather than looking to determine the bulk or boundary field strength transformations explicitly, we can look to the MHD equations which can be expanded as

$$\partial_{[\hat{c}}f_{\hat{a}\hat{b}]} = 0 \qquad \partial_{[\hat{a}}f_{\hat{b}]1} = 0 \qquad \partial_{\hat{i}}v^{\hat{i}} = 0$$
 (5.84)

$$f_{\tau 1} = -r_c \partial_{\hat{j}} f_{1\hat{j}} - v^{\hat{j}} f_{\hat{j}1} \qquad f_{\tau \hat{\imath}} = -r_c \partial_{\hat{j}} f_{\hat{\imath}\hat{\jmath}} - v^{\hat{j}} f_{\hat{\jmath}\hat{\imath}} - v^1 f_{1\hat{\imath}}$$
(5.85)

$$\partial_{\tau}v_1 + v^{\hat{j}}\partial_{\hat{j}}v_1 - r_c\partial^{\hat{j}}\partial_{\hat{j}}v_1 = -\frac{1}{4\pi}\partial^{\hat{j}}\left(f_{\hat{j}\hat{k}}f_{1\hat{k}}\right)$$
(5.86)

$$\partial_{\tau} v_{\hat{\imath}} + v^{\hat{\jmath}} \partial_{\hat{\jmath}} v_{\hat{\imath}} - \partial_{\hat{\imath}} \left(P - \frac{1}{16\pi} \frac{d}{d-2} f_{\hat{l}\hat{k}} f_{\hat{l}\hat{k}} \right) - r_c \partial^{\hat{\jmath}} \partial_{\hat{\jmath}} v_{\hat{\imath}} = -\frac{1}{4\pi} \partial^{\hat{\jmath}} \left(f_{\hat{\jmath}\hat{k}} f_{\hat{\imath}\hat{k}} - f_{\hat{l}\hat{k}} f_{\hat{l}\hat{k}} \delta_{\hat{\jmath}\hat{\imath}} \right).$$
(5.87)

We see that a reflection (5.83) with (5.85) requires a reflection

$$f_{\hat{a}1} \to -f_{\hat{a}1} \qquad f_{\hat{a}\hat{b}} \to f_{\hat{a}\hat{b}} \tag{5.88}$$

while all other equations are invariant under (5.83) and (5.88). This transformation is simply equivalent to reflection of the coordinate $x^1 \to -x^1$. One can find transformations allowed by other spatial Killing vectors by suitable rotations of the x^i coordinates.

On projecting solution-generating transformations to the dual fluid, we had hoped to arrive at some non-trivial fluid transformations deriving from non-trivial symmetries in the vacuum field equations, such as the $F \to 1/F$ of Buchdahl. For example, perfect fluids with an isometry along x^1 are invariant under $v_1 \to 1/v_1$. However, we have arrived, particularly for spacelike Killing vectors, at rather trivial transformations. It would be interesting to see if extensions of this work could uncover further non-trivial physics, and if not, whether this trivial result was an inevitable product of some unknown property of the fluid/gravity correspondence.

Chapter 6

Conclusion and outlook

6.1 Double field theory and U-duality-invariant M-theory

We have presented new derivations of the equations of motion for the generalised coset metrics of DFT and SL(5)-invariant extended geometry from their respective actions. In the DFT case, we offered a pp-wave solution in the doubled geometry, which reduces under two different solutions to the section condition to the F1 string and the spacetime pp-wave. The Goldstone modes of the DFT pp-wave are found, in the string case, to be governed by the equations of motion of the duality-symmetric string of Duff [10] and Tseytlin [11, 12]. In the SL(5) geometry, we offer a pp-wave generalised metric solution which reduces in the supergravity frame to the M2-brane.

We have looked at the $O(d, d, \mathbb{R})$ -covariant string (gauged) σ -model actions of [11,12,58,60], and applied a Buscher-type procedure with both one and two Lagrange multipliers. With one Lagrange multiplier, the resulting action is not of the form of a DFT σ -model. For two Lagrange multipliers, we find the expected $O(1, 1, \mathbb{Z})$ T-duality transformation.

There are plenty of avenues of further research in the duality-invariant geometries. U-duality-invariant theories have been constructed in all cases where the dimension of the duality group is finite, i.e. $3 \le d \le 8$. However, the full equations of motion have been constructed only for d = 4, and it would be relatively simple to extend our chain rule approach to higher dimensional geometries, where the Lagrangians, vielbein, and Y-tensors are known [97, 102]. However, we note that one can find *some* solutions in these geometries purely by variation of the Lagrangian with respect to the generalised metric treated as an arbitrary symmetric object. For example, in the SL(5) geometry, solving $\mathcal{K} = 0$. Indeed, the SL(5) pp-wave generalised metric does this. That the generalised metric takes the appropriate coset form only projects out, via the appropriate projector P, some components of \mathcal{K} —that is while all solutions satisfy $P\mathcal{K} = 0$, some also satisfy $\mathcal{K} = 0$. A related example of note is the self-dual solution in E_7 exceptional field theory constructed in [118], which reproduces the complete 1/2-BPS spectrum of ten- and eleven-dimensional supergravity. However, determining the full equations of motion would be required in, for example, a Goldstone analysis of these solutions.

6.2 Solution-generating techniques in supergravity, double field theory and beyond

In our study of symmetries of NS-NS supergravity with one non-null isometry, we decompose all field content into fields living on a codimension one submanifold defined by the Killing vector. Specifically, we define a set of *p*-form potentials and a submanifold reduced metric. We restrict to the cases where these potentials are scalars, and determine the equations of motion of these potentials and the reduced metric. Via the corresponding effective actions, we determine a set of symmetry groups which preserve the submanifold metric. In static Einstein-dilaton theory we find a non-trivial $\mathbb{Z}_2 \times \mathbb{Z}_2$ symmetry group, and in five-dimensional static magnetic supergravity, a non-trivial $SL(2, \mathbb{R}) \times \mathbb{Z}_2$ group.

We looked at how similar techniques might be applied to DFT with an isometry. We discussed which components of the generalised metric to transform, and which to keep fixed when looking for symmetry groups, and looked at a specific case corresponding in the supergravity frame to static Einstein-dilaton gravity. In this case, we recovered only the \mathbb{Z}_2 T-duality we encountered in Einstein-dilaton theory, which is of course manifest in DFT. We would like to see how the remaining \mathbb{Z}_2 appears in the DFT context. Moreover, it would be interesting to see this extended to the full DFT case, especially in a covariant form. One could perhaps decompose the DFT fields in to potentials analogous to those for Einstein gravity in §4.1, by employing the studies of DFT geometry in [69–74], where analogues of the covariant derivative and curvature tensors are constructed.

For example, in [218] the author constructed an effective action for stationary four-dimensional NS-NS supergravity containing two sigma models, each with $SL(2,\mathbb{R})$ invariance. One group corresponded to S-duality and the other to an Ehlers-type symmetry. Moreover, they point out that these sigma models can be mapped into each other by an $O(1, 1, \mathbb{Z})$ T-duality transformation. That is, in that case, the Ehlers symmetry is T-dual to S-duality. It would be interesting to see how this would play out within DFT. Since the theory is T-duality invariant in the presence of an isometry, we ask if Ehlers-type transformations would appear on the same footing as S-duality, where one would rotate between the two for different solutions to the section condition. Perhaps this is one way of recovering U-duality from DFT.

Returning to solution-generating transformations in Einstein gravity coupled to form fields, a much greater goal would be to be able to find the symmetry groups for any spacetime dimension and with any form-field content (it would of course be ideal to verify if these methods are valid for null Killing vectors, not only spacelike/timelike). The potential space line element method in §4.2 clearly would not suffice, since the potentials are no longer scalars, though it would be interesting to see if one can construct appropriate generalisations of this method.

One could of course introduce more isometries, e.g. stationary-axisymmetric systems relevant to stars and black holes. This includes the generalisation of the Ehlers symmetry to the $SL(d-2,\mathbb{R})$ invariance of *d*-dimensional vacuum Einstein gravity with (d-3) isometries. One could also consider solution-generating isometries in other gravitational theories; the presence of matter, for example a perfect fluid [234–241]; form fields with Lagrangian contributions beyond the quadratic term, such as the Chern-Simons term appearing in eleven-dimensional supergravity (3.1); and Lanczos-Lovelock gravity [242,243], where the Ricci curvature in the Lagrangian is generalised to

$$L = \sum_{k=0}^{k_{\max}} a_k \delta^{\rho_1}_{[\mu_1} \delta^{\sigma_1}_{\nu_1} \dots \delta^{\rho_k}_{\mu_k} \delta^{\sigma_k}_{\nu_k]} \prod_{l=1}^k R^{\mu_l \nu_l}{}_{\nu_l \sigma_l},$$
(6.1)

where $k_{\text{max}} \leq (d-1)/2$ and a_k are dimensionful constants. It was shown by Brustein and Medved [244] that Lovelock gravity is equivalent to Einstein gravity coupled to forms fields. It would be interesting to see how this works with regards to the potentials which appear in the presence of a Killing vector. After all, our work in §4.2 shows how the metric and Riemann tensor can, via (4.19) and (4.4) be decomposed in terms of the twist (4.5), norm and reduced metric (4.3). On the other hand, equation (4.45) illustrates how form fields decompose into two potentials each. Brustein and Medved's work then suggests that the potentials encoding the metric in Lovelock gravity in some way correspond to the potentials encoding form fields in Einstein gravity.

6.3 Solution-generating transformations in the fluid/gravity correspondence

We have explored how solution-generating transformations of the Einstein equations in the presence of a Killing vector may be mapped to the codimension one hydrodynamics holographically dual under the fluid/gravity correspondence. Our focus has been on the incompressible Navier-Stokes fluid dual to vacuum Rindler spacetime, where we have uncovered a selection of fluid transformations.

Firstly, we discovered a linear scaling of the fluid velocity and total energy with the viscosity, corresponding in the gravity dual to renormalisation group flow of the hypersurface through the bulk. This scaling invariance is exact, in that it receives no corrections at higher orders in the hydrodynamic expansion. It would be relatively simple to solve the Killing equations for the fluid metric at higher orders and check the validity of the scaling invariance at the corresponding order in the INS equations. Higher order corrections for the metric and INS equations are given in [183]. Secondly, we find that spacelike isometries in the fluid yield somewhat trivial \mathbb{Z}_2 transformations of the fluid velocity (it would be interesting to see if other Killing vectors, such as those with components along both radial and hypersurface directions, might yield more non-trivial results). Explicit examples are given of reflection-like symmetries for translational and rotational isometries in the fluid. Finally, we determine that all of our transformations are trivial from the point of view of the bulk spacetime.

There are plenty of immediate extensions to this work from the a large literature on gravitational systems dual to hydrodynamics, such as: fluid flow on a sphere dual to the Schwarzschild black hole [192]; cyclonic flow dual to the Kerr-Newman AdS₄ black hole [245]; (non-)conformal relativistic hydrodynamics dual to vacuum Einstein gravity [152, 178–181]; forced fluids dual to non-vacuum gravity, such as bulk ideal fluids [158]. In particular, we ask if it is possible to transform between fluid solutions on flat backgrounds and those on curved backgrounds. Not only would this allow us to find fluid flows on curved backgrounds, where the equations are often more difficult to solve than the INS equations, one could possibly generate non-trivial fluid flows on flat space from trivial flows on curved space.

Extensions further afield include condensed matter systems dual to gravity (see [246] for a review). One could even look to solution-generating techniques in holog-raphy beyond the hydrodynamic limit. Indeed, in the epilogue of [177], the authors consider the following. If the Einstein equations in the long-wavelength limit are

dual to fluid dynamics, then perhaps the there may be a similar connection between the full Einstein equations and a strong coupling analogue of the Boltzmann transport equations, describing the thermodynamics of a system out of thermal equilibrium. Essentially, we wish to open up the use of gravitational solution-generating symmetries in holography.

Appendix A

Conventions and useful formulae

 $\label{eq:Metric signature} \mbox{ Lorentzian: } (-++\ldots+), \mbox{ Riemannian: } (++\ldots+)$

(Anti)symmetrization

$$A_{(\mu_1\dots\mu_p)} = \frac{1}{p!} \sum_{\pi} A_{\mu_{\pi(1)}\dots\mu_{\pi(p)}}$$
(A.1)

$$A_{[\mu_1...\mu_p]} = \frac{1}{p!} \sum_{\pi} \delta_{\pi} A_{\mu_{\pi(1)}...\mu_{\pi(p)}},$$
(A.2)

where the sum is taken over all permutations, π , of 1, ..., p and δ_{π} is +1 for even permutations and -1 for odd permutations.

Alternating symbol and tensor For totally antisymmetric symbol

$$\epsilon^{\mu_1...\mu_d} = \epsilon_{\mu_1...\mu_d} = \epsilon_{[\mu_1...\mu_d]}, \qquad \epsilon_{012...d} = 1,$$
 (A.3)

on a d-dimensional space, we define the totally antisymmetric tensor on Lorentzian space with metric g by

$$\hat{\epsilon}_{\mu_1\dots\mu_d} = \sqrt{-\det(g)} \,\epsilon_{\mu_1\dots\mu_d},\tag{A.4}$$

whose indices can be raised and lowered with g, which gives

$$\hat{\epsilon}^{\mu_1\dots\mu_d} = \frac{-1}{\sqrt{-\det(g)}} \,\epsilon^{\mu_1\dots\mu_d}.\tag{A.5}$$

The antisymmetric tensor is annihilated by the covariant derivative. It contracts as

$$\hat{\epsilon}^{\mu_1\dots\mu_r\nu_{r+1}\dots\nu_d}\hat{\epsilon}_{\rho_1\dots\rho_r\nu_{r+1}\dots\nu_d} = -r!(d-r)!\delta^{\mu_1}_{[\rho_1}\dots\delta^{\mu_r}_{\rho_r]}.$$
(A.6)

A.1 Differential forms

The results in this section are valid for *d*-dimensional Lorentzian spacetimes with metric g, for arbitrary forms A, B, C (where lowercase Roman subscripts indicate rank), and vectors v.

Definitions

$$(\star A)_{\mu_1\dots\mu_{d-p}} = \frac{1}{p!} \hat{\epsilon}_{\mu_1\dots\mu_{d-p}}{}^{\nu_1\dots\nu_p} (A_p)_{\nu_1\dots\nu_p}$$
(A.7)

$$(A_p \wedge B_q)_{\mu_1 \dots \mu_p \nu_1 \dots \nu_q} = \frac{(p+q)!}{p!q!} (A_p)_{[\mu_1 \dots \mu_p} (B_q)_{\nu_1 \dots \nu_q]}$$
(A.8)

$$(\iota_{\upsilon}A_p)_{\mu_2\dots\mu_p} = \upsilon^{\mu_1}(A_p)_{\mu_1\mu_2\dots\mu_p}$$
(A.9)

$$(\mathrm{d}A_p)_{\mu_1\dots\mu_{p+1}} = (p+1)\partial_{[\mu_1}(A_p)_{\mu_2\dots\mu_{p+1}]}$$
(A.10)

Musical notation is defined for vectors v and one-forms ω as

$$(\upsilon_{\flat})_{\mu} = g_{\mu\nu}\upsilon^{\mu} \qquad (\omega^{\sharp})^{\mu} = g^{\mu\nu}\omega_{\nu}. \tag{A.11}$$

Useful formulae

For the following identities, references are given in the second column in certain cases.

$$\mathcal{L}_{\upsilon}A = (\iota_{\upsilon}d + d\iota_{\upsilon})A \qquad [247, p. 202] \qquad (A.12)$$

$$\star^2 A_p = (-1)^{1+p(d-p)} A_p \qquad [247, p. 291] \qquad (A.13)$$

$$(A \wedge B) \wedge C = A \wedge (B \wedge C)$$
 [247, p. 198] (A.14)

$$A_p \wedge B_q = (-1)^{pq} B_q \wedge A_p$$
 [247, p. 198] (A.15)

$$A_p \wedge \star B_p = B_p \wedge \star A_p \qquad [247, p. 292] \qquad (A.16)$$

$$d(A \wedge B) = d(A \wedge B) + (-1)^p A \wedge d(B) \qquad [247, p. 100] \qquad (A.17)$$

$$d(A_p \wedge B_q) = dA_p \wedge B_q + (-1)^p A_p \wedge dB_q \qquad [247, p. 199] \qquad (A.17)$$

$$\iota_{\upsilon}(A_p \wedge B_q) = \iota_{\upsilon}A_p \wedge B_q + (-1)^p A_p \wedge \iota_{\upsilon}B_q \qquad [247, \, \text{p. 202}] \qquad (A.18)$$

$$\mathcal{L}_{\upsilon}(A \wedge B) = (\mathcal{L}_{\upsilon}A) \wedge B + A \wedge \mathcal{L}_{\upsilon}B \tag{A.19}$$

$$(\star d \star A_p)_{\mu_1...\mu_{p-1}} = (-1)^{pd} \nabla^{\sigma} (A_p)_{\sigma\mu_1...\mu_{p-1}}$$
(A.20)

$$\star (A_p \wedge \star A_p) = \frac{1}{p!} (-1)^{1+p(d-p)} A_{\mu_1 \dots \mu_p} A^{\mu_1 \dots \mu_p}$$
(A.21)

$$\upsilon_{\flat} \wedge \star A_p = (-1)^{d-p} \star \iota_{\upsilon} A_p \tag{A.22}$$

$$\star(\upsilon_{\flat} \wedge A_p) = (-1)^{d-p+1} \iota_{\upsilon} \star A_p \tag{A.23}$$

$$(-1)^{pd+1}(i_{\upsilon}\upsilon_{\flat})A_{p} = \left[(i_{\upsilon}\star)^{2} + (-1)^{d}(\star i_{\upsilon})^{2}\right]A_{p}$$
(A.24)

Bibliography

- [1] J. Berkeley and D. S. Berman, "The Navier-Stokes equation and solution generating symmetries from holography," *JHEP*, vol. 1304, p. 092, 2013.
- [2] J. Berkeley, D. S. Berman, and F. J. Rudolph, "Strings and Branes are Waves," *JHEP*, vol. 1406, p. 006, 2014.
- [3] P. Horava and E. Witten, "Heterotic and type I string dynamics from elevendimensions," Nucl. Phys., vol. B460, pp. 506–524, 1996.
- [4] E. Cremmer, B. Julia, and J. Scherk, "Supergravity Theory in Eleven-Dimensions," *Phys.Lett.*, vol. B76, pp. 409–412, 1978.
- [5] C. Hull and P. Townsend, "Unity of superstring dualities," Nucl. Phys., vol. B438, pp. 109–137, 1995.
- [6] N. Hitchin, "Generalized Calabi-Yau manifolds," ArXiv Mathematics e-prints, Sept. 2002.
- [7] M. Gualtieri, "Generalized complex geometry," ArXiv Mathematics e-prints, Jan. 2004.
- [8] W. Siegel, "Superspace duality in low-energy superstrings," *Phys.Rev.*, vol. D48, pp. 2826–2837, 1993.
- W. Siegel, "Two vierbein formalism for string inspired axionic gravity," *Phys.Rev.*, vol. D47, pp. 5453–5459, 1993.
- [10] M. Duff, "Duality Rotations in String Theory," Nucl. Phys., vol. B335, p. 610, 1990.
- [11] A. A. Tseytlin, "Duality Symmetric Formulation of String World Sheet Dynamics," *Phys.Lett.*, vol. B242, pp. 163–174, 1990.

- [12] A. A. Tseytlin, "Duality symmetric closed string theory and interacting chiral scalars," Nucl. Phys., vol. B350, pp. 395–440, 1991.
- [13] D. S. Berman and D. C. Thompson, "Duality Symmetric String and M-Theory," *Phys.Rept.*, vol. 566, pp. 1–60, 2014.
- [14] O. Hohm, D. Lüst, and B. Zwiebach, "The Spacetime of Double Field Theory: Review, Remarks, and Outlook," *Fortsch.Phys.*, vol. 61, pp. 926–966, 2013.
- [15] G. Aldazabal, D. Marques, and C. Nunez, "Double Field Theory: A Pedagogical Review," *Class. Quant. Grav.*, vol. 30, p. 163001, 2013.
- [16] D. S. Berman and D. C. Thompson, "Duality Symmetric Strings, Dilatons and O(d,d) Effective Actions," *Phys.Lett.*, vol. B662, pp. 279–284, 2008.
- [17] D. S. Berman, N. B. Copland, and D. C. Thompson, "Background Field Equations for the Duality Symmetric String," *Nucl. Phys.*, vol. B791, pp. 175–191, 2008.
- [18] D. S. Berman and N. B. Copland, "The String partition function in Hull's doubled formalism," *Phys.Lett.*, vol. B649, pp. 325–333, 2007.
- [19] C. Hull and B. Zwiebach, "Double Field Theory," *JHEP*, vol. 0909, p. 099, 2009.
- [20] C. Hull and B. Zwiebach, "The Gauge algebra of double field theory and Courant brackets," *JHEP*, vol. 0909, p. 090, 2009.
- [21] S. K. Kwak, "Invariances and Equations of Motion in Double Field Theory," *JHEP*, vol. 1010, p. 047, 2010.
- [22] D. C. Thompson, "T-duality Invariant Approaches to String Theory," 2010.
- [23] O. Hohm, C. Hull, and B. Zwiebach, "Background independent action for double field theory," *JHEP*, vol. 1007, p. 016, 2010.
- [24] O. Hohm, C. Hull, and B. Zwiebach, "Generalized metric formulation of double field theory," *JHEP*, vol. 1008, p. 008, 2010.
- [25] O. Hohm and S. K. Kwak, "Double Field Theory Formulation of Heterotic Strings," JHEP, vol. 1106, p. 096, 2011.

- [26] N. B. Copland, "Connecting T-duality invariant theories," Nucl. Phys., vol. B854, pp. 575–591, 2012.
- [27] I. Jeon, K. Lee, and J.-H. Park, "Double field formulation of Yang-Mills theory," *Phys.Lett.*, vol. B701, pp. 260–264, 2011.
- [28] O. Hohm and S. K. Kwak, "Massive Type II in Double Field Theory," JHEP, vol. 1111, p. 086, 2011.
- [29] C. Albertsson, S.-H. Dai, P.-W. Kao, and F.-L. Lin, "Double Field Theory for Double D-branes," *JHEP*, vol. 1109, p. 025, 2011.
- [30] I. Jeon, K. Lee, and J.-H. Park, "Stringy differential geometry, beyond Riemann," *Phys.Rev.*, vol. D84, p. 044022, 2011.
- [31] N. Kan, K. Kobayashi, and K. Shiraishi, "Equations of Motion in Double Field Theory: From particles to scale factors," *Phys.Rev.*, vol. D84, p. 124049, 2011.
- [32] N. Kan, K. Kobayashi, and K. Shiraishi, "Equations of motion in Double Field Theory: from classical particles to quantum cosmology," 2012.
- [33] O. Hohm and B. Zwiebach, "Large Gauge Transformations in Double Field Theory," JHEP, vol. 1302, p. 075, 2013.
- [34] D. Andriot, O. Hohm, M. Larfors, D. Lust, and P. Patalong, "A geometric action for non-geometric fluxes," *Phys.Rev.Lett.*, vol. 108, p. 261602, 2012.
- [35] D. Andriot, O. Hohm, M. Larfors, D. Lust, and P. Patalong, "Non-Geometric Fluxes in Supergravity and Double Field Theory," *Fortsch.Phys.*, vol. 60, pp. 1150–1186, 2012.
- [36] E. Musaev, "U-dualities in Type II string theories and M-theory," 2013.
- [37] J.-H. Park, "Comments on double field theory and diffeomorphisms," JHEP, vol. 1306, p. 098, 2013.
- [38] O. Hohm and H. Samtleben, "Gauge theory of Kaluza-Klein and winding modes," *Phys. Rev.*, vol. D88, p. 085005, 2013.
- [39] O. Hohm, W. Siegel, and B. Zwiebach, "Doubled α' -geometry," *JHEP*, vol. 1402, p. 065, 2014.

- [40] C. D. Blair, E. Malek, and A. J. Routh, "An O(D, D) invariant Hamiltonian action for the superstring," Class. Quant. Grav., vol. 31, no. 20, p. 205011, 2014.
- [41] O. Hohm and B. Zwiebach, "Double field theory at order α'," JHEP, vol. 1411, p. 075, 2014.
- [42] D. S. Berman, M. Cederwall, and M. J. Perry, "Global aspects of double geometry," JHEP, vol. 1409, p. 066, 2014.
- [43] G. Papadopoulos, "C-spaces, generalized geometry and double field theory," 2014.
- [44] C.-T. Ma, "Gauge Transformation of Double Field Theory for Open String," 2014.
- [45] M. Cederwall, "T-duality and non-geometric solutions from double geometry," *Fortsch.Phys.*, vol. 62, pp. 942–949, 2014.
- [46] U. Naseer, "A note on large gauge transformations in double field theory," 2015.
- [47] R. Blumenhagen, P. d. Bosque, F. Hassler, and D. Lust, "Generalized Metric Formulation of Double Field Theory on Group Manifolds," 2015.
- [48] A. Betz, R. Blumenhagen, D. Lüst, and F. Rennecke, "A Note on the CFT Origin of the Strong Constraint of DFT," JHEP, vol. 1405, p. 044, 2014.
- [49] O. Hohm, S. K. Kwak, and B. Zwiebach, "Unification of Type II Strings and T-duality," *Phys.Rev.Lett.*, vol. 107, p. 171603, 2011.
- [50] O. Hohm and S. K. Kwak, "N=1 Supersymmetric Double Field Theory," JHEP, vol. 1203, p. 080, 2012.
- [51] I. Jeon, K. Lee, and J.-H. Park, "Incorporation of fermions into double field theory," *JHEP*, vol. 1111, p. 025, 2011.
- [52] O. Hohm, S. K. Kwak, and B. Zwiebach, "Double Field Theory of Type II Strings," *JHEP*, vol. 1109, p. 013, 2011.
- [53] I. Jeon, K. Lee, and J.-H. Park, "Supersymmetric Double Field Theory: Stringy Reformulation of Supergravity," *Phys. Rev.*, vol. D85, p. 081501, 2012.

- [54] I. Jeon, K. Lee, J.-H. Park, and Y. Suh, "Stringy Unification of Type IIA and IIB Supergravities under N=2 D=10 Supersymmetric Double Field Theory," *Phys.Lett.*, vol. B723, pp. 245–250, 2013.
- [55] I. Jeon, K. Lee, and J.-H. Park, "Ramond-Ramond Cohomology and O(D,D) T-duality," JHEP, vol. 1209, p. 079, 2012.
- [56] W. Cho, J. Fernández-Melgarejo, I. Jeon, and J.-H. Park, "Supersymmetric gauged Double Field Theory: Systematic derivation by virtue of *Twist*," 2015.
- [57] K. Lee, "Quadratic α' -Corrections to Heterotic Double Field Theory," 2015.
- [58] C. Hull, "A Geometry for non-geometric string backgrounds," JHEP, vol. 0510, p. 065, 2005.
- [59] N. B. Copland, "A Double Sigma Model for Double Field Theory," JHEP, vol. 1204, p. 044, 2012.
- [60] K. Lee and J.-H. Park, "Covariant action for a string in "doubled yet gauged" spacetime," Nucl. Phys., vol. B880, pp. 134–154, 2014.
- [61] L. De Angelis, G. Gionti S.J., R. Marotta, and F. Pezzella, "Comparing Double String Theory Actions," *JHEP*, vol. 1404, p. 171, 2014.
- [62] R. Blumenhagen, M. Fuchs, F. Hassler, D. Lüst, and R. Sun, "Non-associative Deformations of Geometry in Double Field Theory," *JHEP*, vol. 1404, p. 141, 2014.
- [63] R. Blumenhagen, F. Hassler, and D. Lüst, "Double Field Theory on Group Manifolds," JHEP, vol. 1502, p. 001, 2015.
- [64] R. Blumenhagen and R. Sun, "T-duality, Non-geometry and Lie Algebroids in Heterotic Double Field Theory," JHEP, vol. 1502, p. 097, 2015.
- [65] C.-T. Ma, "One-loop β function of the double sigma model with constant background," *JHEP*, vol. 1504, p. 026, 2015.
- [66] F. Pezzella, "Two Double String Theory Actions: Covariance vs. Non-Covariance," 2015.
- [67] C.-T. Ma, "Boundary Conditions and the Generalized Metric Formulation of the Double Sigma Model," 2015.

- [68] D. Polyakov, P. Wang, H. Wu, and H. Yang, "Non-commutativity from the double sigma model," *JHEP*, vol. 1503, p. 011, 2015.
- [69] O. Hohm and S. K. Kwak, "Frame-like Geometry of Double Field Theory," J.Phys., vol. A44, p. 085404, 2011.
- [70] I. Jeon, K. Lee, and J.-H. Park, "Differential geometry with a projection: Application to double field theory," *JHEP*, vol. 1104, p. 014, 2011.
- [71] O. Hohm and B. Zwiebach, "On the Riemann Tensor in Double Field Theory," *JHEP*, vol. 1205, p. 126, 2012.
- [72] O. Hohm and B. Zwiebach, "Towards an invariant geometry of double field theory," J.Math.Phys., vol. 54, p. 032303, 2013.
- [73] D. S. Berman, C. D. Blair, E. Malek, and M. J. Perry, "The $O_{D,D}$ geometry of string theory," *Int.J.Mod.Phys.*, vol. A29, no. 15, p. 1450080, 2014.
- [74] M. Cederwall, "The geometry behind double geometry," *JHEP*, vol. 1409, p. 070, 2014.
- [75] D. S. Berman, E. T. Musaev, and M. J. Perry, "Boundary Terms in Generalized Geometry and doubled field theory," *Phys.Lett.*, vol. B706, pp. 228–231, 2011.
- [76] C.-T. Ma and C.-M. Shen, "Cosmological Implications from O(D,D)," *Fortsch.Phys.*, vol. 62, pp. 921–941, 2014.
- [77] H. Wu and H. Yang, "Double Field Theory Inspired Cosmology," JCAP, vol. 1407, p. 024, 2014.
- [78] G. Aldazabal, W. Baron, D. Marques, and C. Nunez, "The effective action of Double Field Theory," *JHEP*, vol. 1111, p. 052, 2011.
- [79] D. Geissbuhler, "Double Field Theory and N=4 Gauged Supergravity," JHEP, vol. 1111, p. 116, 2011.
- [80] M. Grana and D. Marques, "Gauged Double Field Theory," JHEP, vol. 1204, p. 020, 2012.
- [81] G. Dibitetto, J. Fernandez-Melgarejo, D. Marques, and D. Roest, "Duality orbits of non-geometric fluxes," *Fortsch. Phys.*, vol. 60, pp. 1123–1149, 2012.

- [82] D. Geissbuhler, D. Marques, C. Nunez, and V. Penas, "Exploring Double Field Theory," JHEP, vol. 1306, p. 101, 2013.
- [83] D. S. Berman and K. Lee, "Supersymmetry for Gauged Double Field Theory and Generalised Scherk-Schwarz Reductions," *Nucl. Phys.*, vol. B881, pp. 369– 390, 2014.
- [84] F. Hassler and D. Lüst, "Consistent Compactification of Double Field Theory on Non-geometric Flux Backgrounds," *JHEP*, vol. 1405, p. 085, 2014.
- [85] K. Lee, C. Strickland-Constable, and D. Waldram, "New gaugings and nongeometry," 2015.
- [86] B. Julia, "Group disintegrations," in Superspace and Supergravity: Proceedings of the Nuffield Workshop, Cambridge, June 16-July 12, 1980, CUP archive, 1981.
- [87] A. Coimbra, C. Strickland-Constable, and D. Waldram, "Supergravity as Generalised Geometry I: Type II Theories," *JHEP*, vol. 1111, p. 091, 2011.
- [88] A. Coimbra, C. Strickland-Constable, and D. Waldram, "Generalised Geometry and type II Supergravity," *Fortsch.Phys.*, vol. 60, pp. 982–986, 2012.
- [89] C. Hillmann, "Generalized E(7(7)) coset dynamics and D=11 supergravity," JHEP, vol. 0903, p. 135, 2009.
- [90] A. Coimbra, C. Strickland-Constable, and D. Waldram, " $E_{d(d)} \times \mathbb{R}^+$ generalised geometry, connections and M theory," *JHEP*, vol. 1402, p. 054, 2014.
- [91] A. Coimbra, C. Strickland-Constable, and D. Waldram, "Supergravity as Generalised Geometry II: $E_{d(d)} \times \mathbb{R}^+$ and M theory," *JHEP*, vol. 1403, p. 019, 2014.
- [92] H. Godazgar, M. Godazgar, and H. Nicolai, "Generalised geometry from the ground up," *JHEP*, vol. 1402, p. 075, 2014.
- [93] K. Lee, C. Strickland-Constable, and D. Waldram, "Spheres, generalised parallelisability and consistent truncations," 2014.
- [94] E. Malek, "U-duality in three and four dimensions," arXiv preprint arXiv:1205.6403, 2012.
- [95] D. S. Berman and M. J. Perry, "Generalized Geometry and M theory," JHEP, vol. 1106, p. 074, 2011.

- [96] D. S. Berman, H. Godazgar, M. Godazgar, and M. J. Perry, "The Local symmetries of M-theory and their formulation in generalised geometry," *JHEP*, vol. 1201, p. 012, 2012.
- [97] D. S. Berman, H. Godazgar, M. J. Perry, and P. West, "Duality Invariant Actions and Generalised Geometry," *JHEP*, vol. 1202, p. 108, 2012.
- [98] J.-H. Park and Y. Suh, "U-geometry: SL(5)," JHEP, vol. 1304, p. 147, 2013.
- [99] C. D. Blair, E. Malek, and J.-H. Park, "M-theory and Type IIB from a Duality Manifest Action," *JHEP*, vol. 1401, p. 172, 2014.
- [100] D. S. Berman, H. Godazgar, and M. J. Perry, "SO(5,5) duality in M-theory and generalized geometry," *Phys.Lett.*, vol. B700, pp. 65–67, 2011.
- [101] G. Aldazabal, M. Graña, D. Marqués, and J. Rosabal, "Extended geometry and gauged maximal supergravity," *JHEP*, vol. 1306, p. 046, 2013.
- [102] D. S. Berman, M. Cederwall, A. Kleinschmidt, and D. C. Thompson, "The gauge structure of generalised diffeomorphisms," *JHEP*, vol. 1301, p. 064, 2013.
- [103] D. S. Berman and F. J. Rudolph, "Branes are Waves and Monopoles," JHEP, vol. 1505, p. 015, 2015.
- [104] O. Hohm and H. Samtleben, "Exceptional field theory. II. $E_{7(7)}$," *Phys.Rev.*, vol. D89, no. 6, p. 066017, 2014.
- [105] O. Hohm and H. Samtleben, "Exceptional Field Theory I: $E_{6(6)}$ covariant Form of M-Theory and Type IIB," *Phys.Rev.*, vol. D89, no. 6, p. 066016, 2014.
- [106] O. Hohm and H. Samtleben, "Exceptional Form of D=11 Supergravity," *Phys.Rev.Lett.*, vol. 111, p. 231601, 2013.
- [107] C. D. A. Blair and E. Malek, "Geometry and fluxes of SL(5) exceptional field theory," *JHEP*, vol. 1503, p. 144, 2015.
- [108] H. Godazgar, M. Godazgar, O. Hohm, H. Nicolai, and H. Samtleben, "Supersymmetric E₇₍₇₎ Exceptional Field Theory," *JHEP*, vol. 1409, p. 044, 2014.
- [109] E. Musaev and H. Samtleben, "Fermions and supersymmetry in $E_{6(6)}$ exceptional field theory," *JHEP*, vol. 1503, p. 027, 2015.

- [110] O. Hohm and H. Samtleben, "Exceptional field theory. III. $E_{8(8)}$," *Phys.Rev.*, vol. D90, no. 6, p. 066002, 2014.
- [111] M. Cederwall and J. Rosabal, " E_8 geometry," 2015.
- [112] Y.-N. Wang, "Generalized Cartan Calculus in general dimension," 2015.
- [113] E. T. Musaev, "Exceptional Field Theory for $E_{6(6)}$ supergravity," *TSPU Bulletin*, vol. 12, pp. 198–202, 2014.
- [114] O. Hohm and Y.-N. Wang, "Tensor hierarchy and generalized Cartan calculus in $SL(3) \times SL(2)$ exceptional field theory," *JHEP*, vol. 1504, p. 050, 2015.
- [115] A. Abzalov, I. Bakhmatov, and E. T. Musaev, "Exceptional field theory: SO(5,5)," 2015.
- [116] E. Malek, "Timelike U-dualities in Generalised Geometry," JHEP, vol. 1311, p. 185, 2013.
- [117] H. Godazgar, M. Godazgar, and H. Nicolai, "Einstein-Cartan Calculus for Exceptional Geometry," JHEP, vol. 1406, p. 021, 2014.
- [118] D. S. Berman and F. J. Rudolph, "Strings, Branes and the Self-dual Solutions of Exceptional Field Theory," 2014.
- [119] A. Baguet, O. Hohm, and H. Samtleben, " $E_{6(6)}$ Exceptional Field Theory: Review and Embedding of Type IIB," 2015.
- [120] D. S. Berman, E. T. Musaev, D. C. Thompson, and D. C. Thompson, "Duality Invariant M-theory: Gauged supergravities and Scherk-Schwarz reductions," *JHEP*, vol. 1210, p. 174, 2012.
- [121] E. T. Musaev, "Gauged supergravities in 5 and 6 dimensions from generalised Scherk-Schwarz reductions," *JHEP*, vol. 1305, p. 161, 2013.
- [122] O. Hohm and H. Samtleben, "Consistent Kaluza-Klein Truncations via Exceptional Field Theory," JHEP, vol. 1501, p. 131, 2015.
- [123] H. A. Buchdahl, "Reciprocal static solutions of the equations $G_{\mu\nu} = 0$," The Quarterly Journal of Mathematics, vol. 5, no. 1, pp. 116–119, 1954.
- [124] J. M. Maldacena, "The Large N limit of superconformal field theories and supergravity," *Int.J. Theor. Phys.*, vol. 38, pp. 1113–1133, 1999.

- [125] G. Policastro, D. T. Son, and A. O. Starinets, "The Shear viscosity of strongly coupled N=4 supersymmetric Yang-Mills plasma," *Phys.Rev.Lett.*, vol. 87, p. 081601, 2001.
- [126] G. Policastro, D. T. Son, and A. O. Starinets, "From AdS / CFT correspondence to hydrodynamics," *JHEP*, vol. 0209, p. 043, 2002.
- [127] G. Policastro, D. T. Son, and A. O. Starinets, "From AdS / CFT correspondence to hydrodynamics. 2. Sound waves," JHEP, vol. 0212, p. 054, 2002.
- [128] P. Kovtun, D. T. Son, and A. O. Starinets, "Viscosity in strongly interacting quantum field theories from black hole physics," *Phys.Rev.Lett.*, vol. 94, p. 111601, 2005.
- [129] R. Baier, P. Romatschke, D. T. Son, A. O. Starinets, and M. A. Stephanov, "Relativistic viscous hydrodynamics, conformal invariance, and holography," *JHEP*, vol. 0804, p. 100, 2008.
- [130] I. Fouxon and Y. Oz, "CFT Hydrodynamics: Symmetries, Exact Solutions and Gravity," JHEP, vol. 0903, p. 120, 2009.
- [131] S. Bhattacharyya, S. Minwalla, and S. R. Wadia, "The Incompressible Non-Relativistic Navier-Stokes Equation from Gravity," *JHEP*, vol. 0908, p. 059, 2009.
- [132] N. Iqbal and H. Liu, "Universality of the hydrodynamic limit in AdS/CFT and the membrane paradigm," *Phys.Rev.*, vol. D79, p. 025023, 2009.
- [133] S. S. Gubser, S. S. Pufu, and F. D. Rocha, "Bulk viscosity of strongly coupled plasmas with holographic duals," *JHEP*, vol. 0808, p. 085, 2008.
- [134] G. Policastro, "Supersymmetric hydrodynamics from the AdS/CFT correspondence," JHEP, vol. 0902, p. 034, 2009.
- [135] J. Hansen and P. Kraus, "Nonlinear Magnetohydrodynamics from Gravity," JHEP, vol. 0904, p. 048, 2009.
- [136] J. Hansen and P. Kraus, "S-duality in AdS/CFT magnetohydrodynamics," JHEP, vol. 0910, p. 047, 2009.
- [137] C. Eling and Y. Oz, "Relativistic CFT Hydrodynamics from the Membrane Paradigm," JHEP, vol. 1002, p. 069, 2010.

- [138] I. Bredberg, C. Keeler, V. Lysov, and A. Strominger, "Wilsonian Approach to Fluid/Gravity Duality," *JHEP*, vol. 1103, p. 141, 2011.
- [139] D. Brattan, J. Camps, R. Loganayagam, and M. Rangamani, "CFT dual of the AdS Dirichlet problem : Fluid/Gravity on cut-off surfaces," *JHEP*, vol. 1112, p. 090, 2011.
- [140] S. Kuperstein and A. Mukhopadhyay, "The unconditional RG flow of the relativistic holographic fluid," *JHEP*, vol. 1111, p. 130, 2011.
- [141] R. G. Leigh, A. C. Petkou, and P. M. Petropoulos, "Holographic Three-Dimensional Fluids with Nontrivial Vorticity," *Phys. Rev.*, vol. D85, p. 086010, 2012.
- [142] T.-Z. Huang, Y. Ling, W.-J. Pan, Y. Tian, and X.-N. Wu, "From Petrov-Einstein to Navier-Stokes in Spatially Curved Spacetime," *JHEP*, vol. 1110, p. 079, 2011.
- [143] S. Kolekar and D. Kothawala, "Membrane Paradigm and Horizon Thermodynamics in Lanczos-Lovelock gravity," *JHEP*, vol. 1202, p. 006, 2012.
- [144] T.-Z. Huang, Y. Ling, W.-J. Pan, Y. Tian, and X.-N. Wu, "Fluid/gravity duality with Petrov-like boundary condition in a spacetime with a cosmological constant," *Phys.Rev.*, vol. D85, p. 123531, 2012.
- [145] G. Chirco, "Thermodynamic aspects of gravity," 2011.
- [146] A. Buchel, U. Gursoy, and E. Kiritsis, "Holographic bulk viscosity: GPR versus EO," JHEP, vol. 1109, p. 095, 2011.
- [147] A. Adams, L. D. Carr, T. Schäfer, P. Steinberg, and J. E. Thomas, "Strongly Correlated Quantum Fluids: Ultracold Quantum Gases, Quantum Chromodynamic Plasmas, and Holographic Duality," *New J. Phys.*, vol. 14, p. 115009, 2012.
- [148] X. Bai, Y.-P. Hu, B.-H. Lee, and Y.-L. Zhang, "Holographic Charged Fluid with Anomalous Current at Finite Cutoff Surface in Einstein-Maxwell Gravity," *JHEP*, vol. 1211, p. 054, 2012.
- [149] M. M. Caldarelli, R. G. Leigh, A. C. Petkou, P. M. Petropoulos, V. Pozzoli, et al., "Vorticity in holographic fluids," PoS, vol. CORFU2011, p. 076, 2011.

- [150] R.-G. Cai, T.-J. Li, Y.-H. Qi, and Y.-L. Zhang, "Incompressible Navier-Stokes Equations from Einstein Gravity with Chern-Simons Term," *Phys.Rev.*, vol. D86, p. 086008, 2012.
- [151] C.-Y. Zhang, Y. Ling, C. Niu, Y. Tian, and X.-N. Wu, "Magnetohydrodynamics from gravity," *Phys.Rev.*, vol. D86, p. 084043, 2012.
- [152] G. Compere, P. McFadden, K. Skenderis, and M. Taylor, "The relativistic fluid dual to vacuum Einstein gravity," *JHEP*, vol. 1203, p. 076, 2012.
- [153] M. M. Caldarelli, J. Camps, B. GoutA ©raux, and K. Skenderis, "AdS/Ricciflat correspondence," *JHEP*, vol. 1404, p. 071, 2014.
- [154] D.-C. Zou and B. Wang, "Holographic parity violating charged fluid dual to Chern-Simons modified gravity," *Phys.Rev.*, vol. D89, no. 6, p. 064036, 2014.
- [155] S. R. Green, F. Carrasco, and L. Lehner, "Holographic Path to the Turbulent Side of Gravity," *Phys.Rev.*, vol. X4, no. 1, p. 011001, 2014.
- [156] T. Ashok, "Forced Fluid Dynamics from Gravity in Arbitrary Dimensions," JHEP, vol. 1403, p. 138, 2014.
- [157] D. BRATTAN et al., Aspects of hydrodynamics in AdS/CMT. PhD thesis, Durham University, 2013.
- [158] B. Wu and L. Zhao, "Gravity-mediated holography in fluid dynamics," *Nucl. Phys.*, vol. B874, pp. 177–187, 2013.
- [159] S. Kuperstein and A. Mukhopadhyay, "Spacetime emergence via holographic RG flow from incompressible Navier-Stokes at the horizon," *JHEP*, vol. 1311, p. 086, 2013.
- [160] R.-G. Cai, L. Li, Q. Yang, and Y.-L. Zhang, "Petrov type I Condition and Dual Fluid Dynamics," *JHEP*, vol. 1304, p. 118, 2013.
- [161] X. Wu, Y. Ling, Y. Tian, and C. Zhang, "Fluid/Gravity Correspondence for General Non-rotating Black Holes," *Class. Quant. Grav.*, vol. 30, p. 145012, 2013.
- [162] J. Armas, "How Fluids Bend: the Elastic Expansion for Higher-Dimensional Black Holes," JHEP, vol. 1309, p. 073, 2013.

- [163] D. Klemm and A. Maiorana, "Fluid dynamics on ultrastatic spacetimes and dual black holes," *JHEP*, vol. 1407, p. 122, 2014.
- [164] Y. Bu and M. Lublinsky, "Linearized fluid/gravity correspondence: from shear viscosity to all order hydrodynamics," *JHEP*, vol. 1411, p. 064, 2014.
- [165] D. Arean and J. Tarrio, "Bifundamental Superfluids from Holography," JHEP, vol. 1504, p. 083, 2015.
- [166] J. Sadeghi and Z. Amoozad, "Blackfold as a Conformal Fluid," 2015.
- [167] P. M. Chesler and W. van der Schee, "Early thermalization, hydrodynamics and energy loss in AdS/CFT," 2015.
- [168] S. Bhattacharyya, A. De, S. Minwalla, R. Mohan, and A. Saha, "A membrane paradigm at large D," 2015.
- [169] R. Emparan, V. E. Hubeny, and M. Rangamani, "Effective hydrodynamics of black D3-branes," *JHEP*, vol. 1306, p. 035, 2013.
- [170] V. Giangreco M. Puletti, S. Nowling, L. Thorlacius, and T. Zingg, "Magnetic oscillations in a holographic liquid," *Phys.Rev.*, vol. D91, no. 8, p. 086008, 2015.
- [171] X. Hao, B. Wu, and L. Zhao, "Compressible forced viscous fluid from product Einstein manifolds," 2015.
- [172] M. Crossley, P. Glorioso, H. Liu, and Y. Wang, "Off-shell hydrodynamics from holography," 2015.
- [173] J. de Boer, M. P. Heller, and N. Pinzani-Fokeeva, "Effective actions for relativistic fluids from holography," 2015.
- [174] M. Blake, "Momentum relaxation from the fluid/gravity correspondence," 2015.
- [175] A. Donos and J. P. Gauntlett, "Navier-Stokes on Black Hole Horizons and DC Thermoelectric Conductivity," 2015.
- [176] M. Rangamani, "Gravity and Hydrodynamics: Lectures on the fluid-gravity correspondence," *Class. Quant. Grav.*, vol. 26, p. 224003, 2009.
- [177] V. E. Hubeny, S. Minwalla, and M. Rangamani, "The fluid/gravity correspondence," pp. 348–383, 2012.
- [178] S. Bhattacharyya, V. E. Hubeny, S. Minwalla, and M. Rangamani, "Nonlinear Fluid Dynamics from Gravity," *JHEP*, vol. 0802, p. 045, 2008.
- [179] S. Bhattacharyya, R. Loganayagam, I. Mandal, S. Minwalla, and A. Sharma, "Conformal Nonlinear Fluid Dynamics from Gravity in Arbitrary Dimensions," *JHEP*, vol. 0812, p. 116, 2008.
- [180] C. Eling, A. Meyer, and Y. Oz, "The Relativistic Rindler Hydrodynamics," *JHEP*, vol. 1205, p. 116, 2012.
- [181] N. Pinzani-Fokeeva and M. Taylor, "Towards a general fluid/gravity correspondence," *Phys.Rev.*, vol. D91, no. 4, p. 044001, 2015.
- [182] I. Bredberg, C. Keeler, V. Lysov, and A. Strominger, "From Navier-Stokes To Einstein," JHEP, vol. 1207, p. 146, 2012.
- [183] G. Compere, P. McFadden, K. Skenderis, and M. Taylor, "The Holographic fluid dual to vacuum Einstein gravity," *JHEP*, vol. 1107, p. 050, 2011.
- [184] R.-G. Cai, L. Li, and Y.-L. Zhang, "Non-Relativistic Fluid Dual to Asymptotically AdS Gravity at Finite Cutoff Surface," JHEP, vol. 1107, p. 027, 2011.
- [185] C. Niu, Y. Tian, X.-N. Wu, and Y. Ling, "Incompressible Navier-Stokes Equation from Einstein-Maxwell and Gauss-Bonnet-Maxwell Theories," *Phys.Lett.*, vol. B711, pp. 411–416, 2012.
- [186] Y.-P. Hu, P. Sun, and J.-H. Zhang, "Hydrodynamics with conserved current via AdS/CFT correspondence in the Maxwell-Gauss-Bonnet gravity," *Phys.Rev.*, vol. D83, p. 126003, 2011.
- [187] R.-G. Cai, L. Li, Z.-Y. Nie, and Y.-L. Zhang, "Holographic Forced Fluid Dynamics in Non-relativistic Limit," *Nucl. Phys.*, vol. B864, pp. 260–284, 2012.
- [188] D.-C. Zou, S.-J. Zhang, and B. Wang, "Holographic charged fluid dual to third order Lovelock gravity," *Phys.Rev.*, vol. D87, no. 8, p. 084032, 2013.
- [189] R.-G. Cai, Q. Yang, and Y.-L. Zhang, "Petrov type I Condition and Rindler Fluid in Vacuum Einstein-Gauss-Bonnet Gravity," *JHEP*, vol. 1412, p. 147, 2014.

- [190] Y. Bu, M. Lublinsky, and A. Sharon, "Hydrodynamics dual to Einstein-Gauss-Bonnet gravity: all-order gradient resummation," 2015.
- [191] D. Anninos, T. Anous, I. Bredberg, and G. S. Ng, "Incompressible Fluids of the de Sitter Horizon and Beyond," *JHEP*, vol. 1205, p. 107, 2012.
- [192] I. Bredberg and A. Strominger, "Black Holes as Incompressible Fluids on the Sphere," JHEP, vol. 1205, p. 043, 2012.
- [193] V. Lysov, "On the Magnetohydrodynamics/Gravity Correspondence," 2013.
- [194] A. Giveon, M. Porrati, and E. Rabinovici, "Target space duality in string theory," *Phys.Rept.*, vol. 244, pp. 77–202, 1994.
- [195] G. Neugebauer and D. Kramer, "Eine methode zur konstruktion stationärer einstein-maxwell-felder," Annalen der Physik, vol. 479, no. 1-2, pp. 62–71, 1969.
- [196] M. Mars, "Space-time Ehlers group: Transformation law for the Weyl tensor," Class. Quant. Grav., vol. 18, pp. 719–738, 2001.
- [197] K. Kikkawa and M. Yamasaki, "Casimir Effects in Superstring Theories," *Phys.Lett.*, vol. B149, p. 357, 1984.
- [198] N. Sakai and I. Senda, "Vacuum Energies of String Compactified on Torus," Prog. Theor. Phys., vol. 75, p. 692, 1986.
- [199] T. Buscher, "A symmetry of the string background field equations," *Physics Letters B*, vol. 194, no. 1, pp. 59–62, 1987.
- [200] T. Buscher, "Path Integral Derivation of Quantum Duality in Nonlinear Sigma Models," *Phys.Lett.*, vol. B201, p. 466, 1988.
- [201] A. A. Tseytlin, "Duality and dilaton," *Modern Physics Letters A*, vol. 6, no. 19, pp. 1721–1732, 1991.
- [202] A. Schwarz and A. A. Tseytlin, "Dilation shift under duality and torsion of elliptic complex," *Nuclear Physics B*, vol. 399, no. 2, pp. 691–708, 1993.
- [203] P. C. Aichelburg and R. U. Sexl, "On the gravitational field of a massless particle," *General Relativity and Gravitation*, vol. 2, no. 4, pp. 303–312, 1971.

- [204] A. Dabholkar, G. W. Gibbons, J. A. Harvey, and F. Ruiz Ruiz, "Superstrings and Solitons," *Nucl. Phys.*, vol. B340, pp. 33–55, 1990.
- [205] T. Adawi, M. Cederwall, U. Gran, B. E. Nilsson, and B. Razaznejad, "Goldstone tensor modes," *JHEP*, vol. 9902, p. 001, 1999.
- [206] D. M. Kaplan and J. Michelson, "Zero modes for the D = 11 membrane and five-brane," *Phys.Rev.*, vol. D53, pp. 3474–3476, 1996.
- [207] R. Floreanini and R. Jackiw, "Selfdual Fields as Charge Density Solitons," *Phys.Rev.Lett.*, vol. 59, p. 1873, 1987.
- [208] B. de Wit and H. Nicolai, "d = 11 Supergravity With Local SU(8) Invariance," Nucl. Phys., vol. B274, p. 363, 1986.
- [209] C. Hull, "Generalised Geometry for M-Theory," JHEP, vol. 0707, p. 079, 2007.
- [210] P. P. Pacheco and D. Waldram, "M-theory, exceptional generalised geometry and superpotentials," *JHEP*, vol. 0809, p. 123, 2008.
- [211] H. Godazgar, M. Godazgar, and M. J. Perry, "E8 duality and dual gravity," JHEP, vol. 1306, p. 044, 2013.
- [212] M. Duff and J. Lu, "Duality Rotations in Membrane Theory," Nucl. Phys., vol. B347, pp. 394–419, 1990.
- [213] J. Ehlers, Konstruktionen und Charakterisierung von Losungen der Einsteinschen Gravitationsfeldgleichungen. PhD thesis, University of Hamburg, 1957.
- [214] R. P. Geroch, "A Method for generating solutions of Einstein's equations," J.Math.Phys., vol. 12, pp. 918–924, 1971.
- [215] B. K. Harrison, "New solutions of the einstein-maxwell equations from old," Journal of Mathematical Physics, vol. 9, no. 11, pp. 1744–1752, 1968.
- [216] H. Buchdahl, "Reciprocal static metrics and scalar fields in the general theory of relativity," *Physical Review*, vol. 115, no. 5, p. 1325, 1959.
- [217] D. Galtsov and O. Kechkin, "Ehlers-Harrison type transformations in dilaton - axion gravity," *Phys.Rev.*, vol. D50, pp. 7394–7399, 1994.
- [218] I. R. Pinkstone, "Structure of dualities in bosonic string theory," 1995.

- [219] O. Kechkin and M. Yurova, "Kramer-Neugebauer transformation for Einstein-Maxwell dilaton - axion theory," *Phys.Rev.*, vol. D54, pp. 6132–6135, 1996.
- [220] S. S. Yazadjiev, "Exact static solutions in einstein-maxwell-dilaton gravity with arbitrary dilaton coupling parameter," *arXiv preprint gr-qc/0101078*, 2001.
- [221] S. S. Yazadjiev, "Generating dyonic solutions in 5D Einstein-dilaton gravity with antisymmetric forms and dyonic black rings," *Phys.Rev.*, vol. D73, p. 124032, 2006.
- [222] R. M. Wald, General Relativity. The University of Chicago Press, first edition ed., 1984.
- [223] D. Kramer, H. Stephani, E. Herlt, and M. MacCallum, Exact Solutions of Einstein's Field equations. Cambridge University Press, first edition ed., 1980.
- [224] L. Landau, "Fluid mechanics: volume 6 (course of theoretical physics) author: Ld landau, em lifshitz, publisher: Bu," 1987.
- [225] P. Constantin, "Some open problems and research directions in the mathematical study of fluid dynamics," in *Mathematics Unlimited — 2001 and Beyond*, pp. 353–360, Springer Verlag, 2001.
- [226] O. A. Ladyzhenskaya and R. A. Silverman, The mathematical theory of viscous incompressible flow, vol. 76. Gordon and Breach New York, 1969.
- [227] T. Damour, Quelques propriétés mécaniques, électromagnétiques, thermodynamiques et quantiques des trous noirs. PhD thesis, Université Pierre et Marie Curie, Paris VI, 1979.
- [228] K. S. Thorne, R. H. Price, and D. A. Macdonald, Black Holes: The Membrane Paradigm. Yale University Press, 1986.
- [229] L. Susskind and E. Witten, "The holographic bound in anti-de sitter space," arXiv preprint hep-th/9805114, 1998.
- [230] C. G. Torre and I. M. Anderson, "Symmetries of the Einstein equations," *Phys.Rev.Lett.*, vol. 70, pp. 3525–3529, 1993.
- [231] D. Maison, "Duality and hidden symmetries in gravitational theories," in *Einstein's field equations and their physical implications*, pp. 273–323, Springer, 2000.

- [232] V. Gusyatnikova and V. Yumaguzhin, "Symmetries and conservation laws of navier-stokes equations," in *Symmetries of Partial Differential Equations*, pp. 65–81, Springer, 1989.
- [233] P. Horvathy and P.-M. Zhang, "Non-relativistic conformal symmetries in fluid mechanics," *Eur. Phys. J.*, vol. C65, pp. 607–614, 2010.
- [234] B. Stewart, "A generalisation of the buchdahl transformation and perfect fluid solutions," *Journal of Physics A Mathematical General*, vol. 15, pp. 1799–1805, 1982.
- [235] S. Berger, R. Hojman, and J. Santamarina, "General exact solutions of einstein equations for static perfect fluids with spherical symmetry," *Journal of Mathematical Physics*, vol. 28, no. 12, pp. 2949–2950, 1987.
- [236] H. Stephani, "Symmetries of einstein's field equations with a perfect fluid source as examples of lie-backlund symmetries," *Journal of Mathematical Physics*, vol. 29, no. 7, pp. 1650–1654, 1988.
- [237] D. Garfinkle, E. Glass, and J. Krisch, "Solution generating with perfect fluids," Gen. Rel. Grav., vol. 29, pp. 467–480, 1997.
- [238] I. Racz and J. Zsigrai, "Generating new perfect fluid solutions from known ones," *Class. Quant. Grav.*, vol. 13, pp. 2783–2796, 1996.
- [239] I. Racz and J. Zsigrai, "Generating new perfect fluid solutions from known ones: 2," *Class.Quant.Grav.*, vol. 14, pp. 1997–2002, 1997.
- [240] J. Zsigrai, "A generalized method for constructing perfect-fluid solutions of einstein's equations," *Progress of Theoretical Physics*, vol. 104, no. 2, pp. 359– 384, 2000.
- [241] P. Boonserm, M. Visser, and S. Weinfurtner, "Solution generating theorems for perfect fluid spheres," J.Phys.Conf.Ser., vol. 68, p. 012055, 2007.
- [242] C. Lanczos, "A remarkable property of the riemann-christoffel tensor in four dimensions," Annals of Mathematics, pp. 842–850, 1938.
- [243] D. Lovelock, "The einstein tensor and its generalizations," Journal of Mathematical Physics, vol. 12, no. 3, pp. 498–501, 1971.

- [244] R. Brustein and A. Medved, "Lovelock gravity is equivalent to Einstein gravity coupled to form fields," *Phys.Rev.*, vol. D88, p. 064010, 2013.
- [245] R. G. Leigh, A. C. Petkou, and P. M. Petropoulos, "Holographic Fluids with Vorticity and Analogue Gravity," *JHEP*, vol. 1211, p. 121, 2012.
- [246] S. A. Hartnoll, "Horizons, holography and condensed matter," arXiv preprint arXiv:1106.4324, 2011.
- [247] M. Nakahara, Geometry, Topology and Physics. Taylor & Francis Group, second edition ed., 2003.